Health-related quality of life and survival in patients with head and neck cancer

First steps towards improved survivorship care

Annette Josephine Wehman - van Nieuwenhuizen

This research in this thesis was performed at the department of Otolaryngology and Head and Neck surgery of the Amsterdam University Medical Center Amsterdam, location VUmc.

This thesis was accomplished with financial support from the Dutch Cancer Society, Wehealthcare, Schoonenberg Hoorsupport and Pentax.

Cover design by publiss.nl Print: Ridderprint | www.ridderprint.nl

Copyright C by A.J. Wehman – van Nieuwenhuizen 2020. All rhights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form of by any means without prior permission of the author.

VRIJE UNIVERSITEIT

Health-related quality of life and survival in patients with head and neck cancer First steps towards improved survivorship care

ACADEMISCH PROEFSCHRIFT

ter verkrijging van de graad Doctor aan de Vrije Universiteit Amsterdam, op gezag van de rector magnificus prof.dr. V. Subramaniam, in het openbaar te verdedigen ten overstaan van de promotiecommissie van de Faculteit der Geneeskunde op vrijdag 4 september 2020 om 11.45 uur in de aula van de universiteit, De Boelelaan 1105

door

Annette Josephine van Nieuwenhuizen

geboren te Oegstgeest

Promotoren:	Prof. dr. I.M. Verdonck – de Leeuw
	Prof. dr. C.R. Leemans
Co-promotor:	Dr. L.M. Buffart

Ter nagedachtenis aan mijn moeder, Marie-Louise

TABLE OF CONTENTS

Chapter 1	General introduction and outline of this thesis	
Chapter 2	The value of quality of life questionnaires in head and neck cancer.	23
Chapter 3	The association between health-related quality of life and survival in patients with head and neck cancer: a systematic review .	35
Chapter 4	Health Related Quality of Life and overall survival: a prospective study in patients with head and neck cancer treated with radiotherapy	65
Chapter 5	Patient-reported physical activity and the association with health-related quality of life in head and neck cancer survivors.	83
Chapter 6	A comprehensive assessment protocol including patient reported outcomes, physical tests, and biological sampling in newly diagnosed patients with head and neck cancer: is it feasible?	101
Chapter 7	General discussion and future perspectives	123
Chapter 8	Summary	139
Chapter 9	Nederlandse samenvatting	145
Chapter 10	Addendum	151



- CHAPTER 1

GENERAL INTRODUCTION AND OUTLINE OF THIS THESIS

Encouraging results have been made in the past decades towards improving clinical outcomes in cancer treatment.1 Nowadays health related quality of life (HRQoL) is recognized as an inseparable factor to account for along the cancer trajectory. Sometimes the medical benefits might even be outweighed by the deleterious impact on HRQoL. A cancer diagnosis and it's treatment can severely impair an individual's HRQoL. Patients are often confronted with various symptoms and toxicities that are associated with cancer diagnosis and treatment such as pain and fatigue, but also physical, emotional, functional, social, and occupational dysfunction.²⁻⁴ Previous research has shown that HRQoL is significantly associated with survival.⁵⁻⁹ In addition, lifestyle behaviors (e.g. smoking, alcohol use, dietary intake and physical activity) have been shown to be associated with survival.¹⁰⁻¹⁵ In this thesis, the focus is on HRQoL in head and neck cancer (HNC) patients. These patients have to cope with, in addition to generic cancer-related symptoms, specific HNC symptoms, such as oral dysfunction, and speech and swallowing dysfunction.¹⁶⁻²² This thesis addresses HRQoL in relation to survivor and survivorship in people with head and neck cancer. In this chapter, background information is provided on HNC and its treatment, HRQoL, and HNC survivorship. The aims of this thesis are described and the outline of this thesis is presented.

Head and neck cancer

Head and neck cancer (HNC) encompasses tumors in the upper respiratory or digestive tract. The most common sites in the head and neck region are oral cavity, oropharynx, hypopharynx or larynx. The main tumor histopathology is squamous cell carcinoma which develops in the epithelial layer of the mucosal surfaces from the affected tumor sites.²³ To date, HNC is the sixth leading type of cancer worldwide, with approximately 705.781 new cases and 358.144 deaths in 2018.²⁴ In Europe and the United States, five year survival rate has slightly improved over the past decades. In The Netherlands, five year survival rate was 59% in 2015.²⁵ Main risk factors for the development of HNC are tobacco use and excessive alcohol consumption, which seem to have a synergistic effect.^{26,27} The presence of the oncogenic human papilloma virus (HPV) infection is another factor that is associated with an increased risk mainly for developing oropharyngeal cancer and to a lesser extent oral carcinoma. ^{26,27} Male-female ratio of HNC is approximately 1.5. The incidence in men tends to stabilize while the incidence in women still increases due to increased tobacco consumption by females in the 1980s. Most newly diagnosed patients with HNC are over 45 years of age.^{26,27}



Schematic representation of the head and neck region relevant for head and neck carcinomas.

Diagnosis and treatment

The first symptoms of a tumor in the head and neck region comprise mouth pain, non-healing ulcers, dysphagia, hoarseness, swelling in the neck or referred otalgia. In general, one third of the patients with HNC are diagnosed with an early stage (I and II) of disease, while two-third are diagnosed with advanced disease (III and IV).²³ Treatment of patients with HNC is multidisciplinary and based on tumor site, TNM stage (the size of the primary tumor (T), presence and number of regional lymph node metastasis (N) and presence of distant metastasis (M)) and overall condition of the patient (i.e. age or cognitive status).^{28,29} Treatment for patients with HNC often involves surgery, radiotherapy, chemotherapy or a combination of these modalities. Patients with early stage disease have small tumors confined to the primary site, without metastasis, and are usually treated with single treatment modality (surgery or radiotherapy). Patients with advanced disease have larger tumors and / or locally metastasized tumors. In case of curative treatment, this mostly consists of a combination of surgery and postoperative (chemo)radiotherapy or the concomitant application of chemotherapy combined with locoregional radiotherapy.^{28,29}

Heath-related quality of life

HRQoL is a multidimensional concept and encompasses physical, psychological and social well-being and functioning.³⁰ Patients with HNC may be confronted with specific symptoms, such as oral dysfunction, swallowing and speech problems.^{16,17,19-21,31-35} This may have a distinct impact on HRQoL. Previous studies in patients with HNC showed that several domains of HRQoL, including general health, mental health, physical function, appearance, employment and social functioning, decline during and immediately after treatment, but often improve after 6 months.^{17,20-22} Previous research also showed that HRQoL seems to stabilize from one year after treatment up to 5-year follow up.¹⁶⁻²² However, late effects of HNC cancer and its treatment may also occur. ³⁶ To provide individualized (supportive) care, monitoring HRQoL in a structured manner in clinical research and practice is important. HRQoL is typically assessed by patient reported outcome measures (PROMs).37.38 PROMs supplement clinician rated scores and/or objective testing by revealing the impact of the disease and its treatment on physical, psychological and general functioning of the patient.³⁹ Wide1y-used PROMs among patients with HNC are the European Organization for Research and Treatment of Cancer, (EORTC) Quality of Life Questionnaires.^{39,40} The EORTC Quality of Life Questionnaire (QLQ-C30) provide insight into patients' global quality of life, health related functioning (e.g. physical, emotional, social) and general cancer symptoms (e.g. pain and fatigue).^{39,40} The EORTC Head and Neck Module (EORTC QLQ-HN35) is tailored to HNC symptoms such as problems with swallowing and speech.⁴¹ Recently, this PROM was updated and extended into the EORTC QLQ-HN43.42

HRQoL in relation to survival

Interestingly the course of HRQoL has shown to be different between HNC survivors and non-survivors during the first 2 years after treatment.³⁵ Also a recent study demonstrated that overall HRQoL decreased progressively until a year, then recovered toward baseline between 2 and 5 years. However, patients with shorter survival had lower HRQoL pre-diagnosis with a steeper decline in HRQoL during diagnosis and treatment. Higher pre-diagnosis HRQOL was independently associated with improved overall survival.¹¹

Other studies investigating the association between HRQoL and survival showed a significant association, independently from other known demographic, lifestyle-related and clinical factors.⁴³⁻⁵² A review of Montazeri et al.⁶ revealed a significant association between HRQoL and survival for most cancers, but findings of studies on HNC patients were inconsistent and based on a limited number of studies. The review of Montazeri et al.⁶ and a meta-analysis from Quinten et al⁵³ showed a stronger association between pre-treatment physical functioning and survival compared to other HRQoL domains

in patients with various cancer sites. Quinten et al found a median survival time of 20 months among patients with low physical functioning compared to 65 months among patients with high physical functioning.⁵³

However, the observational studies in patients with HNC examining the association between HRQoL and survival had several limitations. For example, relevant confounders were not taken into account in the association between HRQoL and survival, such as co-morbidity^{44,46,47,54-61} or lifestyle (e.g. smoking and alcohol consumption).^{44,55,58,62-64} Furthermore, most often pre-treatment HRQoL was investigated, whereas also HRQoL after treatment and changes in HRQoL from diagnosis to follow-up may be important in the association with survival. The review of Montazeri et al.⁶ showed that for some cancers, HRQoL after treatment was prognostic for survival, where pre-treatment HRQoL was not. Although HRQoL after treatment and change in HRQoL seems to be associated with survival as well, there is only a limited number of studies and there is large variation in these studies concerning assessment time points.^{45-47,59,61,64} As a consequence, it remains difficult to draw firm conclusions on the association between HRQoL and survival in HNC patients.

Head and neck cancer survivorship

With the increase in incidence and improved survival rates, more people have to cope with living beyond HNC cancer.^{65,66} Cancer and its treatments have long lasting effects, and survivors are at higher risks for comorbidities, physical and psychosocial problems throughout their lifetime.⁶⁷ When a patient becomes a cancer survivor is debatable, at diagnosis, after completion of definite treatment or when at least 5 years have elapsed without the sign of any recurrence.⁶⁸ From a more individualized perspective it is defined as "living with, through and beyond a cancer diagnosis.⁶⁹ For HNC specifically, patients and their families are confronted with profound and numerous disabilities due to the anatomical complexity of the head and neck region. Swallowing and speech impairments are common throughout the entire cancer trajectory. Within the first year of radiotherapy treatment, approximately half of HNC survivors suffer from dysphagia and dysarthria.⁷⁰⁻⁷² Unfortunately long term effects of radiotherapy are often present on speech and swallowing. For instance, a majority (68%) of HNC survivors reported voice problems even 10 years after radiotherapy.73 Furthermore, the two-years prevalence of dysphagia is 45% among HNC survivors (all therapies) and 4.6-7.8 times more likely to occur in comparison to non-cancer controls.⁷⁴ These findings are reflected in the needs of patients with HNC, where maintaining the ability to swallow and speak as their top 2 functional priorities.⁷⁵ These impairments in swallowing and / or speech are primary concerns for HNC survivors and directly associated with reduced HRQoL.⁷⁶ In addition HNC symptoms seem to be significantly associated with reduced physical activity because of dry mouth or throat, difficulty eating, and shortness of breath).⁷⁷ In cancer survivorship promoting a healthy lifestyle is critical. In patients with HNC awareness on lifestyle related factors such as smoking, alcohol use and nutritional status seems to be eminent. However, physical activity seems to be less recognized as an important lifestyle related factor. Studies suggests that only 9% of nurses and from 19% to 23% of oncology physicians refer patients with cancer to exercise programming.⁷⁸ Also, in contrast to HNC symptoms and other domains of HRQoL not much is known on physical activity in HNC survivors.

Physical activity can be defined as any bodily movement that results in energy expenditure.⁷⁹ In patients with other types of cancer, mainly breast and prostate cancer, physical activity has shown to have beneficial effects on physical and psychosocial function and HRQoL during and after treatment.⁸⁰⁻⁸⁴ Individualized participant data of 4519 individual patients with cancer revealed that exercise significantly improved physical function and global QoL.⁸⁵ Furthermore higher levels of moderate-to-vigorous physical activity have been associated with lower mortality risk in survivors of breast, colon and prostate cancer.⁸⁶⁻⁹⁰ Also in HNC survivors, higher pre-treatment levels of physical activity and physical function were found to be associated with higher HRQoL.^{91,92} Physical activity seems to decrease following HNC diagnosis and during treatment.⁹²⁻⁹⁵ Additionally, little is known on physical activity in HNC survivors in the longer term.

Objectives and outline of this thesis

Summarizing the current literature, there are still important issues that remain to be answered. Two key gaps in knowledge are addressed in this thesis. It is still unclear which HRQoL domains are associated survival in HNC, at which time-point across the cancer journey, and whether absolute values and/or changes in HRQoL are associated with survival. Furthermore, more knowledge on HRQoL in HNC survivorship is needed and especially the role of physical activity.

Therefore, this thesis aims to investigate the association between HRQoL and survival in patients with HNC, and to investigate physical activity and the relationship with HRQoL in long-term HNC survivors.

First, the current knowledge on the value of using HRQoL questionnaires in in patients with HNC is reviewed in-depth. Chapter 2 describes a literature review on the value of HRQoL questionnaires in patients with HNC.

The association between HRQoL and survival in patients with HNC is investigated in the studies described in Chapter 3 and 4. Chapter 3 reports on a systematic literature review of prospective studies on the association between HRQoL and survival in patients with HNC. This study aimed to identify which HRQoL domains are most

14

strongly associated with survival, adjusting for important clinical, demographic and lifestyle-related factors.

Chapter 4 describes the results of a prospective study among 948 newly diagnosed HNC patients. This study aimed to examine whether pre-treatment, post-treatment or change in HRQoL is associated with survival.

The study described in Chapter 5 aimed to obtain insight into patient reported physical activity in long-term HNC survivors in relation to HRQoL, adjusted for relevant demographic, lifestyle-related and clinical factors.

Chapter 6 describes the feasibility of performing a comprehensive study, including questionnaires (635 items), a home visit (including a (psychiatric) interview, physical tests, blood and saliva collection), and tissue collection.

Chapter 7 discusses the main findings of this thesis, the methodological considerations, implications for clinical practice, recommendations for further research and a conclusion.

REFERENCES

- Stewart BW, Wild CP: World Cancer Report 2014. Lyon, France: International Agency for research on cancer, 2014
- 2. Brown LF, Kroenke K: Cancer-related fatigue and its associations with depression and anxiety: a systematic review. Psychosomatics 50:440-7, 2009
- Foster C, Wright D, Hill H, et al: Psychosocial implications of living 5 years or more following a cancer diagnosis: a systematic review of the research evidence. Eur J Cancer Care (Engl) 18:223-47, 2009
- 4. Harrington CB, Hansen JA, Moskowitz M, et al: It's not over when it's over: long-term symptoms in cancer survivors--a systematic review. Int J Psychiatry Med 40:163-81, 2010
- Quinten C, Coens C, Mauer M, et al: Baseline quality of life as a prognostic indicator of survival: a meta-analysis of individual patient data from EORTC clinical trials. Lancet Oncol 10:865-71, 2009
- 6. Montazeri A: Quality of life data as prognostic indicators of survival in cancer patients: an overview of the literature from 1982 to 2008. Health Qual. Life Outcomes 7:102, 2009
- 7. Quinten C, Martinelli F, Coens C, et al: A global analysis of multitrial data investigating quality of life and symptoms as prognostic factors for survival in different tumor sites. Cancer, 2013
- Ediebah DE, Quinten C, Coens C, et al: Quality of life as a prognostic indicator of survival: A pooled analysis of individual patient data from canadian cancer trials group clinical trials. Cancer 124:3409-3416, 2018
- 9. Gotay CC, Kawamoto CT, Bottomley A, et al: The prognostic significance of patient-reported outcomes in cancer clinical trials. J Clin Oncol 26:1355-63, 2008
- McTiernan A, Friedenreich CM, Katzmarzyk PT, et al: Physical Activity in Cancer Prevention and Survival: A Systematic Review. Med Sci Sports Exerc 51:1252-1261, 2019
- Rettig EM, D'Souza G, Thompson CB, et al: Health-related quality of life before and after head and neck squamous cell carcinoma: Analysis of the Surveillance, Epidemiology, and End Results-Medicare Health Outcomes Survey linkage. Cancer 122:1861-70, 2016
- 12. Beynon RA, Lang S, Schimansky S, et al: Tobacco smoking and alcohol drinking at diagnosis of head and neck cancer and all-cause mortality: Results from head and neck 5000, a prospective observational cohort of people with head and neck cancer. Int J Cancer 143:1114-1127, 2018
- Bruixola G, Caballero J, Papaccio F, et al: Prognostic Nutritional Index as an independent prognostic factor in locoregionally advanced squamous cell head and neck cancer. ESMO Open 3:e000425, 2018
- Kim SA, Roh JL, Lee SA, et al: Pretreatment depression as a prognostic indicator of survival and nutritional status in patients with head and neck cancer. Cancer 122:131-40, 2016
- 15. Moon H, Roh JL, Lee SW, et al: Prognostic value of nutritional and hematologic markers in head and neck squamous cell carcinoma treated by chemoradiotherapy. Radiother Oncol 118:330-4, 2016
- Abendstein H, Nordgren M, Boysen M, et al: Quality of life and head and neck cancer: a 5 year prospective study. Laryngoscope 115:2183-2192, 2005
- 17. Bjordal K, Ahlner-Elmqvist M, Hammerlid E, et al: A prospective study of quality of life in head and neck cancer patients. Part II: Longitudinal data. Laryngoscope 111:1440-1452, 2001
- de Graeff A, de Leeuw JR, Ros WJ, et al: Long-term quality of life of patients with head and neck cancer. Laryngoscope 110:98-106, 2000
- 19. Funk GF, Karnell LH, Christensen AJ: Long-term health-related quality of life in survivors of head and neck cancer. Arch. Otolaryngol. Head Neck Surg 138:123-133, 2012
- 20. Goldstein DP, Hynds KL, Christensen AJ, et al: Health-related quality of life profiles based on

16

survivorship status for head and neck cancer patients. Head Neck 29:221-229, 2007

- 21. Hammerlid E, Bjordal K, Ahlner-Elmqvist M, et al: A prospective study of quality of life in head and neck cancer patients. Part I: at diagnosis. Laryngoscope 111:669-680, 2001
- 22. Hammerlid E, Silander E, Hornestam L, et al: Health-related quality of life three years after diagnosis of head and neck cancer--a longitudinal study. Head Neck 23:113-125, 2001
- 23. Rettig EM, D'Souza G: Epidemiology of head and neck cancer. Surg Oncol Clin N Am 24:379-96, 2015
- 24. Organisation WH: http://gco.iarc.fr/today/data/factsheets/populations/900-world-fact-sheets.pdf, in GLOBOCAN Tgco (ed). 2018
- 25. http://www.cijfersoverkanker.nl/, 2015
- Braakhuis BJ, Visser O, Leemans CR: Oral and oropharyngeal cancer in The Netherlands between 1989 and 2006: Increasing incidence, but not in young adults. Oral Oncol 45:e85-e89, 2009
- 27. Braakhuis BJM, Leemans CR, Visser O: Incidence and survival trends of head and neck squamous cell carcinoma in the Netherlands between 1989 and 2011. Oral Oncol accepted for publication, 2014
- 28. Guideline database [head and neck tumors] accessed december 13, 2018
- 29. Dutch Association of Otolaryngology (NVKNO) [guideline head and neck tumours]. 2014
- Hayes RD, Reeve BB: Epidemiology and demography in public health, in Killewo J, Heggenhougen HK, Quah SR (eds): Measurement and modeling of health-related quality of life. . San Diego, Academic Press, 2010, pp 195–205.
- 31. Borghgraef K, Delaere P, Van den Bogaert W, et al: Quality of life in head and neck cancer patients. Acta Otorhinolaryngol. Belg 51:69-72, 1997
- 32. de Graeff A, de Leeuw J, Ros WJ, et al: Long-term quality of life of patients with head and neck cancer. Laryngoscope 110:98-106, 2000
- 33. Oskam IM, Verdonck-de Leeuw IM, Aaronson NK, et al: Prospective evaluation of health-related quality of life in long-term oral and oropharyngeal cancer survivors and the perceived need for supportive care. Oral Oncol 49:443-448, 2013
- 34. Rogers SN, Ahad SA, Murphy AP: A structured review and theme analysis of papers published on 'quality of life' in head and neck cancer: 2000-2005. Oral Oncol 43:843-868, 2007
- 35. Verdonck-de Leeuw IM, Buffart LM, Heymans MW, et al: The course of health-related quality of life in head and neck cancer patients treated with chemoradiation: a prospective cohort study. Radiother. Oncol 110:422-428, 2014
- Mehanna HM, Morton RP: Deterioration in quality-of-life of late (10-year) survivors of head and neck cancer. Clin. Otolaryngol 31:204-211, 2006
- Kanatas AN, Rogers SN: A systematic review of patient self-completed questionnaires suitable for oral and maxillofacial surgery. Br J Oral Maxillofac Surg 48:579-90, 2010
- Rogers SN, Forgie S, Lowe D, et al: Development of the International Classification of Functioning, Disability and Health as a brief head and neck cancer patient questionnaire. Int J Oral Maxillofac Surg 39:975-82, 2010
- 39. Aaronson NK, Ahmedzai S, Bergman B, et al: The European Organization for Research and Treatment of Cancer QLQ-C30: a quality-of-life instrument for use in international clinical trials in oncology. J. Natl. Cancer Inst 85:365-376, 1993
- Fayers P, Bottomley A: Quality of life research within the EORTC-the EORTC QLQ-C30. European Organisation for Research and Treatment of Cancer. Eur. J. Cancer 38 Suppl 4:S125-S133, 2002
- 41. Bjordal K, Ahlner-Elmqvist M, Tollesson E, et al: Development of a European Organization for Research

and Treatment of Cancer (EORTC) questionnaire module to be used in quality of life assessments in head and neck cancer patients. EORTC Quality of Life Study Group. Acta Oncol 33:879-885, 1994

- 42. Singer S, Amdal CD, Hammerlid E, et al: International validation of the revised European Organisation for Research and Treatment of Cancer Head and Neck Cancer Module, the EORTC QLQ-HN43: Phase IV. Head Neck 41:1725-1737, 2019
- 43. de Graeff A, de Leeuw JR, Ros WJ, et al: Pretreatment factors predicting quality of life after treatment for head and neck cancer. Head Neck 22:398-407, 2000
- 44. Fang FM, Liu YT, Tang Y, et al: Quality of life as a survival predictor for patients with advanced head and neck carcinoma treated with radiotherapy. Cancer 100:425-432, 2004
- 45. Karvonen-Gutierrez CA, Ronis DL, Fowler KE, et al: Quality of life scores predict survival among patients with head and neck cancer. J. Clin. Oncol 26:2754-2760, 2008
- 46. Mehanna HM, Morton RP: Does quality of life predict long-term survival in patients with head and neck cancer? Arch. Otolaryngol. Head Neck Surg 132:27-31, 2006
- 47. Meyer F, Fortin A, Gelinas M, et al: Health-related quality of life as a survival predictor for patients with localized head and neck cancer treated with radiation therapy. J. Clin. Oncol 27:2970-2976, 2009
- 48. Oskam IM, Verdonck-de Leeuw IM, Aaronson NK, et al: Quality of life as predictor of survival: a prospective study on patients treated with combined surgery and radiotherapy for advanced oral and oropharyngeal cancer. Radiother. Oncol 97:258-262, 2010
- 49. Osthus AA, Aarstad AK, Olofsson J, et al: Head and neck specific Health Related Quality of Life scores predict subsequent survival in successfully treated head and neck cancer patients: a prospective cohort study. Oral Oncol 47:974-979, 2011
- 50. Osthus AA, Aarstad AK, Olofsson J, et al: Prediction of survival by pretreatment health-related qualityof-life scores in a prospective cohort of patients with head and neck squamous cell carcinoma. JAMA Otolaryngol. Head Neck Surg 139:14-20, 2013
- 51. Aarstad HJ, Osthus AA, Aarstad HH, et al: General health-related quality of life scores from head and neck squamous cell carcinoma patients obtained throughout the first year following diagnosis predicted up to 10-year overall survival. Eur Arch Otorhinolaryngol 275:207-217, 2018
- 52. Yang CJ, Roh JL, Kim MJ, et al: Pretreatment quality of life as a prognostic factor for early survival and functional outcomes in patients with head and neck cancer. Qual Life Res 25:165-74, 2016
- 53. Quinten C, Coens C, Mauer M, et al: Baseline quality of life as a prognostic indicator of survival: a meta-analysis of individual patient data from EORTC clinical trials. Lancet Oncol 10:865-871, 2009
- 54. Coyne JC, Pajak TF, Harris J, et al: Emotional well-being does not predict survival in head and neck cancer patients: a Radiation Therapy Oncology Group study. Cancer 110:2568-2575, 2007
- 55. Curran D, Giralt J, Harari PM, et al: Quality of life in head and neck cancer patients after treatment with high-dose radiotherapy alone or in combination with cetuximab. J. Clin. Oncol 25:2191-2197, 2007
- de Graeff A, de Leeuw J, Ros WJ, et al: Sociodemographic factors and quality of life as prognostic indicators in head and neck cancer. Eur. J. Cancer 37:332-339, 2001
- 57. Fang FM, Tsai WL, Chien CY, et al: Pretreatment quality of life as a predictor of distant metastasis and survival for patients with nasopharyngeal carcinoma. J. Clin. Oncol 28:4384-4389, 2010
- Nordgren M, Jannert M, Boysen M, et al: Health-related quality of life in patients with pharyngeal carcinoma: a five-year follow-up. Head Neck 28:339-349, 2006
- Osthus AA, Aarstad AK, Olofsson J, et al: Health-related quality of life scores in long-term head and neck cancer survivors predict subsequent survival: a prospective cohort study. Clin. Otolaryngol 36:361-368, 2011
- 60. Siddiqui F, Pajak TF, Watkins-Bruner D, et al: Pretreatment quality of life predicts for locoregional
- 18

control in head and neck cancer patients: a radiation therapy oncology group analysis. Int. J. Radiat. Oncol. Biol. Phys 70:353-360, 2008

- 61. Thompson TL, Pagedar NA, Karnell LH, et al: Factors associated with mortality in 2-year survivors of head and neck cancer. Arch. Otolaryngol. Head Neck Surg 137:1100-1105, 2011
- 62. Nordgren M, Abendstein H, Jannert M, et al: Health-related quality of life five years after diagnosis of laryngeal carcinoma. Int. J. Radiat. Oncol. Biol. Phys 56:1333-1343, 2003
- 63. Nordgren M, Hammerlid E, Bjordal K, et al: Quality of life in oral carcinoma: a 5-year prospective study. Head Neck 30:461-470, 2008
- 64. Oskam IM, Verdonck-de L, I, Aaronson NK, et al: Quality of life as predictor of survival: a prospective study on patients treated with combined surgery and radiotherapy for advanced oral and oropharyngeal cancer. Radiother. Oncol 97:258-262, 2010
- 65. Ringash J: Survivorship and Quality of Life in Head and Neck Cancer. J Clin Oncol 33:3322-7, 2015
- 66. Ringash J, Bernstein LJ, Devins G, et al: Head and Neck Cancer Survivorship: Learning the Needs, Meeting the Needs. Semin Radiat Oncol 28:64-74, 2018
- 67. Elliott J, Fallows A, Staetsky L, et al: The health and well-being of cancer survivors in the UK: findings from a population-based survey. Br J Cancer 105 Suppl 1:S11-20, 2011
- 68. Bell K, Ristovski-Slijepcevic S: Cancer survivorship: why labels matter. J Clin Oncol 31:409-11, 2013
- **69**. national coalition for cancer survivorship: defining cancer survivorship. http://www.canceradvocacy. org/news/defining-cancer-survivorship/, 2014
- 70. Garcia-Peris P, Paron L, Velasco C, et al: Long-term prevalence of oropharyngeal dysphagia in head and neck cancer patients: Impact on quality of life. Clin Nutr 26:710-7, 2007
- 71. Rinkel RN, Verdonck-de Leeuw IM, Doornaert P, et al: Prevalence of swallowing and speech problems in daily life after chemoradiation for head and neck cancer based on cut-off scores of the patientreported outcome measures SWAL-QOL and SHI. Eur Arch Otorhinolaryngol 273:1849-55, 2016
- Setton J, Lee NY, Riaz N, et al: A multi-institution pooled analysis of gastrostomy tube dependence in patients with oropharyngeal cancer treated with definitive intensity-modulated radiotherapy. Cancer 121:294-301, 2015
- 73. Kraaijenga SA, Oskam IM, van Son RJ, et al: Assessment of voice, speech, and related quality of life in advanced head and neck cancer patients 10-years+ after chemoradiotherapy. Oral Oncol 55:24-30, 2016
- 74. Hutcheson KA, Nurgalieva Z, Zhao H, et al: Two-year prevalence of dysphagia and related outcomes in head and neck cancer survivors: An updated SEER-Medicare analysis. Head Neck 41:479-487, 2019
- 75. Arslan HH, Ahmadov A, Cebeci S, et al: Life Priorities in Head and Neck Cancer Patients Between Ages of 45 to 65. J Craniofac Surg 27:e398-401, 2016
- 76. Hunter KU, Schipper M, Feng FY, et al: Toxicities affecting quality of life after chemo-IMRT of oropharyngeal cancer: prospective study of patient-reported, observer-rated, and objective outcomes. Int J Radiat Oncol Biol Phys 85:935-40, 2013
- 77. Rogers LQ, Courneya KS, Robbins KT, et al: Physical activity correlates and barriers in head and neck cancer patients. Support. Care Cancer 16:19-27, 2008
- Webb J, Foster J, Poulter E: Increasing the frequency of physical activity very brief advice for cancer patients. Development of an intervention using the behaviour change wheel. Public Health 133:45-56, 2016
- Caspersen CJ, Powell KE, Christenson GM: Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. Public Health Rep 100:126-31, 1985
- 80. Speck RM, Courneya KS, Masse LC, et al: An update of controlled physical activity trials in cancer

survivors: a systematic review and meta-analysis. J. Cancer Surviv 4:87-100, 2010

- Mishra SI, Scherer RW, Geigle PM, et al: Exercise interventions on health-related quality of life for cancer survivors. Cochrane Database Syst Rev:CD007566, 2012
- Ferrer RA, Huedo-Medina TB, Johnson BT, et al: Exercise interventions for cancer survivors: a metaanalysis of quality of life outcomes. Ann. Behav. Med 41:32-47, 2011
- Buffart LM, Sweegers MG, May AM, et al: Targeting Exercise Interventions to Patients With Cancer in Need: An Individual Patient Data Meta-Analysis. J Natl Cancer Inst 110:1190-1200, 2018
- 84. Sweegers MG, Altenburg TM, Brug J, et al: Effects and moderators of exercise on muscle strength, muscle function and aerobic fitness in patients with cancer: a meta-analysis of individual patient data. Br J Sports Med, 2018
- 85. Buffart LM, Kalter J, Sweegers MG, et al: Effects and moderators of exercise on quality of life and physical function in patients with cancer: An individual patient data meta-analysis of 34 RCTs. Cancer Treat Rev 52:91-104, 2017
- Holmes MD, Chen WY, Feskanich D, et al: Physical activity and survival after breast cancer diagnosis. JAMA 293:2479-2486, 2005
- Kenfield SA, Stampfer MJ, Giovannucci E, et al: Physical activity and survival after prostate cancer diagnosis in the health professionals follow-up study. J. Clin. Oncol 29:726-732, 2011
- Meyerhardt JA, Giovannucci EL, Ogino S, et al: Physical activity and male colorectal cancer survival. Arch. Intern. Med 169:2102-2108, 2009
- Schmitz KH: Exercise for secondary prevention of breast cancer: moving from evidence to changing clinical practice. Cancer Prev. Res. (Phila) 4:476-480, 2011
- Richman EL, Kenfield SA, Stampfer MJ, et al: Physical activity after diagnosis and risk of prostate cancer progression: data from the cancer of the prostate strategic urologic research endeavor. Cancer Res 71:3889-3895, 2011
- Capozzi LC, Nishimura KC, McNeely ML, et al: The impact of physical activity on health-related fitness and quality of life for patients with head and neck cancer: a systematic review. Br J Sports Med 50:325-38, 2016
- 92. Sammut L, Fraser LR, Ward MJ, et al: Participation in sport and physical activity in head and neck cancer survivors: associations with quality of life. Clin Otolaryngol 41:241-8, 2016
- Duffy SA, Khan MJ, Ronis DL, et al: Health behaviors of head and neck cancer patients the first year after diagnosis. Head Neck 30:93-102, 2008
- 94. Rogers LQ, Courneya KS, Robbins KT, et al: Physical activity and quality of life in head and neck cancer survivors. Support. Care Cancer 14:1012-1019, 2006
- 95. Silver HJ, Dietrich MS, Murphy BA: Changes in body mass, energy balance, physical function, and inflammatory state in patients with locally advanced head and neck cancer treated with concurrent chemoradiation after low-dose induction chemotherapy. Head Neck 29:893-900, 2007



CHAPTER 2

THE VALUE OF QUALITY OF LIFE QUESTIONNAIRES IN HEAD AND NECK CANCER

I.M. Verdonck-de Leeuw A.J. van Nieuwenhuizen C.R. Leemans

Current Opinion in Otolaryngology & Head Neck Surgery (2012) 20:142-147

ABSTRACT

Purpose of review. To review recent literature on health related quality of life (HRQOL) in head and neck cancer (HNC) patients.

Recent findings. HRQOL is an important part of well-designed clinical trials. HRQOL seems an independent predictor of survival, but this association may be influenced by various cancer-related, personal, biological, and psychobehavioural, physical, lifestyle-related, and social factors. Less is known about the course of HRQOL over time and about the same above mentioned possible factors associated with (change in) HRQOL of HNC patients. Symptom management and psychosocial care may be beneficial for HNC patients to improve HRQOL, but more randomised controlled trials are needed. Studies on HRQOL in HNC are most often based on cross-sectional designs. The variability in outcome measures hampers the generalizability of the results of these studies. Information on HRQOL of caregivers is scarce.

Summary. Better information on all aspects of the course of HRQOL from diagnosis and treatment to long-term survivorship or death is highly needed in both patients and their caregivers enabling reliable and valid predictive modelling. More evidence of efficacy of (new) treatment options, symptom management, and psychosocial care is needed, also in the context of increasing long-term survival and growing attention for **cancer survivorship**.

INTRODUCTION

Research during the past decades has shown that head and neck cancer (HNC) and its treatment has a specific impact on health related quality of life (HRQOL) compared to other cancer types. In addition to general complaints such as pain and fatigue, HNC patients are often confronted with oral dysfunction, voice, speech and swallowing problems and related social withdrawal and emotional distress. HROOL is typically assessed by patient reported outcomes such as questionnaires.¹ The past years have shown an improvement in technical, surgical, and medical possibilities available to optimize the functional outcomes of HNC patients and thereby improve HRQOL. Examples are laser surgery, surgical voice restoration after laryngectomy, and novel reconstruction techniques. Growing evidence indicates that the delivery of radiotherapy (RT) alone or with concomitant chemotherapy (CT) yields similar tumour control and survival compared to surgery in HNC patients but possibly to the expense of increased morbidity, such as swallowing, speech, and shoulder dysfunction and hearing problems. To provide individualized supportive care it is clear that it is important to monitor quality of life in a structured manner in clinical practice. Supportive care in cancer is the prevention and management of adverse effects of cancer and its treatment including speech and swallowing rehabilitation, physiotherapy, nutritional support, and psychosocial care. The introduction of new treatments and supportive care options aiming at improving HRQOL are promising but the scientific evidence remains to be determined.

Overview of recent literature

As a first step of this review, a literature search was performed in PubMed on "quality of life", "questionnaires" and "head and neck cancer" published between January 2010 and September 2011, yielding 160 hits. Based on the following exclusion criteria: patient population other than HNC (n=19), oesophageal cancer (n=16), outcome measure other than patient reported outcome (n=7), study on validation of questionnaire (n=7), and study on small (n<40) patient populations (n=20), 69 studies were excluded. The majority of the remaining 91 studies focussed on patients and only 2 studies on caregivers. There were 3 studies investigating the association between HRQOL and **survival**.

Regarding study design, a cross-sectional or retrospective design was used in 65 studies, a prospective cohort design in 19 studies, and a randomized controlled trial (RCT) or prospective cross-over trial design in 7 studies. Without defrauding other study designs that may be useful for specific research questions, prospective cohort studies and RCTs are highly needed to obtain reliable and valid insights into HRQOL of HNC patients after various treatment modalities during the whole cancer trajectory

from diagnosis to long-term survival or death.

It is striking that among the remaining 93 studies more than 60 different questionnaires were applied. Regarding HNC-specific HRQOL for instance, 4 different questionnaires were used: European Organisation for Research and Treatment of Cancer, Head and Neck Module (EORTC QLQ-HN35), Functional Assessment of Cancer Treatment Questionnaire, Head and Neck Module (FACT-HN), Head and Neck Specific Needs Questionnaire (HNCNQ), and Head and Neck QOL Questionnaire (HNQOLQ). Focussing on symptom specific HROQL, 3 different questionnaires were counted targeting swallowing (Swallowing Quality of Life Questionnaire (SWAL-QOL), MD Anderson Dysphagia Index (MDADI), Sidney Swallow Questionnaire (SSQ)). Kanatas and Rogers² recently performed a systematic review on patient reported outcomes in HNC and also concluded that there is a large variety of validated questionnaires available. Their summary is useful when selecting a questionnaire for research projects or clinical practice.

The following sections will review the value of HRQOL as part of recent clinical trials, the association between HRQOL and survival, the prognostic factors associated with HRQOL, and HRQOL in relation to supportive care in both patients and caregivers. Some of the studies from the above mentioned first literature search are selected that fit well in this review.

HRQOL in clinical trials

Including HRQOL to examine the impact of treatment on daily life is considered part of well-designed RCTs. For example, Mesia et al.3 reported on HRQOL as part of a phase III trial that demonstrated that cetuximab improves survival when added to platinumbased CT for recurrent and/or metastatic HNC. Of 442 patients randomly assigned, 291 patients completed at least one evaluable questionnaire (compliance 65%). They concluded that adding cetuximab to platinum-fluorouracil does not adversely affect HRQOL of patients with recurrent and/or metastatic HNC. Van Herpen et al.⁴ reported on HRQOL as part of a phase III study in 358 patients with unresectable locoregionally advanced HNC showing an improved progression-free and overall survival with less toxicity when docetaxel was added to cisplatin and 5-fluorouracil for induction and given before RT. Compliance to HRQOL assessments was 97% at baseline, but dropped to 54% by 6 months. They concluded that induction CT with TPF before RT not only improves survival and reduces toxicity compared with PF but also seems to improve global HRQOL. Information on long-term HRQOL and late effects is scarce, since HRQOL assessment in many studies (as the above mentioned) is limited to the first year after treatment. Ackerstaf et al.5 reported on the 5-years results of HRQOL among 71 survivors in a randomized phase III trial, assessing intra-arterial versus standard intravenous CRT for inoperable stage IV HNC. No significant differences between 1

and 5 years follow-up were observed, except for "dry mouth" (gradually improving). At 1 year follow-up, survivors reported lower fatigue levels, better voice, and swallowing than patients who could not complete all subsequent follow-up questionnaires.

These examples of recent trials stress the importance of including HRQOL in clinical trials not only to provide evidence on the (positive or negative) impact of new treatment options on HRQOL, but also to enable early supportive care to manage possible adverse events. However, the often low compliance hampers drawing firm conclusions. Random missing data may be handled by organising patient participation in clinical trials in a more structured way to enhance compliance. Advanced statistical analyses including adjustment for survivorship are warranted to analyze non-random missing data (caused by death).

The association between HRQOL and survival

Regarding the association between HRQOL and survival in cancer patients (all cancer types), Montazeri⁶ performed a review on 104 citations. With few exceptions, the findings showed that HRQOL or some aspects of HRQOL are significant independent predictors of survival. Global quality of life, functioning domains, and symptoms such as appetite loss, fatigue, and pain, were the most important indicators, individually or in combination, for predicting survival in cancer patients after adjusting for one or more demographic and known clinical prognostic factors.

In the past 2 years, 3 new studies were published investigating this association in HNC patients. Osthus et al.⁷ reported that overall survival (mean 75 months) was predicted by HRQOL after treatment in a cohort of 139 HNC patients. Global quality of life, fatigue, dyspnoea, and insomnia predicted overall survival independently from outcome measures as gender, age, neuroticism, coping style, alcohol consumption, smoking status, heart/lung disease, time between diagnosis and inclusion, tumour node metastasis stage, and tumour site. Fang et al.8 examined the prognostic value of pre-treatment HRQOL on locoregional control, distant metastasis-free survival (DMFS), and overall survival (OS) in 347 patients treated by RT for nasopharyngeal carcinoma. Among various HRQOL variables that significantly predicted OS and DMFS, physical functioning was the most powerful predictor. Oskam et al.9 investigated the association between HRQOL and survival in a cohort of 80 patients with advanced oral or oropharyngeal cancer after microvascular reconstructive surgery and adjuvant RT. Deterioration of global HRQOL 6 months after treatment compared to pre-treatment levels predicted overall and disease specific mortality independently from sociodemographic and clinical parameters. These 3 studies add up to 7 earlier published studies on the association between HRQOL and survival in HNC.⁹ In 8 out of these 10 studies, HRQOL appeared to be an independent predictor of survival. However, study designs differed substantially regarding tumour subsite, treatment

modality, and HRQOL outcome measures. Furthermore, new empirical evidence suggests that tumour and patient related biomarkers, psychosocial functioning, and lifestyle may also be related to HRQOL and survival in cancer patients. For instance, several biomarkers of neuroendocrinological and neuroimmunological function have been suggested to play a role in the association between depression, HRQOL, and **survival**.¹⁰ Neuroimmunological explanations include increased immune responses and increased levels of pro-inflammatory cytokines. IL-6 has also been found to be **associated with fatigue, depression, and cognitive impairments**.¹¹ Therefore, large-scaled cohort studies are needed to investigate the association between HRQOL and survival in HNC patients, in relation to broadly defined possible moderating factors as cancer-related, personal, biological, psychobehavioural, physical, lifestyle-related, and **social determinants**.

Predictive modelling of HRQOL

Previous research has indicated a lack of understanding of the factors that influence HRQOL of HNC patients. Building and testing predictive models of HRQOL contribute to obtain more insight into these factors and may also improve clinical practice and provide directions for future research. In this paragraph some examples of recent prospective studies are reviewed focussing on oral pain, nutrition, swallowing, sleep, depression, and coping style.

Chen et al.¹² focussed on orofacial pain among 72 patients treated with surgery and RT for HNC. They reported that older age, eating difficulty, speech difficulty, and depression were significant predictors of orofacial pain.

Nourissat et al.¹³ identified factors associated with weight loss during RT in patients with stage I or II HNC as part of a phase III chemoprevention trial. A total of 540 patients were weighted before and after RT. Eight factors were associated with a greater weight loss: all HNC sites other than the glottic larynx, TNM stage II disease, higher pre-RT body weight, dysphagia before RT, higher mucosa adverse effect of RT, lower dietary energy intake during RT, and worse scores regarding digestion and constipation.

Wilson et al.¹⁴ investigated dysphagia before and after RT or CRT in 167 HNC patients (mostly laryngopharyngeal cancer). Treatment intensity (higher RT dose), younger age, and pre-treatment swallowing problems predicted long-term dysphagia.

Christianen et al.¹⁵ investigated in a group of 354 HNC patients treated with RT or CRT which dose volume histogram parameters and pre-treatment factors predict physician-rated and patient-rated dysphagia at 6 months after treatment. A model based on the mean dose to the superior pharyngeal constrictor muscle and mean dose to the supraglottic larynx was most predictive for physician rated swallowing dysfunction. Regarding patient reported swallowing outcome, models were different and dependent on food intake. Shuman et al.¹⁶ investigated predictors of sleep quality among 457 HNC patients 1 year after diagnosis. Pain, xerostomia, depression, presence

28

of a tracheotomy tube, comorbidities, and younger age were predictors of poor sleep 1 year after diagnosis. Smoking, problem drinking, and female sex were marginally significant (p < .09). Type of treatment (surgery, RT and/or CT), primary tumour site, and cancer stage were not significantly associated with 1-year sleep scores. Scharloo et al.¹⁷ investigated the contribution of illness cognitions to the prediction of HRQOL 2 years after diagnosis among 177 HNC patients. Patients' own implicit common sense beliefs about their illness at baseline added little but significantly to the prediction of HRQOL after 2 years. Less belief in own behaviour causing the illness predicted better functioning and better global health. Strong illness identity beliefs predicted worse functioning and worse global health. Negative perceptions about the duration of the illness (chronic timeline beliefs) and more negative perceived consequences also predicted worse HRQOL.

Howren et al.¹⁸ investigated whether pre-treatment depression contributes to predicting HRQOL among 306 HNC patients. Depressive symptoms before cancer treatment significantly predicted (worse) HRQOL at 3 and 12 month follow-up across the 4 HNC-specific domains of speech, eating, aesthetics, and social disruption, independently from age, gender, marital status, cancer site, stage of disease, alcohol and tobacco use, comorbidity status, and pre-treatment HRQOL.

Previous and recent researches on HRQOL show considerable variation between patients: some patients are at risk for poor HRQOL, while others are protected. It is highly relevant to understand risk factors and protective resources that predict HRQOL. This knowledge enables clinicians to identify HNC patients who are at risk for poor HRQOL and enabling referral to supportive care in an earlier stage. Furthermore, understanding of risk-factors and protective resources allows the development of interventions or treatment innovation aiming at improving HRQOL. As already argued in the paragraph on the association between HRQOL and survival, large-scaled studies are also needed to build prognostic models targeting HRQOL. Among others, age, tumour and treatment related characteristics, comorbidity, depression, and coping style, are potential mediators of HRQOL outcome. Some of these factors seem to be treatable. In the reviewed studies the importance was stressed of supportive care to manage symptoms and improve psychosocial functioning that may contribute improving HRQOL after HNC treatment. Predictive modelling helps to identify patients who are most likely to benefit from which supportive care options.

HRQOL and supportive care

There is growing interest in using patient-reported outcomes (PRO's) to screen for physical and psychosocial problems and the need for supportive care in routine clinical practice and several studies have shown that using PRO's in clinical practice facilitates communication about quality of life between doctors and patients. There

29

is less evidence that this approach may affect patient outcome or improve quality of life and additional efforts are needed to enhance the effect of screening, such as advance palliative care planning, survivorship care plans, and improving evidence based supportive care options. In this paragraph, the focus is on studies providing evidence regarding symptom management and supportive care. Bien et al.¹⁹ reported on a RCT (n=80) investigating the effect of Heat and Moisture Exchanger (HME) use on pulmonary symptoms and quality of life aspects in laryngectomized patients. This study confirms the results of previous studies, showing that pulmonary symptoms decrease significantly with HME use and that related aspects such as speech and sleeping tend to improve.

Nibu et al.²⁰ performed a longitudinal study to assess the impact of rehabilitation and surgical modification on postoperative HRQOL. Patients who had undergone neck dissection (n=224) underwent a rehabilitation program designed for neck dissection. They reported that resection of the sternocleidomastoid muscle (SCM) and spinal accessory nerve (SAN) resulted in shoulder drop. Selective neck dissections and preservation of the SAN and SCM significantly reduced various sensory symptoms of the neck, such as stiffness, pain, numbness, and constriction, and improved shoulder function. Postoperative rehabilitation had a significant effect on arm abduction ability, particularly when the SCM and SAN were resected.

Regarding the need for supportive care among 165 oral cancer patients at time of diagnosis, Chen et al.²¹ reported that patients had high supportive care needs, with the highest prevalence of unmet needs in the area of health system and information, followed by psychological care. Precious et al.²² asked HNC 386 patients about the role fulfilled by their carer(s) and their support. Patients identified the main roles of caregivers as providing emotional support (75%), taking them to healthcare appointments (67%), cleaning the home (62%), and shopping for food (59%). About 60% felt that caring was a considerable burden or very hard for their carers. Patients over 65 years of age and those with low socio-emotional functioning were the most likely to need a considerable amount of care and support.

Shuman et al.²³ carried out a study on the perceived quality of care received by HNC patients at the end of their lives. Results of a survey among families of 58 patients who died of HNC revealed that palliative treatments of HNC, death outside of the hospital, and palliative care team involvement all improve the end-of-life experience in this population.

These studies demonstrate the importance of symptom management and psychosocial care not only in patients but also in their caregivers. Supportive care comes within the responsibility of several care providers: surgeons, radiation and medical oncologists, primary care physicians, and various psychosocial and allied health service providers. These care providers often have disposal over only part of the relevant information

and communication between care providers regarding well-being and psychosocial functioning is limited. Because of this fragmentation of continuity of care, patients and families often complain about feelings of powerlessness and a lack of guidance.

There is growing interest in using quality of life questionnaires in clinical practice to monitor quality of life and to facilitate communication about quality of life between doctors and patients. There is less evidence that this approach may affect patient outcome or improve quality of life and it is argued that additional efforts are needed to enhance the effect of screening.²⁴ For example, the presence of a professional care navigator leads to higher patient satisfaction, shorter duration of hospitalization, fewer cancer-related problems, better emotional quality of life, and patient empowerment.²⁵

CONCLUSION

In summary, HRQOL seems a significant predictor of survival. However, this association may be influenced by various cancer-related, personal, biological, psychobehavioural, physical, lifestyle-related, and social factors, which may also interact with each other. These factors are also of influence in predictive modelling of HRQOL as such. Previous research on HRQOL in HNC was often based on cross-sectional designs or often lacked adequate multivariate statistical analyses to adjust for confounding or moderating factors. The large variety of HRQOL questionnaires and variability in predictors between studies and the highly selected inclusion criteria from the RCTs on which most evidence is based, seriously hampers the generalizability of the results of previous studies. In addition, less is known about the course of HRQOL over time and about the (broad range of) possible factors associated with change in (long-term) HRQOL of HNC patients. Insight in the incidence and prevalence of late effects of HNC and treatment and in their associated risk factors is scarce. Better information on all aspects of HRQOL is therefore highly needed as is evidence regarding cost-effectiveness of (new) treatment options, symptom management, and psychosocial care, also in the context of increasing long-term survival and growing attention for cancer survivorship.

Key points:

- HRQOL is an important outcome measure in clinical trials.
- HRQOL is associated with survival, but large-scaled cohort studies are needed investigating possible moderators and mediators such as cancer-related, personal, biological, psychobehavioural, physical, lifestyle-related, and social factors.
- Predictive models of HRQOL are highly needed.
- Organising supportive care by appointing a care coordinator and implementing a survivorship care plan or advanced palliative care planning is beneficial for HNC cancer patients and their caregivers and may improve their HRQOL.
- More evidence is needed on cost-effectiveness of supportive care.

REFERENCES

- Rogers SN, Forgie S, Lowe D et al. Development of the International Classification of Functioning, Disability and Health as a brief head and neck cancer patient questionnaire. Int J Oral Maxillofac Surg. 2010 Oct;39(10):975-82.
- Kanatas and Rogers. A systematic review of patient self-completed questionnaires suitable for oral and maxillofacial surgery. Br. J. Oral Maxillofac Surg. 2010 Dec;48(8):579-90.
- Mesia R, Rivera F, Kawcki A et al. Quality of life of patients receiving platinum-based chemotherapy plus cetuximab first line for recurrent and/or metastatic squamous cell carcinoma of the head and neck. Ann Oncol. 2010 Oct;21(10):1967-73.
- 4. Van Herpen CM, Mauer ME, Mesia R et al. Short-term health-related quality of life and symptom control with docetaxel, cisplatin, 5-fluorouracil and cisplatin (TPF), 5-fluorouracil (PF) for induction in unresectable locoregionally advanced head and neck cancer patients (EORTC 24971/TAX 323). Br J Cancer. 2010 Oct 12;103(8):1173-81.
- 5. Ackerstaff AH, Rasch CR, Balm AJ. Five-year quality of life results of the randomized clinical phase III (RADPLAT) trial, comparing concomitant intra-arterial versus intravenous chemoradiotherapy in locally advanced head and neck cancer. Head Neck. 2011 Aug 4. doi: 10.1002/hed.21851. Study on the results 5 years after CRT on HRQOL in HNC survivors. Prospective long-term studies focusing on HRQOL after CRT are scarce.
- 6. Montazeri A. Quality of life data as prognostic indicators of survival in cancer patients: an overview of the literature from 1982 to 2008. Health Qual Life Outcomes. **2009** Dec 23;7:102.
- Osthus AA, Aarstad AK, Olofsson J, Aarstad HJ. Head and neck specific Health Related Quality of Life scores predict subsequent survival in successfully treated head and neck cancer patients: A prospective cohort study. Oral Oncol. 2011 Aug 17.
- 8. Fang FM, Tsai WL, Chien CY et al. Pretreatment quality of life as a predictor of distant metastasis and survival for patients with nasopharyngeal carcinoma. J Clin Oncol. 2010 Oct 1;28(28):4384-9.
- 9. Oskam IM, Verdonck-de Leeuw IM, Aaronson NK et al. Quality of life as predictor of survival: a prospective study on patients treated with combined surgery and radiotherapy for advanced oral and oropharyngeal cancer. Radiother Oncol. 2010 Nov;97(2):258-62.
- 10. Reich M. Depression and cancer: recent data on clinical issues, research challenges and treatment approaches. Curr Opin Oncol 2008 Jul;20(4):353-9.
- 11. Seruga B, Zhang H, Bernstein LJ, Tannock IF. Cytokines and their relationship to the symptoms and outcome of cancer. Nat Rev Cancer 2008 Nov;8(11):887-99.
- 12. Chen SC, Liao CT, Chang JT. Orofacial pain and predictors in oral squamous cell carcinoma patients receiving treatment. Oral Oncol. 2011 Feb;47(2):131-5.
- 13. Nourissat A, Bairati I, Fortin A et al. Factors associated with weight loss during radiotherapy in patients with stage I or II head and neck cancer. Support Care Cancer. 2011 Mar 20.
- 14. Wilson JA, Carding PN, Patterson JM. Dysphagia after Nonsurgical Head and Neck Cancer Treatment: Patients' Perspectives. Otolaryngol Head Neck Surg. 2011 Jul 11.
- 15. Christianen ME, Schilstra C, Beetz I et al. Predictive modelling for swallowing dysfunction after primary (chemo)radiation: Results of a prospective observational study. Radiother Oncol. 2011 Sep 8.
- 16. Shuman AG, Duffy SA, Ronis DL et al. Predictors of poor sleep quality among head and neck cancer patients. Laryngoscope. 2010 Jun;120(6):1166-72.
- 17. Scharloo M, Baatenburg de Jong RJ, Langeveld TP et al. Illness cognitions in head and neck squamous cell carcinoma: predicting quality of life outcome. Support Care Cancer. 2010 Sep;18(9):1137-45.

- 18. Howren MB, Christensen AJ, Karnell LH, Funk GF. Health-related quality of life in head and neck cancer survivors: impact of pretreatment depressive symptoms. Health Psychol. 2010 Jan;29(1):65-71.
- Bień S, Okła S, van As-Brooks CJ, Ackerstaff AH. The effect of a Heat and Moisture Exchanger (Provox HME) on pulmonary protection after total laryngectomy: a randomized controlled study. Eur Arch Otorhinolaryngol. 2010 Mar;267(3):429-35.
- 20. Nibu K, Ebihara Y, Ebihara M et al. Quality of life after neck dissection: a multicenter longitudinal study by the Japanese Clinical Study Group on Standardization of Treatment for Lymph Node Metastasis of Head and Neck Cancer. Int J Clin Oncol. 2010 Feb;15(1):33-8.
- 21. Chen SC, Yu WP, Chu TL et al. Prevalence and correlates of supportive care needs in oral cancer patients with and without anxiety during the diagnostic period. Cancer Nurs. 2010 Jul-Aug;33(4):280-9.
- 22. Precious E, Haran S, Lowe D, Rogers SN. Head and neck cancer patients' perspective of carer burden. Br J Oral Maxillofac Surg. 2011 Jun 1.
- Shuman AG, Yang Y, Taylor JM, Prince ME. End-of-life care among head and neck cancer patients. Otolaryngol Head Neck Surg. 2011 May;144(5):733-9.
- 24. Snyder CF, Aaronson NK. Use of patient-reported outcomes in clinical practice. Lancet 2009;374:369-70.
- 25. Fillion L, de Serres M, Cook S, Goupil RL, Bairati I, Doll R. Professional patient navigation in head and neck cancer. Semin Oncol Nurs. 2009;25(3):212-21.



- CHAPTER 3

THE ASSOCIATION BETWEEN HEALTH-RELATED QUALITY OF LIFE AND SURVIVAL IN PATIENTS WITH HEAD AND NECK CANCER: A SYSTEMATIC REVIEW

A.J. van Nieuwenhuizen L.M. Buffart J. Brug C.R. Leemans I.M. Verdonck-de Leeuw

Oral Oncology (2015) 51(1):1-11.

ABSTRACT

Objective. To study the association between health-related quality of life (HRQoL) and survival in patients with head and neck cancer (HNC) based on a systematic review of prospective studies.

Methods. A systematic search was conducted in four electronic bibliographic databases. We included studies published up to January 2014, providing data on HRQoL and survival, and the association between HRQoL and survival, among HNC patients. Two researchers independently performed a quality rating. A best evidence synthesis was applied to draw conclusions.

Results. Nineteen studies were included. Twelve studies included all subscales of a HRQoL questionnaire and seven studies focused on specific subscales. The mean quality score was $72 \pm 17\%$ and 58% of the studies were of high quality. According to the best evidence synthesis, we found strong evidence for a positive association between pre-treatment physical functioning and survival and between change in global QoL from pre-treatment to 6 months after treatment and survival. Due to inconsistent findings, we found insufficient evidence for an association with survival of other HRQoL domains, including role, emotional, cognitive and social functioning, mental health and well-being.

Discussion. This systematic review showed that higher levels of pre-treatment physical function and change in global QoL from pre-treatment to 6 months after treatment are associated with survival in patients with HNC. Future high quality studies with a longitudinal design are needed to examine the complex associations between HRQoL and survival.
INTRODUCTION

Patient reported-outcomes are increasingly used in clinical trials in cancer patients besides the traditional outcome measures as tumor control, overall survival, morbidity **and complications**.¹⁻⁵ This has led to an increased understanding about the course of health-related quality of life (HRQoL) in patients with cancer. It is well known that patients have to deal with various physical and psychosocial side effects of cancer and cancer treatment, including reduced physical fitness and function, reduced self-esteem, increased emotional distress and fatigue, negatively affecting HRQoL.¹⁻⁶ Furthermore, there is evidence of a positive association between HRQoL and survival in patients with cancer. Previous reviews and meta-analyses in populations with different types of cancer showed that lower pre-treatment HRQoL was associated with reduced **surviva**l.⁷⁻¹⁰ Recently, in a pooled analysis, Quinten et al.¹⁰ examined the association between pre-treatment HRQoL and survival separately for 11 different cancers. She concluded that for each cancer, at least one HRQoL domain provided prognostic information additional to clinical and demographic variables such as age, performance **status and metastasis, increasing the accuracy of survival prognosis**.¹⁰

Little is known about the association between HRQoL and survival in head and neck cancer patients (HNC). Investigators have found that HNC and its intensive treatment has a distinct impact on HRQoL different from other cancers. In addition to physical and psychosocial problems that patients with all cancers are facing, patients with HNC are often confronted with oral dysfunction, swallowing and speech problems.¹¹⁻¹⁸ Previous studies showed a decline in general health, mental health, physical function, appearance, employment, and social functioning during and immediately after treatment for HNC.^{11;12;14;15;17-27} Also, many HNC survivors continue to suffer from various disease and treatment related physical and psychosocial problems for many years after treatment.^{11;12;18;19} Two other prospective cohort studies reported that HRQoL ten years after diagnosis was significantly lower than HRQoL before treatment.^{21;28}

In 2008, Mehanna et al.²⁹ published a systematic review summarizing the available studies examining the association between a broad range of psychosocial factors and **survival in patients with HNC.** They reported that psychosocial complaints, physical self-efficacy and higher physical functioning were associated with increased survival. In addition, they found that global QoL one year after diagnosis was significantly associated with survival, but this was not the case for global QoL at diagnosis.²⁹ The review of Montazeri et al.⁸ revealed significant associations between HRQoL and survival for most cancers, but findings of studies on HNC patients were inconsistent and based on a limited number of studies. More studies evaluating the association between HRQoL and survival in patients with HNC have been published since the two previous reviews, which warrants a new systematic review of the present evidence.

Furthermore these previous reviews have not applied a best evidence synthesis to summarize the data, which hampers interpretability. Therefore, the objective of this systematic review is to study the association between HRQoL and survival in patients with HNC with profound methodology, and to identify which HRQoL domains are most strongly associated with survival, adjusted for important clinical, demographic and lifestyle-related factors.

MATERIALS AND METHODS:

Selection of studies

A literature search up to 23rd of January 2014 was conducted in four electronic bibliographic databases (PubMed, EMBASE, PsychINFO and CINAHL) in collaboration with a librarian. In order to identify all relevant papers, we used keywords, Mesh terms and free terms for the following search terms including "Head and neck neoplasm", "Quality of life" or "patient reported outcome" and "survival" or "prognostic". The complete search strategy of the literature search is available upon request. The reference lists of all selected papers were screened for additional relevant papers.

Study inclusion criteria

Studies were included if they (1) included patients with HNC, (2) had a prospective study design, (3) assessed HRQoL with a standardized questionnaire, (4) measured mortality and/or survival, (5) analyzed the association between HRQoL and survival, and (6) were full text articles written in English, German or Dutch. Studies focusing on patients with cancer of the thyroid, esophagus, skin or skull base were excluded. We also excluded studies that were part of a supportive care intervention aiming to improve the HRQoL of patients.

Selection procedure and quality assessment

Title and abstract of the references were reviewed to exclude articles out of scope (AN). Full texts of potentially relevant articles were assessed for eligibility by two independent reviewers (AN and LB). Two authors (AN and LB) independently assessed the methodological quality of the included articles. Disagreements were discussed and resolved. The methodological quality of the included studies was scored on a methodological criteria list based on the criteria list suggested by Hayden et al,³⁰ who developed the list after an extensive review and critical appraisal of systematic reviews of prognostic studies supplemented by recent methodological studies. The list comprises 6 potential biases for prognostic studies i.e. study participation, study attrition, prognostic factors, outcome measurement, confounding measurements and account and analysis. Because most of the included studies focused on HRQoL

assessed at only one time point, we excluded the item study attrition. Table 1 presents the methodological scoring list used in the present study. We defined clinical variables (tumor location and stage), demographic variables (age, gender), and comorbidity and lifestyle factors (smoking and alcohol intake) as important confounders that should be accounted for in the analysis, since they are important prognostic predictors for survival. The list contains 11 criteria which can be scored positive (1) or negative (0). A positive score was given when sufficient and adequate information on the criterion was available. A negative score indicates that the paper provided no or insufficient information about this criterion. For the last item in the quality assessment (presentation of points estimates and measures of variability) we gave 0.5 points if all data from the univariate analysis were present but not from multivariate analysis. If the study referred to another publication containing relevant information about the same study, we retrieved the additional publication to score the criterion of concern. The total score is the sum of all criteria that were scored positively, with a maximum score of 11 points. For each study we calculated the percentage of items that a study scored positively on methodological quality. A study was considered of 'high quality' if the quality score was $\geq 75\%$.³¹

Data extraction

The following data were extracted (AN): first author, year of publication, number of patients included, cancer location and stage, assessment of HRQoL, subscales included in the analysis, assessment and period of survival, univariate and multivariate association between HRQoL and overall survival (including hazard ratios with 95% confidence intervals (CI) and p-values), and covariates.

Level of scientific evidence

To synthesize the methodological quality of the studies and to be able to draw conclusions regarding the relationship between HRQoL and survival, we applied a best-evidence synthesis.^{31:32} The rating system consists of three levels and takes into account the number, methodological quality and consistency of outcomes of these studies as follows: (A) Strong evidence, provided by generally consistent findings in multiple (\geq 2) high quality studies, (B)

We considered results to be consistent when $\geq 75\%$ of the studies showed results in the same direction, which was defined according to significance (p<0.05). If two or more studies were of high methodological quality, we disregarded the studies of low methodological quality in the evidence synthesis. We included the multivariate results in the best evidence synthesis, and it was applied separately for each time point of HRQoL measurements (e.g. pre-treatment, post-treatment), and for different HRQoL domains (e.g. global QoL, functioning, well-being).

	Stur A	в	C	D	E Pro
	dy participation The sampling frame and recruit- ment are adequately described (setting and geographic location)	Inclusion and exclusion criteria	are adequately described The baseline study sample (participants) are adequately described for key characteristics	(age, gender, tumor location and stage) There is adequate participation in the study by eligible patients (> 75%) or differences between responders and non-responders	Is adequately described spective factor measurement HRQoL is assessed with a valid measurement instrument
Coyne-2007 [38]	-	1	1	1	1
Curran-2007 [49]	-	1	1	Т	1
De Graeff-2001 [40]	-	1	1	Т	1
Fang-2010 [43]	-	1	1	1	1
Fang-2003 [50]	-	1	1	1	Т
65] 7002-nonginÐ	-	1	1	0	1
[47] Xarvonen-2008 [47]	-	1	1	0	1
Lango-2013 [41]	-	1	1	1	Т
Mehanna-2006 [56]	0	0	1	1	Т
Meyer-2009 [44]	-	1	1	0	1
Nordgren-2007 [53]	-	1	1	0	1
Nordgren-2006 [52]	-	1	1	0	1
Nordgren-2003 [51]	-	1	1	0	1
Oskam-2010 [45]	-	1	1	1	1
[84] £102-2013[48]	-	1	1	1	Т
[64] 1102-suftsO	-	1	1	1	1
[95] 7002-iupibbi8	-	1	1	1	т
Tarsitano – 2012 [42]	-	1	1	0	т
[46] 1102-noslqmodT	-	1	1	0	Т
Proportion of posicine	95	95	100	58	100

	D L	C oi	Ξ	Ani	П	ſ	K
	come measurement Survival is clearly defined and measured appropriatelya	founding measurement account All important confounders are measured (key demographic, clinical factors, lifestyle factors	and comorbidity)b Important potential confounders are accounted for in the analysis (appropriated adjustment)	lysis	There is sufficient presentation of HRQoL, mean (SD)	There is sufficient presentation of	The strategy for model building is clearly described and appropriate
Coyne-2007 [38]	0	0	1		f o	f 1	6 3
Curran-2007 [49]	0	0	1		0	1	1
De Graeff-2001 [40]	0	0	1		1	1	1
Fang-2010 [43]	0	1	0		1	1	1
Fang-2003 [50]	0	0	1		1	0	1
Grignon-2007 [55]	0	0	0		0	1	1
Karvonen-2008 [47]	1	1	Т		1	1	1
Lango-2013 [41]	0	1	1		1	1	1
Mehanna-2006 [56]	1	0	1		0	1	1
Meyer-2009 [44]	Т	1	0		1	1	1
Nordgren-2007 [53]	0	0	1		0	1	0
Nordgren-2006 [52]	0	0	1		0	1	ο
Nordgren-2003 [51]	0	0	0		0	1	0
Oskam-2010 [45]	0	0	Т		0	1	1
[84] £102-sufteO	1	1	ц		1	1	1
[94] 1102-suftsO	0	1	ц		1	1	1
[6£] 7002-iupibbi8	0	0	T.		1	1	1
Tarsitano – 2012 [42]	0	1	0		1	1	1
Thompison-2011 [54]	0	0			0	1	T
Proportion of positive scores (%)	21	37	74		53	95	84

Proportion of positive scores (%)	53	1	72
[44] Thomplson-2011	0	7	58
Tarsitano – 2012 [42]	1	6	75
[95] 7002-iupibbi8	0	6	75
[64] 1102-2011[46]	ч	11	92
[84] £102-2013[48]	ч	12	100
Oskam-2010 [45]	ч	6	75
Nordgren-2003 [51]	0	5	42
Nordgren-2006 [52]	0	9	50
Nordgren-2007 [53]	0	9	50
Meyer-2009 [44]	н	10	83
Mehanna-2006 [56]	н	7	58
Lango-2013 [41]	н	11	92
Karvonen-2008 [47]	н	11	92
Grignon-2007 [55]	0	9	50
Fang-2003 [50]	0.5	8.5	Ľ
Fang-2010 [43]	ч	10	83
De Graeff-2001 [40]	0	6	75
Curran-2007 [49]	0	œ	67
Coyne-2007 [38]	ч	6	75
	L Point estimate and measures of variability (HR and 95% CI) were presented(0.5 points if all data were presented of univariate analysis, 1 point for multivariate analysis)	Total quality score, points	Percentage of maximum score (%)

available or findings were inconsistent in multiple (≥ 2) studies. We considered results to be consistent when $\geq 75\%$ of the studies showed results in the same direction, which was defined according to significance (p<0.05). If two or more studies were of high methodological quality, we disregarded the studies of low methodological quality in the evidence synthesis. We included the multivariate results in the best evidence synthesis, and it was applied separately for each time point of HRQoL measurements (e.g. pre-treatment, post-treatment), and for different HRQoL domains (e.g. global QoL, functioning, well-being). Moderate evidence, provided by generally consistent findings in one high quality study and (z1) low quality studies, (C) insufficient evidence, when only one study was

RESULTS

Identification and selection of the literature

After removing duplicates, the literature search yielded 2481 unique articles. For 82 potentially relevant articles, we checked full text (Figure 1). The majority of the studies (n=29) were excluded because they lacked information on survival or they did not assess the association between HRQoL and survival (n=29). Of the articles identified in the database search, 14 met the inclusion criteria. We found five additional articles from the reference lists of the included studies. Consequently 19 studies were included in this review comprising 5207 patients. Two articles^{33:34} included the same patient population and were counted as one population (n=1093).

Quality assessment

Results of the methodological quality assessment are presented in Table 1. The mean quality score was 72±17% (range 42-100%). Eleven studies³³⁻⁴³ were of high quality. Of all studies, 79% had shortcomings in the assessment and definition of survival: methods for survival assessment were unknown for 13 studies, ^{33:34:36-38:40:41:44-49} whereas two studies^{35:50} used medical records and. In addition, 63% of the studies had shortcomings in assessments of important confounders: eleven studies^{33-35:38:39:41:44:45:47:49:51} did not assess comorbidity, and seven^{40:44-48:50} studies did not assess smoking and alcohol consumption. Shortcomings in presenting point estimates and measures of variability were reported in 47% of the studies: Eight studies^{34:35:40:44:46-50} reported only significance levels of the association without presenting point estimates and measures of variability and one study⁴⁵ presented results of the univariate associations but only reported point estimates and measures of variability of associations that were significant in the multivariate analysis. All studies except for one,⁵¹ adequately described baseline characteristics, recruitment, and in- and exclusion criteria of the study sample.

In addition, all studies used a valid questionnaire to assess HRQoL, and one study⁴⁹ used an additional single question on global QoL.



Figure 1. Flowchart of study inclusion

Data extraction

Characteristics of the included studies are presented in Table 2. Sixteen studies^{33-40;43-48;50;51} assessed HRQoL before treatment. One study⁴² assessed the baseline assessment of HRQoL within 2 years after diagnosis, and two studies^{41;49} assessed the HRQoL of patients after a minimal disease free period of 1 year or 2 years. Fifteen studies^{33-37;39-45;49-51} included mixed sites of HNC, Nordgren et al.⁴⁶⁻⁴⁸ published three articles in which the HNC sub-sites oral cavity, pharyngeal and larynx were separately presented (Table 2), and one study³⁸ included patients with nasopharyngeal carcinomas only. Thirteen studies^{35-38;41-43;46-51} included patients with all stages of the tumor, three studies^{33:34;40} included patients with tumor stage II, III and IV, two studies^{44;45} included patients with tumor stage I and II. Twelve studies^{35:37-41;43-48} used the European Organization for Research and Treatment of Cancer Quality of Life Questionnaire Core 30 (EORTC-QLQ-C30) to

assess HRQoL, three studies^{42;49;50} used the Short Form-36 health survey (SF-36), two studies^{33;34} used the Functional Assessment of Cancer Therapy – General (FACT-G), one study⁵¹ the Auckland quality of life questionnaire, one study³⁶ used the EuroQol 5D (EQ5D), and one study⁴⁹ used a self-administered single question on general QoL (a Likert type scaling of overall QoL ranging from 1 (very poor) to 5 (excellent)) in addition to the SF-36.

Pre-treatment HRQoL and the association with survival

Table 3 presents an overview of the associations between survival and HRQoL assessed pre-treatment (top) and during or after treatment (bottom). Pre-treatment global QoL was assessed in thirteen studies, of which six^{35:37-40;43} were of high quality. Pre-treatment physical function was assessed in ten studies of which four^{35:38:39;43} were of high quality. Nine studies assessed pre-treatment emotional functioning, social functioning, role functioning and cognitive functioning, of which four^{35:38:39;43} were of high quality. One high quality study³⁴ included pre-treatment functional, physical, emotional and social well-being and one high quality study³⁶ included pre-treatment HRQoL using the EuroQol5D (EQ5D) questionnaire.⁵² One low quality study⁵⁰ determined pre-treatment mental component summary scores.

Of all high quality studies,75% showed that a higher pre-treatment physical functioning was associated with reduced mortality (HR = 0.64 to 0.99) in the multivariate analysis after adjusting for important clinical, demographic and lifestyle related factors, indicating strong evidence (Table 3). This significant association was supported by 50% of low quality studies.

urvival Assessment Association between QoL and survival Factors adjusted eriod of survival for for	SD) univariate multivariate		HR (95% CI) p-value HR (CI) p-value		nknown Unclear EWB 1.02 (0.99-1.04) 0.09 1.02(0.99-1.04) 0.13 T, N stage, KPS score, cigarette use, age, income, use, age, income,	marital status	3m and Unclear gQoL n.m. <0.001 1.66 0.005 T, N stage, tumor	9.3m PF n.m. <0.001 n.m. n.s. site, KPS, age, sex	RF n.m. n.s. n.a.	EF n.m. n.s. n.a.	CF n.m. n.s. n.a.	SF n.m. n.s. n.a.	1:45m At time of gQoL n.m. n.s. n.m. n.s. Sex, age, marital	(6-62) analysis PF n.m. <0.05 n.m. n.s. status, stage,	urviving survival was RF n.m. n.s. n.m. n.s. growth pattern,	atients checked EF n.m. n.s. n.m. n.s. treatment, radi-	CF n.m. < 0.05 1.9 (1.10-3.26) < 0.05 cality and all QoL	CF nm 2005 nm ne enhevelee emok-	
I SULVIVAI			H		nclear EWB 1.02		nclear gQoL n.m.	PF n.m.	RF n.m.	EF n.m.	CF n.m.	SF n.m.	t time of gQoL n.m.	nalysis PF n.m.	urvival was RF n.m.	necked EF n.m.	CF n.m.	SF n.m.	
Survival A: Period of (mean	(ISH)	(uc+			G unknown U		49m and U	29.3m					Fu:45m A	(36-62) aı	surviving sı	patients ch			
or QoL PRO					l/ FACT0		7 C30						C30						
ncer Tumc e stage				eatment	, II/III , IV		, III/IV	0					C, All	^	с, L				
Patients Ca (n) sit				ssed pre-tre	1093 OC OF HI		419 OF	ΗΗ	Г				208 00	OF	Η				
First author and year of publication	hummand			HRQoL asse	Coynea 2007 [38]		Curran	2007 [49]					de Graeffa	2001 [40]					

Table 2. Con	tinued											
First author and year of	Patients (n)	Cancer site	Tumor stage	QoL PRO	Survival Period	Assessment of survival		Associati	on betwee	en QoL and survival		Factors adjusted for
publication					(mean +SD)			univariate		multivariat	e	
								HR (95% CI)	p-value	HR (CI)	p-value	
Fanga	347	NP	AII	C30	5y; No	Unclear	gOoL	n.m.	< 0.05	0.83 (0.72-0.94)	0.002	Age, sex, smoking,
2010 [43]					survival		PF	n.m.	< 0.05	0.74 (0.64-0.85)	< 0.001	cancer stage
					period		RF	n.m.	< 0.05	0.88 (0.79-0.96)	0.005	
							EF	n.m.	n.s.	0.93 (0.82-1.03)	0.16	
							CF	n.m.	< 0.05	0.89 (0.80-0.99)	0.03	
							\mathbf{SF}	n.m.	< 0.05	0.87 (0.78-0.95)	0.003	
Fang	102	0C,	VI/III	C30	5y; No	Unclear	gOoL	(66.0-86.0) 66.0	0.04	n.m.	n.s.	Stage, KPS, N
2003 [50]		OP,			survival		PF	(66.0-70.0) 99.0	0.03	n.m.	n.s.	status
		HP, L			period		RF	0.99 (0.99-1.00)	0.22		n.a.	
							EF	0.99 (0.98-1.01)	0.27		n.a.	
							CF	0.99 (0.99-1.01)	0.37		n.a.	
							\mathbf{SF}	0.99 (0.98-1.00)	0.14		n.a.	
Grignon	571	0C,	All	SF-36	5y; No	Medical	PCS	n.m.	n.m.	0.97	<0.001	Site, stage, co-
2007 [55]		OP,			survival	record	MCS	n.m.	n.m.	n.m.	0.56	morbidity, mental
		HP, L			period							component SF-36

Factors adjusted for	T and N classifica- tion, performance status, smoking and weight loss	T. N stage, disease stage, age, gender, smoking, alcohol consumption	Cancer stage, cancer site, age, smoking, alcohol, BMI, trial arm assignment
te	г 0.07 п.а. п.а. п.а. п.а. п.а. п.а. п.а.	0.26 <0.001	n.a. < 0.001 0.019 n.a. n.a.
n QoL and surviva multivaria	2.1 (0.94-5.0) 2.4 (1.1-5.6)	0.9 (0.5-1.4) 3.0 (2.1-4.2)	0.87 (0.81-0.94) 0.93 (0.89-0.99)
ion betwee	1.02 0.19 0.67 0.14 0.05 0.07 0.00 0.00 0.005 0.001	0.02	0.46 <0.001 0.04 0.32 0.32 0.89
Associat univariat	0.02 1.9 0.006 2.7 Referent 2.8 8.3 8.3 8.3 10	1.6 (1.10-2.39) 1.5 (1.02-2.22)	0.98 0.85 0.94 1.03 1.03 0.99
	PRHS Mob Sc Act P&D P&D P&D - none - none D&A D&A - none - none - none - sev	GHQ LSS	gQoL PF RF EF CF SF
Assessment of survival	Unclear	Centralized national clinical booking system	Mortality files, death certificates
Survival Period (mean +SD)	Fu:32m (1-71)	10y; Median survival: 6y	5y; Median follow-up: 6.5y
QoL PRO	EQ5D	Auck- land QoL	C30
Tumor stage	II	All	1/П
Cancer site	HNC	HNC	HNC (lar- ynx and oth- ers)
Patients (n)	159	200	540
First author and year of publication	Langoa 2013 [41]	Mehanna 2006 [56]	Meyera 2009 [44]

Table 2. Continued

	Factors adjusted for			Advanced disease	stage (III/IV), co-	morbidity present,	fatigue, pain C30,	pain HN35, sense,	speech, teeth,	opening mouth	wide, dry mouth	Advanced disease	stage (III/IV), no	brachytherapy	given, fatigue,	dyspnea, speech,	coughing, feeling	III	Radiotherapy /	other treatment,	stage I + II, stage	III +IV, HADS	depression,	comorbidity and	speech scale
	_	te	p-value	n.a.	n.s.	n.a.	n.a.	n.s.	n.s.			n.a.	0.002	n.a.	n.a.	n.a.	n.a.		n.s.	< 0.001	n.s.	n.a.	n.a	n.s.	
	n QoL and surviva	multivaria	HR (CI)		n.m.			n.m.	n.m.				0.98(0.97-0.99)						n.m.	0.98 (0.96-0.99)	n.m.			n.m.	
	on betwee		p-value	n.s.	< 0.001	n.s.	n.s.	<0.05	<0.05			n.s.	0.027	n.s.	n.s.	n.s.	n.s.		0.001	< 0.001	< 0.001	n.s.	n.s.	<0.001	
	Associati	univariate	HR (95% CI)	n.m.	0.98 (0.97-0.99)	n.m.	n.m.	n.m.	n.m.			n.m.	0.99 (0.98-1.00)	n.m.	n.m.	n.m.	n.m.		0.98 (0.96-0.99)	0.97 (0.96-0.98)	0.98 (0.97-0.99)	n.m.	n.m.	0.98 (0.97-0.99)	
				gQoL	ΡF	RF	EF	CF	\mathbf{SF}			gQoL	PF	RF	EF	CF	\mathbf{SF}		gOoL	ΡF	RF	EF	CF	\mathbf{SF}	
	Assessment of survival			Unclear								Unclear							Unclear						
	Survival Period	(mean +SD)		5y; Sur-	vival rate	(5y): 52%						5y;	Survival	rate (5y):	41%-50%				5y; Sur-	vival rate	(5y): 62%				
	QoL PRO			C30								C30							C30						
	Tumor stage			All								All							All						
	Cancer site			00								phar-	ynx						L						
ıtinued	Patients (n)			122								89							86						
Table 2. Con	First author and year of	publication		Nordgren	2007 [53]							Nordgren	2006 [52]						Nordgren	2003[51]					

Factors adjusted for	Tumor site, stage, surgical status, metastasis, radio- therapy, age, gen- der, marital status, comorbidity	Sex, age, coping by suppression, coping by disen- gagement, levels of neuroticism, TNM stage, tumor site, alcohol consump- tion, smoking sta- tus, comorbidity	T, N stage, KPS score, primary site, cigarette use, age, income, marital status	Age, tumor loca- tion, treatment modality, stage
te p-value	n.s.	0.08 < 0.001 n.a. n.a. n.a.	n.s. n.s. n.s. n.s.	<0.001
en QoL and surviva multivaria HR (CI)		0.47 (0.20-1.10) 0.97 (0.95-0.99)	п.т. п.т. п.т.	0.86 (0.80-0.93)
ion betwee e p-value	n.s.	0.001 < 0.001 0.18 0.07 0.75 0.79		<0.001
Associat univariat HR (95% CI)		0.31 (0.15-0.61) 0.98 (0.96-0.99) 0.09 (0.98-1.00) 1.00 (0.98-1.01) 1.00 (0.98-1.01) 1.00 (0.98-1.01)	n.a. n.a. n.a. n.a.	0.83 (0.77-0.88)
	gQoL	gQoL PF EF CF SF	PWB SWB EWB FWB	gQoL
Assessment of survival	Unclear	National population register	Unclear	unclear
Survival Period (mean +SD)	5y; Median survival: 58m (3.5- 102)	> 4y; Survival rate (5y): 69.1%	Unknown	5y, OS 55.3% at 60m
QoL PRO	C30	C3 o	FACT G	C30
Tumor stage	II/III/ IV	IIA	П/ ПЛ/	All
Cancer site	OC, OP	OC, OP, L, other	OC, OP, HP, L	OC
Patients (n)	80	106	1093	142
First author and year of publication	Oskama 2010 [45]	Osthusa 2013 [48]	Siddiquia 2007 [39]	Tarsitanoa 2012 [42]

Table 2. Continued

Factors adjusted for		Age, time since diagnose, marital status, education, site, stage, comor- bidity, smoking	disease stage, age, gender, smoking, alcohol con- sumption, nodal involvement and tumor site	Cancer stage, cancer site, age, smoking, alcohol, BMI, trial arm assignment	Tumor site, stage, surgical status, metastasis, radio- therapy, age, gen- der, marital status, comorbidity
e p-value		< 0.001 0.589	0.72 0.001	0.023 <0.001 <0.001 0.024 0.037 0.011	0.00
en QoL and survival multivariat HR (CI)		0.98(0.80-0.93) 0.98 (0.93-1.05)	1.1 (0.6-2.1) 2.5 (1.4-4.3)	0.91)0.84-0.99) 0.75 (0.68-0.83) 0.85 (0.79-0.91) 0.91 (0.84-0.99) 0.91 (0.84-0.99) 0.91 (0.82-0.97) 0.89 (0.82-0.97)	0.96 (0.94-0.99) 5.08(2.30-14.60)
ion betwee e p-value		<0.001		0.01 <0.001 0.002 0.03 0.03 0.03	≤0.01 ≤0.01
Associat univariat HR (95% CI)		0.83(0.77-0.88) 0.96(0.91-1.02)	.m.n	0.91 0.96 0.91 0.92 0.92 0.92	n.m. n.m.
		PCS MCS	GHQ LSS	Ch-QoL Ch-PF Ch-RF Ch-EF Ch-EF Ch-SF	QoL6m QoL-de- teriora- tion
Assessment of survival		Death file for veterans	Centralized national clinical booking system	Mortality files, death certificates	Unclear
Survival Period (mean +SD)		FU: 5.1y (13d-7.2y)	10y Median survival: 6y	5y; Median follow-up: 6.5y	5y Median survival: 58m (3.5- 102)
QoL PRO	atment	SF-36	Auck- land QoL	C30	C30
Tumor stage	after tre	All	All	П/П	II/III/ IV
Cancer site	ring or (HNC	HNC	HNC (lar- ynx and oth- ers)	OC, OP
Patients (n)	essed du	495	200	540	8
First author and year of publication	HRQoL ass	Karvonena 2008 [47]	Mehanna 2006 [56]	Meyera 2009 [44]	0skama 2010 [45]

Table 2. Contir

-

Table 2. Com	tinued											
First author and year of	Patients (n)	Cancer site	Tumor stage	QoL PRO	Survival Period	Assessment of survival		Associat	ion betwee	n QoL and survival	_	Factors adjusted for
publication					(mean +SD)			univariat	e	multivaria	te	
								HR (95% CI)	p-value	HR (CI)	p-value	
Osthusa 2011 [46]	139	SCC SCC	Ч	C30	> 5y; fu: 75m (SD 4m)	Unclear	QoL func. score	n.m. n.m.		0.97 (0.93-1.01) 0.97 (0.93-1.01)	0.154	TNM stage, site, alcohol consump- tion, currently smoking, heart and lung disease, gender, age, time between diagnosis and inclusion, neuroticism, avoidance focused coping by suppression of competing activity
Thompson 2011 [54]	276	0C, 0P, HP, L	All	SF-36	3/4/5y Survival rate (5y): 61.1%	Unclear	ToD SHG	n.m. n.m.	0.01 <0.001	n.m. 4.62	0.29 0.02	Stage, alcohol abuse, tobacco use, age
Abbreviations Questionnaire ing; EQ5D, Eu health questio Quality of Life	: Act, probl c Core 30; C moQol-5d-5 mnaire; gQ	ems perf F: cognit L; FACT oL, globa aire Head	orming us live functi -G, Funct I quality c d and Nec	sual activit oning; Ch: ional Asse of life; HAI k 35; HNC	ies; BMI, bo : change; CI, ssment of Ca DS, Hospital C, head and r	dy mass index; confidence int mcer Therapy - Anxiety and D neck cancer; H1	; C3o, Eur erval; D& – General epression NSCC, he:	opean Organization A, depression and <i>z</i> fu: follow up; func Scale; HN35, Euro ad and neck squam	n for Resea unxiety; EF , function pean Orga ous cell car	rch and Treatment : emotional functio ing; FWB: function nization for Resear cinoma; HP, hypol	of Cancer ning; EW nal well-be ch and Tre pharynx; F	Quality of Life B: emotional well-be- ing; GHQ, general atment of Cancer IR, hazard ratio;

Table 2. Continued

KPS, Karnofsky Performance Score; L, larynx; LSS, Life Satisfaction Score; m, months; MCS, mental component summery; Mob, problems with mobility; n, number; n.a., not assessed; n.m., not mentioned; NP, nasopharynx, n.s., not significant; OC, oral cavity; OP, oropharynx; OS, overall survival; P&D, pain and discomfort; PCS, physical component summery; PF, physical function; PHS: physical health score; PRHS, patient related health state; PRO; Patient Reported Outcome; PWB, physical well-being; QoL: quality of life; RF: role functioning; Sc, problems with self-care; SD, standard deviation; SF-36, short form 36 health survey; SF: social functioning; SWB: social well-being; (T)(N)(M)-stage, tumor, node, metastasis; Y: year(s)

^a indicates the high quality study

Inconsistent findings were found for global QoL, role functioning, cognitive functioning, and social functioning, with 25-50% of the high quality studies reporting significant positive associations with survival, indicating insufficient evidence (level C). Insufficient evidence was also found for the positive associations of emotional function and mental HRQoL with survival, as none of the high quality studies reported a significant association (Table 3). Two separate articles^{33:34} on the same patient population reported no significant association of physical, functional, emotional and social well-being with survival. One study³⁶ using the Eq5D showed that severe pain and discomfort was significantly associated with impaired survival, and no significant associations were found for mobility, self-care, usual activities, and anxiety and **depression**.

HRQoL during or after treatment and the association with survival

Four high quality studies³⁹⁻⁴² and two low quality studies^{49;51} examined the association between HRQoL during or after treatment and survival, in which HRQoL was assessed at different time points varying from during treatment to being disease free for more than 2 years (Table 3). Three studies reported that physical HRQoL assessed within 2 years after diagnosis,⁴² global QoL 6 months post-treatment⁴⁰ and low QoL one year after treatment⁵¹ were significantly positively associated with survival. Fang et al.⁴⁵ reported that improvements in these domains from pre-treatment to HRQoL assessed during radiotherapy were not significantly associated with survival. In patients who were disease free for more than 2 years, Thompson et al.⁴⁹ reported a 4 times higher likelihood of death in patients with a higher QoL compared to patients with a lower QoL. In contrast, Osthus et al.⁴¹ found no significant association between global QoL and survival in patients who were disease free for at least one year (Table 2).

In addition, two high quality studies^{39:40} included change in HRQoL from pretreatment to 6 months after treatment. Both studies reported a significant association between change in global QoL and survival, after adjusting for clinical, demographic and lifestyle-related factors indicating strong evidence (Table 3). Regarding the other subscales (physical functioning, role functioning, emotional functioning, cognitive functioning and social functioning), improvement from pre-treatment to 6 months after treatment was also significantly positively associated with overall survival³⁹ (Table 2).

	Studies included,		Ass	ociat	ion be and su	etwee urviva	n HR(al	QoL		Level of evidence
	n	n.a.	univa n.s.	riate s.	e%	n.a.	multiv n.s.	ariat s.	e	
HRQoL assessed pre	-treatment									
Global QoL										
All studies	12	0	5	7	58	3	5	4	33	
High quality	6	0	3	3	50	1	3	2	33	Insufficient
Low quality	6	0	2	4	67	2	2	2	33	
Physical function										
All studies	10	1	0	9	90	0	4	6	60	
High quality	4	0	0	4	100	0	1	3	75	Strong
Low quality	6	1	0	5	83	0	3	3	50	
Role function										
All studies	9	0	6	3	33	5	2	2	22	
High quality	4	0	2	2	50	1	2	2	50	Insufficient
Low quality	5	0	4	1	20	4	0	0	0	
Emotional function										
All studies	9	0	9	0	0	7	2	0	0	
High quality	4	0	4	0	0	2	2	0	0	Insufficient
Low quality	5	0	5	0	0	5	0	0	0	
Cognitive function										
All studies	9	0	6	3	33	6	1	2	22	
high quality	4	0	2	2	50	2	0	2	50	Insufficient
Low quality	5	0	4	1	20	4	1	0	0	
Social function										
All studies	9	0	5	4	44	4	4	1	11	
High quality	4	0	2	2	50	2	1	1	25	Insufficient
Low quality	5	0	3	2	40	2	3	0	0	
Mental HRQoL										
All studies	2	1	1	0	0	0	2	0	0	
High quality	1	0	1	0	0	0	1	0	0	Insufficient

Table 3. Overview of the level of evidence regarding the associations between HRQoL and survival

Table 3. Continued

	Studies included,		Asso	ociat	ion be and su	etwee: urviva	n HR(1	QoL		Level of evidence
	n		univa	riate	<u>,</u>	1	multiv	aria	te	
		n.a.	n.s.	s.	%	n.a.	n.s.	s.	%	
Low quality	1	1	0	0	0	0	1	0	0	
HRQoL assessed afte	er treatment									
Change Global QoL										
All studies	2	0	0	2	100	0	0	2	100	
High quality	2	0	0	2	100	0	0	2	100	Strong
Low quality	0	-	-	-		-	-	-		

Abbreviations: HRQoL, health related quality of life; n.a., not applicable; n.s., not significant; QoL, quality of life; s., significant.

DISCUSSION

This study reviewed the available evidence on the association between HRQoL and survival in patients with HNC using a best evidence synthesis. Our review resulted in two important findings. First, we found strong evidence that higher pre-treatment physical functioning of patients with HNC was associated with increased survival. This finding is in line with studies among patients with other cancers.⁷⁻¹⁰ Moreover, Montazeri et al.⁸ found a stronger association between pre-treatment physical functioning and survival compared to other HRQoL domains in patients with various cancer sites, although not in patients with HNC, which may be due to the limited number of studies on HNC.

We found insufficient evidence for an association with survival for other pre-treatment HRQoL subscales (global QoL, role functioning, emotional functioning, cognitive functioning, social functioning and mental HRQoL), after applying a best evidence **synthesis**. Although Mehanna et al.²⁹ reported in their review an association between cognitive functioning and survival, this finding was based on a single study. Quinten et al.¹⁰ did a pooled analysis using data from randomized controlled trials using the EORTC-QLQ C30. Among patients with HNC no association for the majority of HRQoL subscales and survival was seen, however they found a remarkable association between lower emotional functioning and increased survival. Regarding psychosocial factors, a **meta-analysis showed that psychosocial factors were associated with poorer survival in cancer patients.**⁵³ Obviously, future high quality studies are needed to clarify the associations between cognitive functioning or emotional functioning and survival. Second, we found strong evidence for the association between change in global QoL from pre-treatment to 6 months post-treatment and survival.

These results support the findings of the review of Montazeri et al.⁸ who reported for some cancers, that changes in global QoL scores were prognostic for survival, when pre-treatment global QoL was not. Although HRQoL after treatment seems to be associated with survival as well, the limited number of studies and the large variation in assessment time points hampered us to apply a best evidence synthesis.

Future studies focusing on HRQoL should not only focus on pre-treatment HRQoL, but also include HRQoL at different time points after treatment, since they may have predictive value for survival.^{8;29;39;40;51}

This systematic review included a quality rating and 58% of the studies were of high quality. Major concerns were the inadequate descriptions of survival assessment, adjustment for possible confounders and presentation of point estimates and measures of variability. As a consequence associations between HRQOL and survival might be biased. Because it is well established that age, tumor location and stage, smoking, alcohol intake and co-morbidity are significantly associated with survival, we selected these variables as confounders to be adjusted for in the reviewed studies.⁵⁴⁻⁵⁶ However the choice of confounders remains somewhat arbitrary, and it is known that other factors such as more extended tumor characteristics (e.g. extra nodal tumor growth, HPV status, histological grading) and other socio-demographic factors (e.g. marital or socioeconomic status) may be important as well.^{35:40:57:58}

Strengths of this study were the methodological quality assessment and best evidence synthesis. the inclusion of only prospective studies, it's solely focus on patients with HNC, and the systematic search in collaboration with a librarian. However some limitations must be noted. The majority of the studies seemed to selectively report the data since they did not report hazard ratios and confidence intervals of non-significant associations. This hampered us to conduct a meta-analysis. Instead we used a best evidence synthesis to summarize the available studies, since this allowed us to take into account non-significant associations as well. In our best evidence synthesis we included only high quality studies to reduce possible bias. However, similar to other reviews and meta-analysis, publication bias endangers the external validity, therefore publication bias cannot be ruled out.

Finally, although in our best evidence synthesis we only included high quality in which the association between HRQoL and survival was adjusted for important demographic, clinical and lifestyle-related factors, some residual confounding may still be present. For instance, although in most high quality studies tumor location and stage were taken into account as possible confounders, it cannot be ruled out that HRQoL may be a reflection of disease severity, and thereby associated with survival. HRQoL and particularly physical functioning and change in global QoL were independently associated with survival. Improving these HRQoL domains may therefore be an interesting target to improve the survival. However, it is currently unclear whether poorer HRQoL represents a more advance and severe disease, or whether improving these HRQoL domains would result in improved survival.

In conclusion, this systematic review showed that higher levels of pre-treatment physical function and change in global QoL from pre-treatment to 6 months after treatment were associated with increased survival in patients with HNC. There is currently insufficient evidence for an association between other HRQoL domains and survival. Future high quality prospective studies should provide more insight into which time points are most predictive for survival and also determine the role of other HRQoL domains. To obtain a more precise estimate of the association between HRQoL and survival, we recommend that these future studies should more clearly describe how they collected survival data, adjust for relevant confounders and present point of estimates and measures of variability of the association.

REFERENCES

- 1. Courneya KS, Friedenreich CM. Physical exercise and quality of life following cancer diagnosis: a literature review. Ann Behav Med 1999;21(2):171-9.
- Courneya KS. Exercise in cancer survivors: an overview of research. Med Sci Sports Exerc 2003 Nov;35(11):1846-52.
- Curt GA. Impact of fatigue on quality of life in oncology patients. Semin Hematol 2000 Oct;37(4 Suppl 6):14-7.
- de JN, Courtens AM, Abu-Saad HH, Schouten HC. Fatigue in patients with breast cancer receiving adjuvant chemotherapy: a review of the literature. Cancer Nurs 2002 Aug;25(4):283-97.
- Harrison JD, Young JM, Price MA, Butow PN, Solomon MJ. What are the unmet supportive care needs of people with cancer? A systematic review. Support Care Cancer 2009 Aug;17(8):1117-28.
- 6. Lucia A, Earnest C, Perez M. Cancer-related fatigue: can exercise physiology assist oncologists? Lancet Oncol 2003 Oct;4(10):616-25.
- Gotay CC, Kawamoto CT, Bottomley A, Efficace F. The prognostic significance of patient-reported outcomes in cancer clinical trials. J Clin Oncol 2008 Mar 10;26(8):1355-63.
- Montazeri A. Quality of life data as prognostic indicators of survival in cancer patients: an overview of the literature from 1982 to 2008. Health Qual Life Outcomes 2009;7:102.
- Quinten C, Coens C, Mauer M, Comte S, Sprangers MA, Cleeland C, et al. Baseline quality of life as a prognostic indicator of survival: a meta-analysis of individual patient data from EORTC clinical trials. Lancet Oncol 2009 Sep;10(9):865-71.
- Quinten C, Martinelli F, Coens C, Sprangers MA, Ringash J, Gotay C, et al. A global analysis of multitrial data investigating quality of life and symptoms as prognostic factors for survival in different tumor sites. Cancer 2013 Oct 11;120(2):302-11.
- Abendstein H, Nordgren M, Boysen M, Jannert M, Silander E, Ahlner-Elmqvist M, et al. Quality of life and head and neck cancer: a 5 year prospective study. Laryngoscope 2005 Dec;115(12):2183-92.
- Bjordal K, Ahlner-Elmqvist M, Hammerlid E, Boysen M, Evensen JF, Biorklund A, et al. A prospective study of quality of life in head and neck cancer patients. Part II: Longitudinal data. Laryngoscope 2001 Aug;111(8):1440-52.
- 13. Borghgraef K, Delaere P, Van den BW, Feenstra L. Quality of life in head and neck cancer patients. Acta Otorhinolaryngol Belg 1997;51(2):69-72.
- 14. Funk GF, Karnell LH, Christensen AJ. Long-term health-related quality of life in survivors of head and neck cancer. Arch Otolaryngol Head Neck Surg 2012 Feb;138(2):123-33.
- 15. Goldstein DP, Hynds KL, Christensen AJ, Funk GF. Health-related quality of life profiles based on survivorship status for head and neck cancer patients. Head Neck 2007 Mar;29(3):221-9.
- Hammerlid E, Ahlner-Elmqvist M, Bjordal K, Biorklund A, Evensen J, Boysen M, et al. A prospective multicentre study in Sweden and Norway of mental distress and psychiatric morbidity in head and neck cancer patients. Br J Cancer 1999 May;80(5-6):766-74.
- 17. Hammerlid E, Silander E, Hornestam L, Sullivan M. Health-related quality of life three years after diagnosis of head and neck cancer--a longitudinal study. Head Neck 2001 Feb;23(2):113-25.
- de Graeff A, de Leeuw JR, Ros WJ, Hordijk GJ, Blijham GH, Winnubst JA. Long-term quality of life of patients with head and neck cancer. Laryngoscope 2000 Jan;110(1):98-106.
- Hammerlid E, Taft C. Health-related quality of life in long-term head and neck cancer survivors: a comparison with general population norms. Br J Cancer 2001 Jan;84(2):149-56.
- 20. Hammerlid E, Bjordal K, Ahlner-Elmqvist M, Boysen M, Evensen JF, Biorklund A, et al. A prospective

study of quality of life in head and neck cancer patients. Part I: at diagnosis. Laryngoscope 2001 Apr;111(4 Pt 1):669-80.

- 21. Mehanna HM, Morton RP. Deterioration in quality-of-life of late (10-year) survivors of head and neck cancer. Clin Otolaryngol 2006 Jun;31(3):204-11.
- 22. Morton RP, Izzard ME. Quality-of-life outcomes in head and neck cancer patients. World J Surg 2003 Jul;27(7):884-9.
- 23. de Graeff A, de Leeuw JR, Ros WJ, Hordijk GJ, Blijham GH, Winnubst JA. A prospective study on quality of life of patients with cancer of the oral cavity or oropharynx treated with surgery with or without radiotherapy. Oral Oncol 1999 Jan;35(1):27-32.
- Bozec A, Poissonnet G, Converset S, Lattes L, Chamorey E, Demard F, et al. Head and neck reconstructive surgery with free flaps and quality of life: a prospective study]. Rev Laryngol Otol Rhinol (Bord) 2007;128(1-2):11-8.
- 25. List MA, Siston A, Haraf D, Schumm P, Kies M, Stenson K, et al. Quality of life and performance in advanced head and neck cancer patients on concomitant chemoradiotherapy: a prospective examination. J Clin Oncol 1999 Mar;17(3):1020-8.
- 26. List MA, Ritter-Sterr CA, Baker TM, Colangelo LA, Matz G, Pauloski BR, et al. Longitudinal assessment of quality of life in laryngeal cancer patients. Head Neck 1996 Jan;18(1):1-10.
- 27. Borggreven PA, Aaronson NK, Verdonck-de Leeuw IM, Muller MJ, Heiligers MLCH, Bree Rd, et al. Quality of life after surgical treatment for oral and oropharyngeal cancer: a prospective longitudinal assessment of patients reconstructed by a microvascular flap. Oral Oncol 2007 Nov;43(10):1034-42.
- 28. Oskam IM, Verdonck-de Leeuw IM, Aaronson NK, Witte BI, de BR, Doornaert P, et al. Prospective evaluation of health-related quality of life in long-term oral and oropharyngeal cancer survivors and the perceived need for supportive care. Oral Oncol 2013 May;49(5):443-8.
- 29. Mehanna HM, De Boer MF, Morton RP. The association of psycho-social factors and survival in head and neck cancer. Clin Otolaryngol 2008 Apr;33(2):83-9.
- Hayden JA, Cote P, Bombardier C. Evaluation of the quality of prognosis studies in systematic reviews. Ann Intern Med 2006 Mar 21;144(6):427-37.
- Singh A, Uijtdewilligen L, Twisk JW, van MW, Chinapaw MJ. Physical activity and performance at school: a systematic review of the literature including a methodological quality assessment. Arch Pediatr Adolesc Med 2012 Jan;166(1):49-55.
- Slavin RE. Best evidence synthesis: an intelligent alternative to meta-analysis. J Clin Epidemiol 1995 Jan;48(1):9-18.
- 33. Coyne JC, Pajak TF, Harris J, Konski A, Movsas B, Ang K, et al. Emotional well-being does not predict survival in head and neck cancer patients: a Radiation Therapy Oncology Group study. Cancer 2007 Dec 1;110(11):2568-75.
- Siddiqui F, Pajak TF, Watkins-Bruner D, Konski AA, Coyne JC, Gwede CK, et al. Pretreatment quality of life predicts for locoregional control in head and neck cancer patients: a radiation therapy oncology group analysis. Int J Radiat Oncol Biol Phys 2008 Feb 1;70(2):353-60.
- 35. deGraeff A, de Leeuw JR, Ros WJ, Hordijk GJ, Blijham GH, Winnubst JA. Sociodemographic factors and quality of life as prognostic indicators in head and neck cancer. Eur J Cancer 2001 Feb;37(3):332-9.
- Lango MN, Egleston B, Fang C, Burtness B, Galloway T, Liu J, et al. Baseline health perceptions, dysphagia, and survival in patients with head and neck cancer. Cancer 2014;120(6):840-7.
- 37. Tarsitano A, Pizzigallo A, Ballone E, Marchetti C. Health-related quality of life as a survival predictor for patients with oral cancer: is quality of life associated with long-term overall survival? Oral Surg Oral Med Oral Pathol Oral Radiol 2012 Dec;114(6):756-63.

- 38. Fang FM, Tsai WL, Chien CY, Chen HC, Hsu HC, Huang TL, et al. Pretreatment quality of life as a predictor of distant metastasis and survival for patients with nasopharyngeal carcinoma. J Clin Oncol 2010 Oct 1;28(28):4384-9.
- 39. Meyer F, Fortin A, Gelinas M, Nabid A, Brochet F, Tetu B, et al. Health-related quality of life as a survival predictor for patients with localized head and neck cancer treated with radiation therapy. J Clin Oncol 2009 Jun 20;27(18):2970-6.
- 40. Oskam IM, Verdonck-de L, I, Aaronson NK, Kuik DJ, de BR, Doornaert P, et al. Quality of life as predictor of survival: a prospective study on patients treated with combined surgery and radiotherapy for advanced oral and oropharyngeal cancer. Radiother Oncol 2010 Nov;97(2):258-62.
- Osthus AA, Aarstad AK, Olofsson J, Aarstad HJ. Health-related quality of life scores in long-term head and neck cancer survivors predict subsequent survival: a prospective cohort study. Clin Otolaryngol 2011 Aug;36(4):361-8.
- Karvonen-Gutierrez CA, Ronis DL, Fowler KE, Terrell JE, Gruber SB, Duffy SA. Quality of life scores predict survival among patients with head and neck cancer. J Clin Oncol 2008 Jun 1;26(16):2754-60.
- 43. Osthus AA, Aarstad AK, Olofsson J, Aarstad HJ. Prediction of survival by pretreatment health-related quality-of-life scores in a prospective cohort of patients with head and neck squamous cell carcinoma. JAMA Otolaryngol Head Neck Surg 2013 Jan;139(1):14-20.
- 44. Curran D, Giralt J, Harari PM, Ang KK, Cohen RB, Kies MS, et al. Quality of life in head and neck cancer patients after treatment with high-dose radiotherapy alone or in combination with cetuximab. J Clin Oncol 2007 Jun 1;25(16):2191-7.
- 45. Fang FM, Liu YT, Tang Y, Wang CJ, Ko SF. Quality of life as a survival predictor for patients with advanced head and neck carcinoma treated with radiotherapy. Cancer 2004 Jan 15;100(2):425-32.
- 46. Nordgren M, Abendstein H, Jannert M, Boysen M, Ahlner-Elmqvist M, Silander E, et al. Health-related quality of life five years after diagnosis of laryngeal carcinoma. Int J Radiat Oncol Biol Phys 2003 Aug 1;56(5):1333-43.
- 47. Nordgren M, Jannert M, Boysen M, Ahlner-Elmqvist M, Silander E, Bjordal K, et al. Health-related quality of life in patients with pharyngeal carcinoma: a five-year follow-up. Head Neck 2006 Apr;28(4):339-49.
- **48**. Nordgren M, Hammerlid E, Bjordal K, Ahlner-Elmqvist M, Boysen M, Jannert M. Quality of life in oral carcinoma: a 5-year prospective study. Head Neck 2008 Apr;30(4):461-70.
- Thompson TL, Pagedar NA, Karnell LH, Funk GF. Factors associated with mortality in 2-year survivors of head and neck cancer. Arch Otolaryngol Head Neck Surg 2011 Nov;137(11):1100-5.
- Grignon LM, Jameson MJ, Karnell LH, Christensen AJ, Funk GF. General health measures and long-term survival in patients with head and neck cancer. Arch Otolaryngol Head Neck Surg 2007 May;133(5):471-6.
- Mehanna HM, Morton RP. Does quality of life predict long-term survival in patients with head and neck cancer? Arch Otolaryngol Head Neck Surg 2006 Jan;132(1):27-31.
- 52. Brooks R. EuroQol: the current state of play. Health Policy 1996 Jul;37(1):53-72.
- Chida Y, Hamer M, Wardle J, Steptoe A. Do stress-related psychosocial factors contribute to cancer incidence and survival? Nat Clin Pract Oncol 2008 Aug;5(8):466-75.
- 54. Boje CR, Dalton SO, Gronborg TK, Primdahl H, Kristensen CA, Andersen E, et al. The impact of comorbidity on outcome in 12 623 Danish head and neck cancer patients: a population based study from the DAHANCA database. Acta Oncol 2013 Feb;52(2):285-93.
- 55. Hoff CM, Grau C, Overgaard J. Effect of smoking on oxygen delivery and outcome in patients treated with radiotherapy for head and neck squamous cell carcinoma--a prospective study. Radiother Oncol

2012 Apr;103(1):38-44.

- 56. Dikshit RP, Boffetta P, Bouchardy C, Merletti F, Crosignani P, Cuchi T, et al. Lifestyle habits as prognostic factors in survival of laryngeal and hypopharyngeal cancer: a multicentric European study. Int J Cancer 2005 Dec 20;117(6):992-5.
- 57. Lindenbergh-van der Plas M, Brakenhoff RH, Kuik DJ, Buijze M, Bloemena E, Snijders PJ, et al. Prognostic significance of truncating TP53 mutations in head and neck squamous cell carcinoma. Clin Cancer Res 2011 Jun 1;17(11):3733-41.
- 58. Mehanna H, West CM, Nutting C, Paleri V. Head and neck cancer--Part 2: Treatment and prognostic factors. BMJ 2010;341:c4690.



- CHAPTER 4

HEALTH RELATED QUALITY OF LIFE AND OVERALL SURVIVAL: A PROSPECTIVE STUDY IN PATIENTS WITH HEAD AND NECK CANCER TREATED WITH RADIOTHERAPY

A.J. van Nieuwenhuizen L.M. Buffart J.A. Langendijk M.R. Vergeer J. Voortman C.R. Leemans I.M. Verdonck-de Leeuw

Submitted to: Quality of Life Research

ABSTRACT

Purpose We aimed to examine whether pre-treatment, post-treatment and change in health- related quality of Life (HRQoL) is associated with survival, in patients with head and neck cancer (HNC).

Methods We included 948 newly diagnosed HNC patients treated with primary or adjuvant (chemo)radiotherapy with curative intent. The EORTC QLQ-C30 questionnaire was assessed pre-treatment and at 6 weeks, 6 months and 12 months post-treatment. Multivariable Cox regression analyses were performed to examine whether HRQoL at all time-points and changes in HRQoL over time were associated with survival, after adjusting for demographic, clinical and lifestyle-related variables.

Results Higher HRQoL scores were significantly associated with improved 5-year overall survival at all time-points, except for the subscale global QoL at 6 weeks. Changes in HRQoL at 6 weeks post-treatment compared to pre-treatment were not significantly associated with survival. Changes in physical (HR: 0.88 95% CI: 0.82-0.96) and emotional functioning (HR: 0.90 95% CI: 0.85-0.96) from pre-treatment to 6 months post-treatment and changes in global QOL, and physical, emotional, and social functioning from pre-treatment to 12 months post-treatment were significantly associated with survival.

Conclusion Higher HRQoL reported pre-treatment and post-treatment (6 weeks, 6 months and 12 months) are significantly associated with improved survival, as well as changes in HRQoL at 6 and 12 months compared to pre-treatment.

Implications for cancer survivors Our results highlight the value of monitoring HRQoL and to identify those patients that report decreased or deteriorated HRQOL. This may help to further improve cancer care in a timely and efficient manner.

INTRODUCTION

Many patients with head and neck cancer (HNC) have to deal with severe physical and psychosocial problems because of the disease and its treatment. Additionally, they are often confronted with HNC specific problems, such as oral dysfunction, swallowing and speech impediments.¹⁻¹⁰ These disorders have a distinct impact on the health-related quality of life (HROoL) of patients with HNC. It has been shown that the initial course of HRQoL during the first 2 years following treatment is favorable in HNC survivors compared patients who ultimately succumb to the disease. 9 Furthermore, previous observational studies showed a significant association between HRQoL and survival, independently from other demographic, lifestyle-related and clinical factors.¹¹⁻²⁰ In a previous systematic review, we found evidence for a significant association between pre-treatment physical functioning and survival, and between change in global QoL from pre-treatment to 6 months follow-up and survival in patients with HNC.²¹ However, we noticed that only a small majority (58%) of the existing studies was of high quality. Particularly, 63% of the studies included in that review did not consider relevant confounders (e.g. eleven studies did not assess comorbidity, and seven studies did not assess smoking and alcohol consumption).²¹

As a consequence, it remains difficult to draw firm conclusions on the association between HRQoL and survival. Therefore, the aim of this prospective study was to examine whether pre-treatment HRQoL, HRQoL at 6 weeks, and 6 and 12 months after treatment and change in HRQoL is associated with survival, after adjusting for demographic, clinical, and lifestyle-related factors in patients with HNC.

Patients and Methods

Study population

Between January 1999 and October 2009, all newly diagnosed patients with HNC who were planned to be treated with primary or adjuvant (chemo)radiotherapy in the Amsterdam University Medical Centres, location VUmc, completed questionnaires on HRQoL before treatment, and at 6 weeks, and 6 and 12 months after treatment as part of clinical routine. Patients were eligible for the current analyses if they: 1) were diagnosed with primary squamous cell carcinomas of the mucosal surfaces of the oral cavity, oropharynx, hypopharynx and larynx, 2) were treated with (chemo) radiotherapy or surgery combined with (chemo)radiotherapy with curative intent, 3) were ≥ 18 years old, 4) were able to read and understand the Dutch language, and 5) completed the pre-treatment questionnaire). Patients were excluded if they had a distant metastasis, were previously treated with surgery or radiotherapy in the head and neck area, or brachytherapy, or had a serious cognitive impairment at baseline.

Health-related quality of life

HRQoL was assessed using the 30-item European Organization for Research and Treatment of Cancer, (EORTC) Quality of Life Questionnaire core module (QLQ-C30)²². For the current analyses, we included the global quality of life (QoL) scale and the five function scales (physical, role, emotional, cognitive, and social functioning). Higher scores on the global QoL and functioning scales represent higher HRQoL.

<u>Survival</u>

Five-year survival was assessed by linking medical records to the Dutch death certificate register. Survival was calculated from the date of inclusion (pre-treatment questionnaire) until death.

Demographic, lifestyle-related and clinical factors

Demographic (i.e. gender, age, socio-economic status (SES)), lifestyle-related (i.e. smoking in packyears, smoking history, alcohol use (units per day), alcohol abuse (≥ 5 units per day)), and clinical factors (i.e. tumor site, stage, HPV status, types of treatment and comorbidity) were obtained from medical records. Socio-economic status was determined using zip codes of patients' living area. Zip codes were translated to SES according to The Netherlands Institute for Social Research ²³. This system describes the social status of a district compared to other districts in The Netherlands using an algorithm based on mean income, percentage of people with low income, percentage of people with low education and percentage of people without a job. Therefore, the mean score of all districts in The Netherlands is zero. We dichotomized SES scores to high (> mean value) versus low (< mean value).

Tumor stage was determined according to the American Joint Committee on cancer (AJCC) TNM staging system (seventh ed., 2010). Tumor site was categorized into cancer of the oral cavity, HPV positive oropharynx, HPV negative oropharynx, larynx or hypopharynx. All biopsies of patients with oropharyngeal cancer were tested for HPV on formalin-fixed, paraffin embedded tumor specimen according to a validated test algorithm.^{24,25}

Treatment modality was categorized into radiotherapy alone, chemoradiation, or surgery followed by adjuvant (chemo)radiation. Additionally, we recorded whether the patients were treated with 3D-CRT (3-Dimensional Conformal Radiotherapy) or Intensity Modulated Radiotherapy (IMRT), that was introduced in our hospital in 2004. Comorbidity was assessed by a research physician (AvN) using the Adult Comorbidity Evaluation 27 (ACE-27) score,²⁶ a validated chart built instrument examining the presence of any of the following medical conditions: cardiovascular, respiratory, gastro-intestinal, renal, endocrine, neurological, immunological, previous malignancies, psychiatric disorders, alcohol use, and severe overweight, resulting in a

total comorbidity score of none, mild, moderate or severe.

Statistical analysis

Descriptive statistics (mean, standard deviation (SD), or numbers and percentages) were generated for demographic, lifestyle-related, clinical factors, and HRQoL. Univariable and multivariable Cox proportional hazard regression analyses were used to examine the association between HRQoL and survival. In the multivariable analyses, we adjusted for relevant demographic, lifestyle-related and clinical variables Separate models were built for each HRQoL subscale and for the different time-points (pre-treatment, 6 weeks, 6 months and 12 months after treatment, and change in HRQoL at 6 weeks compared to pre-treatment, change at 6 months compared to pre-treatment, and change at 12 months compared to pre-treatment). In the regression analyses, we divided all HRQoL scores by 10 because such changes are considered clinically meaningful.²⁷ For all statistical analyses, p<0.05 was considered statistically significant.

RESULTS

Patient characteristics

From January 1999 and October 2009, 948 newly diagnosed patients with HNC met the inclusion criteria for the current analyses. All patients completed the questionnaire pre-treatment. After treatment, questionnaires were completed by 703 patients of the 947 alive (74%) at 6 weeks, 654 patients of the 914 alive (72%) at 6 months and 579 patients of the 838 alive (69%) at 12 months.

Demographic, lifestyle-related and clinical characteristics of the study population are presented in Table 1. The most frequent tumor site was larynx (43%). Among the patients with oropharyngeal cancer, 58% were diagnosed with a HPV negative tumor (HPV status was unknown in 14%). Overall, 60% of patients were alive after 5 years.

Characteristics	Patients $(n = 948)$		
Demographic factors			
Gender, n (%) male	692 (73%)		
Age, mean (SD) years	62 (11)		
High SES (above average), n (%)	139 (15%)		
Lifestyle related factors			
Smoking (packyears), mean (SD)	31 (22)		
Former or current smoker, n (%)	806 (85)		
Alcohol use (units per day), mean (SD)	3 (3)		
Former or current alcohol abuse [†] , n (%)	262 (28)		
Clinical factors			
Tumor site, n (%)			
Oral Cavity	152 (16)		
Oropharynx	306 (32)		
Oropharynx HPV positive*	86 (28)		
Oropharynx HPV negative*	176 (58)		
Oropharynx HPV unknown*	44 (14)		
Larynx	413 (44)		
Hypopharynx	77 (8)		
Disease Stage, n (%)			
Ι	171 (18)		
II	193 (20)		
III	181 (19)		
IV	402 (43)		
Comorbidity, n (%)			
None	297 (31)		
Mild	322 (34)		
Moderate	239 (25)		
Severe	90 (10)		
Type of treatment, n (%)			
Radiotherapy	522 (55)		
Chemoradiation	224 (24)		
Primary surgery with adjuvant treatment	203 (21)		

Table 1. Pre-treatment demographic, lifestyle-related and clinical characteristics of the study population

Table 1.	Continu	ed
----------	---------	----

Characteristics	Patients (n= 948)		
RT technique, n (%)			
IMRT	593 (63)		
5 year overall survival rate (%)	570 (60)		
Drop-out due to death, n (%)			
6 weeks	10 (1)		
6 months	34 (4)		
12 months	110 (12)		

Health-Related Quality of Life in relation to survival

Mean (SD) scores on the HRQoL subscales and results of Cox regression analyses are presented in Table 2. Adjusted for all included demographic, lifestyle-related and clinical factors, higher (better) scores on all subscales (global QoL, physical functioning, role functioning, emotional functioning, cognitive functioning and social functioning) as measured pre-treatment and at 6 and 12 months after treatment were significantly associated with longer survival, (Table 2). At 6 weeks after treatment, higher scores on all subscales were also significantly associated with longer survival, except for global QoL.

	Mean (SD)	Univariable model HR (95% CI)	p-value*	Multivariable model HR (95% CI) †	p-value*			
EORTC QLQ-C30 pre-treatment (n=948)								
Global quality of life	66.6 (22.3)	0.89 (0.85-0.93)	0.00	0.91 (0.87-0.96)	0.00			
Physical function	82.3 (20.8)	0.84 (0.81-0.88)	0.00	0.87 (0.83-0.91)	0.00			
Role functioning	73.4 (32.3)	0.92 (0.90-0.95)	0.00	0.93 (0.90-0.96)	0.00			
Emotional functioning	68.3 (23.4)	0.93 (0.90-0.97)	0.00	0.94 (0.90-0.97)	0.01			
Cognitive functioning	85.1 (20.9)	0.92 (0.88-0.96)	0.00	0.91 (0.87-0.95)	0.00			
Social functioning	82.4 (24.6)	0.92 (0.88-0.95)	0.00	0.91 (0.87-0.95)	0.00			
EORTC QLQ-C30 6 weeks (n=703)								
Global quality of life	66.2 (21.5)	0.90 (0.85-0.95)	0.00	0.94 (0.89-1.00)	0.06			
Physical function	74.6 (22.3)	0.86 (0.82-0.90)	0.00	0.90 (0.85-0.95)	0.00			
Role functioning	66.5 (30.6)	0.91 (0.88-0.95)	0.00	0.93 (0.90-0.97)	0.00			
Emotional functioning	76.2 (23.6)	0.93 (0.89-0.98)	0.00	0.94 (0.89-0.99)	0.02			
Cognitive functioning	83.1 (21.1)	0.92 (0.87-0.97)	0.00	0.93 (0.88-0.91)	0.01			
Social functioning	77.6 (25.1)	0.93 (0.89-0.97)	0.00	0.93 (0.88-0.98)	0.00			
EORTC QLQ-C30 6 m	onths (n=654	.)						
Global quality of life	71.0 (21.7)	0.86 (0.81-0.91)	0.00	0.87 (0.82-0.93)	0.00			
Physical function	79.7 (19.9)	0.79 (0.75-0.84)	0.00	0.80 (0.75-0.86)	0.00			
Role functioning	73.9 (29.2)	0.89 (0.85-0.93)	0.00	0.90 (0.86-0.94)	0.00			
Emotional functioning	78.9 (24.1)	0.88 (0.84-0.92)	0.00	0.88 (0.83-0.93)	0.00			
Cognitive functioning	85.2 (21.1)	0.91 (0.86-0.96)	0.00	0.89 (0.84-0.95)	0.00			
Social functioning	82.5 (23.9)	0.89 (0.85-0.93)	0.00	0.89 (0.844-0.95)	0.00			
EORTC QLQ-C30 12 months (n=579)								
Global quality of life	73.9 (21.5)	0.82 (0.77-0.87)	0.00	0.81 (0.76-0.87)	0.00			
Physical function	82.1 (19.5)	0.79 (0.74-0.85)	0.00	0.81 (0.74-0.87)	0.00			
Role functioning	78.1 (28.1)	0.86 (0.82-0.91)	0.00	0.86 (0.81-0.91)	0.00			
Emotional functioning	81.7 (22.2)	0.86 (0.81-0.92)	0.00	0.82 (0.76-0.88)	0.00			
Cognitive functioning	86.2 (19.7)	0.90 (0.84-0.97)	0.00	0.89 (0.82-0.96)	0.00			
Social functioning	85.3 (22.2)	0.85 (0.80-0.90)	0.00	0.84 (0.78-0.89)	0.00			

 † adjusted for age, gender, socio-economic status, smoking (packyears), alcohol abuse (current or history), comorbidity, tumor site, tumor stage, treatment modality

*P of the log likelihood test

Footnote: Higher global QoL and functioning scores indicates higher HRQoL (scale 0-100).
Table 3 presents the mean changes in HRQoL at 6 weeks, 6 months and 12 months after treatment, respectively, compared to pre-treatment. Changes in HRQoL from pre-treatment to 6 weeks after treatment were not significantly associated with survival for any of the subscales (Table 4). Deterioration in physical and emotional functioning at 6 months post-treatment compared to pre-treatment was significantly associated with shorter survival. Deterioration in global QoL, physical, emotional and social functioning at 12 months after treatment compared to pre-treatment was significantly associated with shorter survival.

	Mean (SD) change
EORTC QLQ-C30 Δ 6 weeks (n=703)	
Global quality of life	-1.4 (23.7)
Physical function	-8.3 (20.8)
Role functioning	-7.2 (37.4)
Emotional functioning	6.9 (24.1)
Cognitive functioning	-2.2 (23.6)
Social functioning	-5.0 (28.1)
EORTC QLQ-C30 Δ 6 months (n=654)	
Global quality of life	3.3 (23.6)
Physical function	-4.2 (18.8)
Role functioning	-1.0 (33.9)
Emotional functioning	9.9 (24.6)
Cognitive functioning	-0.6 (22.1)
Social functioning	-1.0 (28.3)
EORTC QLQ-C30 Δ 12 months (n=579)	
Global quality of life	5.3 (23.5)
Physical function	-2.8 (18.9)
Role functioning	3.6 (34.9)
Emotional functioning	12.2 (23.0)
Cognitive functioning	-0.1 (20.9)
Social functioning	1.5 (27.1)

Table 3. Mean (SD) change scores in HRQoL

 Δ change compared to pre-treatment. A negative mean change score indicates worsening of HRQoL after treatment compared to pre-treatment.

DISCUSSION

This comprehensive prospective study among a large group of patients with HNC showed that better HRQoL was significantly associated with longer survival, adjusted for demographic, lifestyle-related and clinical factors. This association was found for global QoL, and physical, role, emotional, cognitive, and social functioning before treatment as well as 6 weeks, 6 months and 12 months after treatment. Changes in HRQoL at 6 weeks after treatment compared to pretreatment were not significantly associated with survival. However, deterioration in physical and emotional functioning at 6 and 12 months after treatment was significantly associated with shorter survival, as well as deterioration in global QoL social functioning at 12 months.

Our finding that worse HRQoL before and after treatment is significantly associated with shorter survival supports results from previous observational studies in patients with HNC.^{11-16,18-20,28} In contrast to previous studies that reported an association with survival of some HRQoL domains and measured at different time points,^{21,29,30} we consistently found that global QoL and all function domains of HRQoL assessed at all time points during the first year after cancer diagnosis were associated with survival. The inconsistent findings across the different subscales and time points in the previous studies may be related to the smaller sample sizes in those studies^{12,14,16,17,19,20,31,32} and the heterogeneity of the tumor sites and stages.^{13,15,33-41}

Interestingly, where HRQoL measured 6 weeks after treatment was significantly associated with survival, change HRQoL as measured at 6 weeks after treatment compared to pre-treatment was not. This may be explained by the fact that shortly after treatment, many patients still suffer from the acute side effects of treatment and change in HRQoL at short term is not yet a discriminating factor.^{6,9} Most of these acute adverse effects are absent from 6 months onwards.^{1,2,6,9}

	Univariable model HR (95% CI)	p-value*	Multivariable model HR (95% CI) †	p-value*
EORTC QLQ-C30 Δ 6 v	weeks (n=703)			
Global quality of life	1.01 (0.96-1.07)	0.62	1.02 (0.96-1.07)	0.59
Physical function	0.98 (0.93-1.04)	0.57	0.98 (0.93-1.04)	0.56
Role functioning	0.99 (0.96-1.02)	0.61	0.99 (0.96-1.03)	0.71
Emotional functioning	0.99 (0.94-1.04)	0.74	1.00 (0.95-1.05)	0.86
Cognitive functioning	1.01 (0.96-1.06)	0.81	1.01 (0.96-1.07)	0.66
Social functioning	1.00 (0.96-1.05)	0.89	1.00 (0.95-1.05)	0.96
EORTC QLQ-C30 A 6 1	nonths (n=654)			
Global quality of life	0.95 (0.90-1.01)	0.10	0.94 (0.88-1.00)	0.05
Physical function	0.90 (0.84-0.97)	0.01	0.88 (0.82-0.96)	0.00
Role functioning	0.98 (0.94-1.02)	0.33	0.97 (0.93-1.02)	0.23
Emotional functioning	0.91 (0.86-0.97)	0.00	0.90 (0.85-0.96)	0.00
Cognitive functioning	0.96 (0.90-1.03)	0.25	0.96 (0.90-1.03)	0.24
Social functioning	0.96 (0.92-1.01)	0.12	0.97 (0.92-1.03)	0.29
EORTC QLQ-C30 Δ 12	months (n=579)			
Global quality of life	0.93 (0.86-0.99)	0.03	0.90 (0.84-0.97)	0.00
Physical function	0.91 (0.83-0.99)	0.03	0.89 (0.81-0.97)	0.01
Role functioning	0.97 (0.92-1.02)	0.19	0.96 (0.91-1.01)	0.12
Emotional functioning	0.91 (0.84-0.97)	0.01	0.87 (0.81-0.94)	0.00
Cognitive functioning	0.97 (0.89-1.05)	0.45	0.96 (0.88-1.04)	0.33
Social functioning	0.92 (0.86-0.98)	0.01	0.90 (0.84-0.96)	0.00

Table 4. Uni- and Multivariable Cox regression analyses on the association between change in HRQoL after treatment compared to pre-treatment and survival

 $^{\dagger}adjusted$ for age, gender, socio-economic status, smoking (packyears), alcohol abuse (current or history), comorbidity, tumor site, tumor stage, treatment

* p-value of the log likelihood

 Δ change compared to pre-treatment.

Worse physical and emotional function at 6 and 12 months after treatment compared to pre-treatment was significantly associated with shorter survival. The association between physical functioning and survival has been shown in previous studies, also in patients with cancer types other than HNC.^{29,30,42} For instance, a recent study in patients with advanced colorectal cancer revealed that physical functioning assessed with patient-reported outcomes had more prognostic value in predicting overall survival than physician assessed world health organization (WHO) performance status.⁴³

The association between emotional functioning and survival corresponds with findings from a previous longitudinal study in a large cohort of patients HNC showing an significant association between depressive symptoms and shorter survival.⁴⁴ These findings in HNC patients confirm the association between depression and survival in the community and disease specific populations.^{45,46}

In addition to deteriorations in physical and emotional functioning, deteriorations in global QOL and social functioning at 12 months after treatment were also associated with reduced survival. Perhaps, reduced physical and emotional functioning over time also affects global QOL and social functioning. Shortly after diagnoses these problems could be more thoroughly present in patients' lives, where the effects on social or global QoL is postponed. However, when acute symptoms have stabilized after 12 months^{1,2,6,9} patients' will be more aware of the persistent effects of HNC and its treatment and the consequences on their social life and global QoL. On the other hand, patients with advanced illness could also not be able to perform in social activities.

Based on our results, monitoring changes in HRQOL (especially physical and emotional functioning) over time in clinical practice seems important, as these scores may be sensitive for signaling clinical deterioration. Symptom monitoring (such as dyspnea, fatigue and pain) in routine care of patients seems to be associated with increased **survival compared to usual care**.⁴⁷ This can be explained by the early responses of nurses to symptom alerts with clinical interventions, and better chemotherapy **toleration compared to the usual care** group.⁴⁷

Strengths of our study include the large prospective sample of newly diagnosed patient with HNC, allowing to incorporate multiple relevant demographic, lifestyle-related and clinical factors in our statistical models, including HPV status. Another strength is that we investigated the association between survival and HRQoL at different time points before and after treatment. However, some limitations must be noted. We included only patients that received primary or adjuvant (chemo)radiotherapy, and thus excluded patients treated with surgery. Also, the study cohort was treated before 2010, thus not including patients who were treated by recent improvements in (chemo)radiotherapy. These limitations may hamper generalizability of the results. Furthermore, because demographic, lifestyle-related, and clinical variables were retrieved from medical records, we may have missed other important variables that may be predictive for survival such as physical activity, nutritional intake, or marital **status**, **income and occupation**.⁴⁸ Finally, we were unable to retrieve data on disease-specific survival, which limited our analysis to overall survival.

In conclusion, (change in) HRQoL is significantly associated with survival in addition to demographical, lifestyle-related and clinical measures, not only pre-treatment, but also 6 weeks, 6 months and 12 months after treatment. This highlights the value of monitoring HRQoL in (clinical) practice to identify those patients that report changes in HRQOL at 6 and 12 months after treatment. This may help to further improve cancer care in a timely and efficient manner.

REFERENCES

- Abendstein H, Nordgren M, Boysen M, et al: Quality of life and head and neck cancer: a 5 year prospective study. Laryngoscope 115:2183-2192, 2005
- 2. Bjordal K, Ahlner-Elmqvist M, Hammerlid E, et al: A prospective study of quality of life in head and neck cancer patients. Part II: Longitudinal data. Laryngoscope 111:1440-1452, 2001
- 3. Borghgraef K, Delaere P, Van den Bogaert W, et al: Quality of life in head and neck cancer patients. Acta Otorhinolaryngol. Belg 51:69-72, 1997
- 4. de Graeff A, de Leeuw JR, Ros WJ, et al: Long-term quality of life of patients with head and neck cancer. Laryngoscope 110:98-106, 2000
- 5. Funk GF, Karnell LH, Christensen AJ: Long-term health-related quality of life in survivors of head and neck cancer. Arch. Otolaryngol. Head Neck Surg 138:123-133, 2012
- 6. Hammerlid E, Bjordal K, Ahlner-Elmqvist M, et al: A prospective study of quality of life in head and neck cancer patients. Part I: at diagnosis. Laryngoscope 111:669-680, 2001
- Oskam IM, Verdonck-de Leeuw IM, Aaronson NK, et al: Prospective evaluation of health-related quality of life in long-term oral and oropharyngeal cancer survivors and the perceived need for supportive care. Oral Oncol 49:443-448, 2013
- 8. Rogers SN, Ahad SA, Murphy AP: A structured review and theme analysis of papers published on 'quality of life' in head and neck cancer: 2000-2005. Oral Oncol 43:843-868, 2007
- Verdonck-de Leeuw IM, Buffart LM, Heymans MW, et al: The course of health-related quality of life in head and neck cancer patients treated with chemoradiation: a prospective cohort study. Radiother. Oncol 110:422-428, 2014
- 10. Goldstein DP, Hynds KL, Christensen AJ, et al: Health-related quality of life profiles based on survivorship status for head and neck cancer patients. Head Neck 29:221-229, 2007
- 11. de Graeff A, de Leeuw JR, Ros WJ, et al: Pretreatment factors predicting quality of life after treatment for head and neck cancer. Head Neck 22:398-407, 2000
- 12. Fang FM, Liu YT, Tang Y, et al: Quality of life as a survival predictor for patients with advanced head and neck carcinoma treated with radiotherapy. Cancer 100:425-432, 2004
- Karvonen-Gutierrez CA, Ronis DL, Fowler KE, et al: Quality of life scores predict survival among patients with head and neck cancer. J. Clin. Oncol 26:2754-2760, 2008
- 14. Mehanna HM, Morton RP: Does quality of life predict long-term survival in patients with head and neck cancer? Arch. Otolaryngol. Head Neck Surg 132:27-31, 2006
- 15. Meyer F, Fortin A, Gelinas M, et al: Health-related quality of life as a survival predictor for patients with localized head and neck cancer treated with radiation therapy. J. Clin. Oncol 27:2970-2976, 2009
- Oskam IM, Verdonck-de Leeuw IM, Aaronson NK, et al: Quality of life as predictor of survival: a prospective study on patients treated with combined surgery and radiotherapy for advanced oral and oropharyngeal cancer. Radiother. Oncol 97:258-262, 2010
- 17. Osthus AA, Aarstad AK, Olofsson J, et al: Head and neck specific Health Related Quality of Life scores predict subsequent survival in successfully treated head and neck cancer patients: a prospective cohort study. Oral Oncol 47:974-979, 2011
- Osthus AA, Aarstad AK, Olofsson J, et al: Prediction of survival by pretreatment health-related qualityof-life scores in a prospective cohort of patients with head and neck squamous cell carcinoma. JAMA Otolaryngol. Head Neck Surg 139:14-20, 2013
- 19. Aarstad HJ, Osthus AA, Aarstad HH, et al: General health-related quality of life scores from head and neck squamous cell carcinoma patients obtained throughout the first year following diagnosis predicted

up to 10-year overall survival. Eur Arch Otorhinolaryngol 275:207-217, 2018

- 20. Yang CJ, Roh JL, Kim MJ, et al: Pretreatment quality of life as a prognostic factor for early survival and functional outcomes in patients with head and neck cancer. Qual Life Res 25:165-74, 2016
- 21. van Nieuwenhuizen AJ, Buffart LM, Brug J, et al: The association between health related quality of life and survival in patients with head and neck cancer: a systematic review. Oral Oncol 51:1-11, 2015
- 22. Bjordal K, Ahlner-Elmqvist M, Tollesson E, et al: Development of a European Organization for Research and Treatment of Cancer (EORTC) questionnaire module to be used in quality of life assessments in head and neck cancer patients. EORTC Quality of Life Study Group. Acta Oncol 33:879-85, 1994
- 23. http://www.scp.nl/Onderzoek/Lopend_onderzoek/A_Z_alle_lopende_onderzoeken/Statusscores, 2014
- 24. Nauta IH, Rietbergen MM, van Bokhoven A, et al: Evaluation of the eighth TNM classification on p16positive oropharyngeal squamous cell carcinomas in the Netherlands and the importance of additional HPV DNA testing. Ann Oncol 29:1273-1279, 2018
- Rietbergen MM, Leemans CR, Bloemena E, et al: Increasing prevalence rates of HPV attributable oropharyngeal squamous cell carcinomas in the Netherlands as assessed by a validated test algorithm. Int J Cancer 132:1565-71, 2013
- Piccirillo JF, Tierney RM, Costas I, et al: Prognostic importance of comorbidity in a hospital-based cancer registry. JAMA 291:2441-2447, 2004
- 27. Osoba D, Rodrigues G, Myles J, et al: Interpreting the significance of changes in health-related qualityof-life scores. J Clin Oncol 16:139-44, 1998
- 28. Grignon LM, Jameson MJ, Karnell LH, et al: General health measures and long-term survival in patients with head and neck cancer. Arch. Otolaryngol. Head Neck Surg 133:471-476, 2007
- 29. Montazeri A: Quality of life data as prognostic indicators of survival in cancer patients: an overview of the literature from 1982 to 2008. Health Qual. Life Outcomes 7:102, 2009
- 30. Quinten C, Coens C, Mauer M, et al: Baseline quality of life as a prognostic indicator of survival: a meta-analysis of individual patient data from EORTC clinical trials. Lancet Oncol 10:865-871, 2009
- A. dG, de Leeuw JR, Ros WJ, et al: A prospective study on quality of life of patients with cancer of the oral cavity or oropharynx treated with surgery with or without radiotherapy. Oral Oncol 35:27-32, 1999
- 32. Lango MN, Egleston B, Fang C, et al: Baseline health perceptions, dysphagia, and survival in patients with head and neck cancer. Cancer 120:840-847, 2014
- 33. Coyne JC, Pajak TF, Harris J, et al: Emotional well-being does not predict survival in head and neck cancer patients: a Radiation Therapy Oncology Group study. Cancer 110:2568-2575, 2007
- 34. Curran D, Giralt J, Harari PM, et al: Quality of life in head and neck cancer patients after treatment with high-dose radiotherapy alone or in combination with cetuximab. J. Clin. Oncol 25:2191-2197, 2007
- 35. Fang FM, Tsai WL, Chien CY, et al: Pretreatment quality of life as a predictor of distant metastasis and survival for patients with nasopharyngeal carcinoma. Journal of Clinical Oncology 28:4384-4389, 2010
- Nordgren M, Abendstein H, Jannert M, et al: Health-related quality of life five years after diagnosis of laryngeal carcinoma. Int J Radiat Oncol Biol. Phys 56:1333-1343, 2003
- 37. Nordgren M, Hammerlid E, Bjordal K, et al: Quality of life in oral carcinoma: a 5-year prospective study. Head Neck 30:461-470, 2008
- Nordgren M, Jannert M, Boysen M, et al: Health-related quality of life in patients with pharyngeal carcinoma: a five-year follow-up. Head Neck 28:339-349, 2006
- Siddiqui F, Pajak TF, Watkins-Bruner D, et al: Pretreatment quality of life predicts for locoregional control in head and neck cancer patients: a radiation therapy oncology group analysis. Int. J. Radiat. Oncol. Biol. Phys 70:353-360, 2008

- 40. Tarsitano A, Pizzigallo A, Ballone E, et al: Health-related quality of life as a survival predictor for patients with oral cancer: is quality of life associated with long-term overall survival? Oral Surg. Oral Med Oral Pathol. Oral Radiol 114:756-763, 2012
- 41. Thompson TL, Pagedar NA, Karnell LH, et al: Factors associated with mortality in 2-year survivors of head and neck cancer. Arch. Otolaryngol. Head Neck Surg 137:1100-1105, 2011
- 42. Quinten C, Martinelli F, Coens C, et al: A global analysis of multitrial data investigating quality of life and symptoms as prognostic factors for survival in different tumor sites. Cancer 120:302-311, 2013
- 43. Mol L, Ottevanger PB, Koopman M, et al: The prognostic value of WHO performance status in relation to quality of life in advanced colorectal cancer patients. Eur J Cancer 66:138-43, 2016
- 44. Jansen F, Verdonck-de Leeuw IM, Cuijpers P, et al: Depressive symptoms in relation to overall survival in people with head and neck cancer: A longitudinal cohort study. Psychooncology 27:2245-2256, 2018
- 45. Cuijpers P, Vogelzangs N, Twisk J, et al: Is excess mortality higher in depressed men than in depressed women? A meta-analytic comparison. J Affect Disord 161:47-54, 2014
- 46. Verdonck-de Leeuw IM, Cuijpers P, Leemans CR: Pretreatment depression as a prognostic indicator of survival and nutritional status in patients with head and neck cancer. Cancer 122:971-2, 2016
- 47. Basch E, Deal AM, Dueck AC, et al: Overall Survival Results of a Trial Assessing Patient-Reported Outcomes for Symptom Monitoring During Routine Cancer Treatment. Jama 318:197-198, 2017
- Duffy SA, Ronis DL, McLean S, et al: Pretreatment health behaviors predict survival among patients with head and neck squamous cell carcinoma. J Clin Oncol 27:1969-1975, 2009



CHAPTER 5

PATIENT-REPORTED PHYSICAL ACTIVITY AND THE ASSOCIATION WITH HEALTH-RELATED QUALITY OF LIFE IN HEAD AND NECK CANCER SURVIVORS

A.J. van Nieuwenhuizen L.M. Buffart C.F. van Uden-Kraan L.A. van der Velden M. Lacko J. Brug C.R. Leemans I.M. Verdonck-de Leeuw

Supportive care in cancer (2017) 11-21

ABSTRACT

Purpose. This study aimed to assess patient-reported levels of physical activity and it's associations with health-related quality of life (HRQoL) adjusted for important demographic, lifestyle-related and clinical factors, among head and neck (HNC) survivors.

Methods. This cross-sectional study included 116 HNC survivors. Physical activity was assessed with the Physical-Activity-Scale-for-the-Elderly (PASE) and HRQoL with the EORTC-QLQ-C30 and EORTC-HN35. Associations were studied using univariable and multivariable regression analyses.

Results. Median PASE score was 100.3 (interquartile range 65.1;170.8) of which 54% were household, 34% leisure time and 12% occupational activities. Younger HNC survivors had higher levels of PA. Higher physical activity was significantly associated with higher global QoL (p < 0.05). Findings for physical function, role function, social function, fatigue and pain were in line, but not statistically significant ($0.05 \le p < 0.10$).

Conclusions. Among HNC survivors, a large proportion of physical activity consists of household activities. Younger HNC survivors had higher physical activity, and higher physical activity levels were associated with higher HRQoL.

INTRODUCTION

Worldwide, the incidence of head and neck cancer (HNC) has increased over the past decades and five year survival rates have improved in Europe¹ and the United States.² As a consequence, more HNC survivors have to cope with physical and psychosocial problems and HNC specific symptoms associated with cancer and its treatment, such as oral dysfunction, swallowing and speech problems, severely compromising health-related quality of life (HRQoL).³⁻⁶

Randomized controlled trials in patients with other types of cancers, mainly breast and prostate cancer, showed that physical activity (PA) can reduce physical and psychosocial problems and improve HRQoL.⁷⁻⁹ Observational studies showed that higher levels of moderate-to-vigorous PA are associated with lower mortality risk in survivors of breast, colon and prostate cancer.¹⁰⁻¹⁴ Also in HNC survivors, higher pretreatment levels of PA and physical function were found to be associated with higher HRQoL^{15, 16} and survival.^{17, 18}

However, PA levels of HNC patients tend to decrease following diagnosis and during treatment.^{16, 19-21} Two previous studies^{16, 20} examined demographic, clinical and lifestyle correlates of PA and had contradictory results. In a sample of 59 HNC survivors Rogers et al.²⁰ found that younger age, the absence of comorbidity and abstinence from alcohol were related to higher levels of patient-reported PA. Sammut et al.¹⁶ found no associations of gender, smoking, comorbidity and age with weekly energy expenditure after treatment in a sample of 172 HNC survivors. Insight into demographic and clinical correlates of PA, may help to identify which subgroups of HNC survivors are more likely to have low PA.

At present, the number of studies evaluating the associations between PA and HRQoL in patients with HNC are scarce, especially as compared to patients with other types of cancer such as breast or prostate.¹⁵ Furthermore, these studies could only include leisure time PA and no data on household or occupational activities were included. Rogers at al.²⁰ found higher levels of leisure time PA, 18.6 (SD 50.9) months after treatment, to be associated with lower fatigue, higher HRQoL and higher functional wellbeing after adjusting for age, presence of comorbidities and alcohol consumption. Sammut et al.¹⁶ reported significant positive correlations between higher levels of PA at 12.9 (SD 12.8) months after treatment and higher HRQoL.

Because of the scarcity of evidence regarding PA levels and the association with HRQoL among HNC survivors, the present study aimed to (1) describe the level of PA among HNC survivors, including leisure-time, household and occupational PA, (2) study demographic, clinical, and lifestyle-related correlates of PA and (3) assess the association between PA and HRQoL adjusted for important demographic, clinical, and lifestyle-related for important demogra

MATERIALS AND METHODS

Setting and patient recruitment

In this cross-sectional study patients were recruited between January and September 2013 from the Departments of Otolaryngology-Head and Neck Surgery from VU University Medical Center, Amsterdam, Leiden University Medical Center and Maastricht University Medical Center. We included data of PA and HRQoL from two separate studies, the OncoQuest study²² and the OncoKompas²³ study. At the VU University Medical Center, Amsterdam, the OncoQuest system is implemented as part of standard care, to assess HRQoL in patients with HNC. Additionally, the OncoKompas study was launched, which is an online self-management application where cancer survivors can monitor their HRQoL and get tailored feedback and personalized advice on supportive care services. The HRQoL questionnaires we included for the current study were administered before the online self-management application was carried out. The OncoQuest study and the OncoKompas study included the same HRQoL questionnaires. To be able to answer our research questions on PA in HNC survivors and associations with HRQoL we added the PASE questionnaire for a limited number of time in both studies. Eligibility criteria and patient recruitment of both studies are presented in Figure 1. Patients were eligible for this cross-sectional study if they were 1) diagnosed with HNC, 2) treated with surgery, radiotherapy, chemoradiation or a combination of these treatments,3) aged 18 years or older, and 4) able to write, read and speak Dutch. Patients were excluded if they were diagnosed with basal cell carcinoma's or lymphoma in the head and neck region, or if they suffered from severe psychiatric co-morbidities (e.g. schizophrenia, Korsakov's syndrome, severe dementia). All patients signed an informed consent statement prior to participation. The study was conducted according to regular procedures of the local ethical committee of the VU University Medical Center, Amsterdam.

Outcome measurements

Physical activity

PA was assessed with the 13-item Physical Activity Scale for the Elderly (PASE), a self-administered 1-week recall questionnaire on leisure time, household, and occupational physical activities.²⁴ The frequency of these activities was recorded as never, seldom (1-2 days a week), sometimes (3-4 days a week) or often (5-7 days a week). The duration of activities was categorized as less than 1 hour, between 1 and 2 hours, between 2 and 4 hours, or more than 4 hours. Paid or volunteer work, except for work that involved mostly sitting activities such as office work, was categorized as less than 1 hour, between 1 and 4 hours, between 5 and 8 hours, or more than 8 hours.²⁵ The total PASE sum score was computed by multiplying the amount of time spent on

each activity (hours/week) by the empirical derived item weights and summing over all activities.²⁴⁻²⁶ The PASE was shown to have good to excellent test-retest reliability, and good content validity among patients with cancer with an average age of 50 (SD 12). Its construct validity (with accelerometers as comparison measure) was comparable to other PA questionnaires.²⁷

Health-Related Quality of Life

HRQoL was assessed with the European Organization for Research and Treatment of Cancer, (EORTC) Quality of Life Questionnaire core module (EORTC QLQ-C30) and the Tumor specific HRQoL was assessed by the EORTC Head and Neck Module (EORTC HN35).²⁶ The EORTC QLQ-C30 is a 30-item questionnaire including a global QoL scale, five function scales, three symptom scales and 6 single items, with higher scores presenting higher global QoL and function, and lower scores presenting higher symptom severity.²⁹ The EORTC HN35 is a 35-item module including HNC specific symptom scales and 10 single items covering several problems.

Demographic, clinical and lifestyle-related factors

Demographic, clinical, and lifestyle-related factors were collected from medical records and included gender, age, zip codes of patients' living area, smoking (pack years, current smoker), alcohol consumption (units per day, current or former abuse (≥5 units a day)), tumor site (oral cavity, oropharynx, hypopharynx, larynx and other), type (squamous cell carcinoma vs. non-squamous cell carcinoma) and stage (I, II, III and IV), tumor recurrence (dichotomized as none versus any, including local, regional and second or third primary tumors), treatment modality (surgery, (chemo) radiotherapy, or surgery followed by (chemo)radiotherapy), time since completion of treatment (months) and comorbidities.

Socio-economic status (SES) was determined using zip codes of patients' living area. Zip codes were translated to SES according to The Netherlands Institute for Social Research.³⁰ This system describes the social status of a district compared to other districts in The Netherlands using an algorithm based on mean income, percentage of people with low income, percentage of people with low education and percentage of people with without a job. Therefore the mean score of all districts in The Netherlands is zero. We dichotomized SES scores to high (> mean value) versus low (\leq mean value). Comorbidities were assessed using the Adult Comorbidity Evaluation 27 (ACE-27), a validated chart built instrument examining the presence of any of the following medical conditions: cardiovascular, respiratory, gastro-intestinal, renal, endocrine, neurological, immunological, previous malignancies, psychiatric disorders, alcohol use and severe overweight, resulting in a total comorbidity score of none, mild, moderate or severe.³¹

Statistical analysis

Descriptive statistics (mean, standard deviation (SD), or numbers and percentages) were generated for demographic, lifestyle-related, and clinical factors, PA and HRQoL. For the continuous variables median and interquartile range (IQR) were reported when outcomes were not normally distributed (skewness scores < -1; > 1). Since total PA score was skewed to the right and the residuals obtained in the regression analysis were not normally distributed, we presented data of total PA as median (IQR) and natural log-transformed the data for analyzing the correlates of PA. We conducted univariable and multivariable linear regression analyses (presenting confidence intervals and standardized regression coefficients) to study demographic, lifestylerelated and clinical correlates of PA. No multicollinearity (rp > 0.75) was found. To determine the maximum number of variables to be included in the regression model, we used the rule of thumb of 10 patients per determinant. Consequently, our sample of 116 allowed to include a maximum of 11 variable into the regression model. To prevent overfitting in the multiple linear regression model, we selected variables using a forward selection procedure starting with the variable that most strongly predicted PA. Variables were selected one by one and all variables with $p \le 0.05$ were inserted in the multiple regression model. We back transformed the results from the final model indicating ratios. The associations between PA and HRQoL were assessed using univariable and multivariable linear regression analyses. For the most accurate estimate of the association, we adjusted for demographic, and lifestyle-related, and clinical characteristics. Due to the maximum number of variables allowed in the regression model, we have chosen tumor stage over tumor location and tumor type because it is more strongly associated with quality of life.³⁻⁶ We explored interactions for the main demographic and clinical characteristics (age, gender, cancer stage and treatment) to study whether the association between PA and HRQoL differed between these subgroups. To limit the number of interactions explored, we tested interactions when the associations between PA and HRQOL had a p-value < 0.10. P-values ≤ 0.05 were considered statistically significant.

Figure 1. Inclusion flowchart



Abbreviations: HNC, head and neck cancer; LUMC, Leiden University Medical Center; MUMC, Maastricht University Medical Center; n, number; VUmc, PRO, patient reported outcome; RR, response rate; VUmc, VU University medical center

Characteristics	Participants (n= 116)
Demographic factors	
Gender, n (%) male	73 (63)
Age, mean (SD) years	60 (10)
SES, mean (SD)	0.54 (0.9)
High SES (above average), n (%)	34 (29)
Lifestyle-related	
Smoking (packyears), median (IQR)	20 (0-40)
Smoking at diagnosis, n (%)	63 (54)
Alcohol use (units per day), mean (SD)	1,5 (2.3)
Alcohol abusea at diagnosis, n (%)	23 (20)
Clinical factors	
Tumor location, n (%)	
Oral Cavity and oropharynx	56 (48)
Larynx and hypopharynx	33 (29)
Otherb	27 (23)
Cancer type, n (%)	
Squamous cell	105 (91)
Non-squamous cell	11 (9)
Disease Stage, n (%)	
I and II	41 (35)
III and IV	75 (65)
Type of treatment, n (%)	
Surgery only	21 (18)
Radiotherapy	28 (24)
Chemoradiotherapy	23 (20)
Surgery combined with (chemo)radiation therapy	44 (38)
Recurrence, n (%)	
None	95 (82)
Any	21 (18)
Comorbidity, n (%)	
None or mild	74 (64)
Moderate or severe	42 (36)
Time since treatment, mediam (IQR) months	14 (7-23)
Physical activity	
Total score, median (IQR)	100.3 (65.1-170.8)
Leisure time activities (% of total PA)	34%

 $\label{eq:table_to_state} \textbf{Table 1}. Demographic, lifestyle-related and clinical characteristics, physical activity (PA) and health-related quality of life (HRQoL)$

Table 1. Continued

Characteristics	Participants (n= 116)
Household activities	54%
Occupational activities	12%
Cancer specific HRQoL, mean (SD)	
Global quality of life	78.2 (15.9)
Physical function	88.0 (13.7)
Role function	85.5 (19.6)
Emotional function	85.9 (14.0)
Cognitive function	89.7 (14.9)
Social function	85.6 (19.0)
Fatigue	23.9 (21.5)
Pain (general)	14.2 (19.9)
Dyspnea	14.4 (22.9)
Insomnia	16.1 (35.4)
Loss of appetite	6.0 (17.9)
Constipation	9.2 (20.9)
Diarrhea	5.2 (13.6)
Financial problems	9.8 (21.1)
Tumor specific HRQoL, mean (SD)	
Pain (mouth)	17.0 (20.9)
Swallowing	16.7 (23.4)
Senses	21.3 (23.4)
Speech problems	16.4 (20.5)
Social eating	15.1 (22.5)
Social contact	5.1 (8.9)
Sexuality	24.5 (29.7)
Teeth	13.9 (24.2)
Opening mouth	13.5 (22.4)
Dry mouth	42.8 (31.3)
Sticky saliva	30.4 (32.3)
Coughing	20.7 (27.3)
Feel ill	9.8 (19.7)

Abbreviations: CRT = chemoradiation; EORTC QLQ C30 = European Organization for Research and Treatment of Cancer Quality of Life Questionnaire Core 30;n: number; RT = radiotherapy; SD: standard deviation; SES: socio-economic status; Surg = surgery;

a) Alcohol abuse defined as ≥ 5 units of alcohol per day

b) Unknown primary, nasopharynx, nasal cavity, nasal sinus, salivary glands, ear and skin

RESULTS

In total, 116 out of 212 HNC survivors met our inclusion criteria and filled out the survey on PA and HRQoL (Figure 1). Mean (SD) age of participants was 60 (10) years and 63% were men. The most frequent tumor site was oropharynx (26%) followed by oral cavity (22%), larynx (22%) and hypopharynx (7%). Most patients were treated by a combination of treatment modalities (58%). Time since completion of treatment was 21 (21) months. Quality of life scores ranged from 78.2 (global quality of life) to 89.7 (cognitive functioning). Regarding cancer specific HRQoL symptom scores ranged from 5.1 (Social contact) to 24.5 (sexuality, Table 1).

Median (IQR) total PASE score was 100.3 (65.1; 170.8), of which 34% consisted of leisure time PA, 54% of household activities, and 12% occupational activities. A younger age was significantly associated with higher levels of PA ($\beta = 0.98, 95\%$ CI = 0.96; 1.00) explaining 5.2% of the variance in PA (Table 2). No significant associations with PA were found for other demographic, clinical, or lifestyle-related variables.

After adjusting for age, gender, SES, smoking, alcohol abuse, comorbidity, tumor stage, treatment modality, recurrence and time since treatment, a higher level of PA was significantly associated with higher global QoL (β : 0.06, 95% CI = 0.03; 0.10). Possible meaningful association were also observed for higher physical function (β : 0.03, 95% CI = -0.00; 0.06), role function (β : 0.04, 95% CI = -0.00; 0.09) and social function (β : 0.04, 95% CI = -0.00; 0.09) and social function (β : 0.04, 95% CI = -0.00; 0.09) and lower level of fatigue (β : -0.05, 95% CI = -0.10; 0.00) and less pain (β : -0.04, 95% CI = -0.09; 0.00), but these associations were not statistically significant (0.05 \leq p < 0.10, Table 3). Explorative analyses showed that gender was a significant effect modifier in the association between PA and general pain ($\beta_{interaction}$ =-0.09, 95% CI = -0.18; -0.005, p= 0.04). Explorative stratified analyses for gender revealed a significant association between PA and general pain (β =-0.11, 95% CI = -0.19; -0.03, p= 0.01), while the association was not statistically significant in men (β =-0.02, 95% CI = -0.07; 0.03, p= 0.46). We also found a significant effect modifier association in patients who were younger ($\beta_{interaction}$ =-0.005, 95% CI = 0.00; 0.01, p= 0.04).

	Ratio (95% CI)	P value	Standardized regression coefficients
Demographic factors			
Gender	1.28 (0.93; 1.77)	0.13	0.14
Age, years	0.98 (0.96; 1.00)	0.01	-0.23
SES	0.95 (0.68; 1.34)	0.78	-0.03
Lifestyle-related factors			
Smoking, packyears	1.00 (1.00; 1.01)	0.32	0.09
Smoking at diagnosis	1.20 (0.88; 1.64)	0.25	0.11
Alcohol, units per day	0.99 (1.07; 1.06)	0.79	0.02
Alcohol abusea	0.92 (0.62; 1.36)	0.67	-0.04
Clinical factors			
Tumor location			
OC and OP	1.33 (0.89; 1.98)	0.16	0.17
L and HP	1.22 (0.78; 1.89)	0.38	0.10
Other	Ref		
Cancer type	1.51 (0.88; 2.57)	0.13	0.14
Disease Stage	0.98 (0.70; 1.36)	0.89	-0.01
Recurrence	0.69 (0.46; 1.04)	0.07	-0.17
Comorbidity	0.82 (0.59; 1.14)	0.23	-0.11
Type of treatment			
single vs multiple	0.92 (0.67; 1.27)	0.62	-0.05
Time since treatment (months)	0.99 (0.99; 1.00)	0.10	-0.15

Table 2. Demographic, lifestyle-related and clinical correlates of physical activity. Results from univariable regression analyses.

Abbreviations: CI = confidence interval, L and HP = larynx and hypopharynx; OC and OP = oral cavity and oropharynx; SES: socio-economic status. a Alcohol abuse defined as \geq 5 units of alcohol per day

Gender (o=male, 1=female), SES (o = other, 1 = high), current smoking (o = never or former, 1 = current), alcohol abuse (o = no abuse, 1 = current or former abuse), cancer type (o= no squamous cell carcinoma, 1 = squamous cell carcinoma), disease stage (o = stage I and II, 1 = stage III and IV), recurrence (o = no recurrence, 1 = any recurrence), comorbidity (o = none or mild, 1 = moderate or severe)

Table 3. Independent correlates of physical activity. Results of the multivariable regression analyses

	Ratio (95% CI)	P value	Standardized regression coefficients
Demographic factors			
Age, years	0.98 (0.96; 1.00)	0.01	-0.23

Variables were selected one by one and all variables with $\rm p \le 0.05$ were inserted in the multiple regression model.

	Univariable analyses	P-value	Multivariable analyses ^a	P-value
HRQoL	β (95% CI)		β (95% CI)	
Global quality of life	0.059 (0.025; 0.092)	0.01	0.061 (0.025; 0.096)	0.00
Physical function	0.034 (0.004; 0.063)	0.03	0.027 (-0.003; 0.057)	0.08
Role functioning	0.040 (-0.003; 0.082)	0.07	0.044 (-0.001; 0.089)	0.05
Emotional functioning	0.011 (-0.020;0.042)	0.49	0.004 (-0.030; 0.038)	0.81
Social functioning	0.034 (-0.008; 0.075)	0.11	0.043 (-0.002; 0.088)	0.06
Cognitive functioning	0.019 (-0.013; 0.052)	0.24	0.018 (-0.019; 0.055)	0.33
Fatigue	-0.043 (-0.090; 0.04)	0.07	-0.050 (-0.103; 0.003)	0.06
Nausea and vomiting	0.000 (-0.017; 0.017)	0.97	-0.004 (-0.024; 0.015)	0.65
Pain (general)	-0.050 (-0.092;-0.007)	0.02	-0.044 (-0.091; 0.003)	0.06
Dyspnoe	0.013 (-0.038; 0.063)	0.62	0.050 (-0.046; 0.056)	0.84
Insomnia	-0.027 (-0.083; 0.029)	0.34	-0.044 (-1.05; 0.017)	0.15
Loss of appetite	0.012 (-0.027; 0.052)	0.54	0.001 (-0.042; 0.045)	0.96
Constipation	-0.026 (-0.072; 0.020)	0.27	-0.021 (-0.072; 0.030)	0.42
Diarrhea	-0.017 (-0.046; 0.013)	0.27	-0.019 (-0.053; 0.015)	0.28
Financial problems	0.09 (-0.037; 0.056)	0.67	0.016 (-0.036; 0.068)	0.54
Tumor specific HRQoL				
Pain (mouth)	-0.015 (-0.061; 0.031)	0.53	-0.009 (-0.060; 0.043)	0.74
Swallowing	-0.022 (-0.074; 0.029)	0.39	-0.022 (-0.073; 0.029)	0.39
Senses	-0.028 (-0.079; 0.024)	0.29	-0.019 (-0.076; 0.038)	0.51
Speech problems	-0.005 (-0.050; 0.040)	0.83	0.004 (-0.045; 0.054)	0.86
Social eating	-0.022 (-0.071; 0.027)	0.38	-0.021 (-0.071; 0.030)	0.42
Social contact	-0.008 (-0.028; 0.012)	0.43	-0.005 (-0.026; 0.017)	0.68
Sexuality	-0.037 (-0.105; 0.031)	0.29	-0.048 (-0.118; 0.022)	0.18
Teeth	0.028 (-0.025; 0.081)	0.30	0.019 (-0.039; 0.077)	0.51
Opening mouth	-0.031 (-0.080; 0.018)	0.21	-0.031 (-0.084; 0.023)	0.26
Dry mouth	0.009 (-0.060; 0.078)	0.80	0.001 (-0.073; 0.075)	0.99
Sticky saliva	0.005 (-0.066; 0.077)	0.88	0.007 (-0.069; 0.083)	0.86
Coughing	-0.028 (-0.088; 0.032)	0.36	-0.026 (-0.090; 0.038)	0.42
Feeling ill	-0.018 (-0.061; 0.026)	0.42	-0.032 (-0.082; 0.016)	0.30

Table 4. The association between physical activity and health-related quality of life (HRQoL)

a) adjusted for age, gender, socio-economic status, smoking, alcohol abuse, comorbidity, tumor stage, treatment, recurrence and time since treatment $% \lambda =0$

DISCUSSION

This cross-sectional study describes self-reported PA levels among HNC survivors, the demographic, lifestyle-related and clinical correlates of PA, and the association between PA and HRQoL. Our median PASE score (100.3, IQR 65.1; 170.8) was comparable to the HNC population (n = 283) in a study from Duffy et al,¹⁹ that reported s mean of 115 pre-treatment and a mean of 106 and 110 at 6 and 9 months after treatment, respectively. However, compared to a non-cancer elderly population also using the PASE questionnaire,²⁴ the population of HNC survivors had lower levels of PA (144.9 vs. 100.3). The findings that HNC survivors are at increased risk for low PA levels and the positive association between PA and HRQoL highlights the relevance for evaluating interventions that aim to improve PA levels in this population.³²

In the study, total PA mainly consisted of household activity (55%). This is comparable with studies in general populations, reporting that 30-60% of total PA consists of household activities^{24, 33, 34} and this proportion tends to increase with age.³³ Because of their significant contribution to total PA levels, it is important to also assess household and occupational activities, and not just leisure time PA as is often the case. Also, for interventions aiming to improve PA levels in HNC survivors, it might be useful to focus on promoting PA during daily routines, especially because HNC survivors reported to prefer exercising alone, unsupervised and at a moderate intensity.³⁴

Our finding that older HNC survivors are less physically active is in line with previous studies among HNC survivors²⁰ as well as in survivors of other types of cancer.³⁶⁻³⁸ This illustrates that it is important to promote PA interventions in elderly (head and neck) cancer survivors, particularly, because they are at high risk for functional decline after cancer diagnosis.³⁹ Unfortunately, current interventions to promote PA may not always reach elderly cancer survivors.⁴⁰ We found no significant associations regarding PA and other demographic factors (gender and SES) which is comparable to the studies of Rogers et al.²⁰ and Sammut et al.¹⁶ However, it should be noted that the information on SES in our study was limited because we estimated SES based on ZIP codes, and did not ask patients to provide data on education or income themselves).

In contrast to previous studies, we found no evidence for an association of smoking (^{37, 38}), alcohol consumption^{20, 38} or clinical factors with PA.^{20, 37, 38, 41} The lack of significant associations for clinical factors (e.g. comorbidity, tumor location, and type of treatment) may indicate that the impact of clinical factors reduces over time and other factors such as motivational factors become more important.⁴¹⁻⁴⁴ Future prospective longitudinal studies with objective PA measurements should further clarify whether these associations (demographic, clinical and lifestyle-related) might be present in HNC survivors or if these are only present during or shortly after treatment.

Our finding that a higher PA level was associated with higher global QoL, and possibly better physical function, role function, social function, and less fatigue and pain among HNC survivors is consistent with previous studies.^{16, 20} This indicates that improving PA might be an intervention target to improve HRQoL. However, due to the cross-sectional design, it is not possible to make causal inferences and it is unclear whether improving PA levels would improve HRQoL, or whether HNC survivors with lower HRQoL are less physically active. In contrast to general HRQoL, the current study found no support for an association of PA and HNC-specific HRQoL. Due to the cross-sectional nature of this study, it may also suggests that HNC symptoms are not a barrier to PA. Several small studies have shown that PA interventions among HNC survivors are feasible and may improve general and HNC specific HRQoL.¹⁵

Strengths of our study are the relatively large sample size of an understudied group of cancer survivors, allowing to adjust analysis for important demographic, lifestyle related and clinical factors. We could also include levels of PA originating from household or occupational activities in addition to leisure time PA. However, some limitations must be noted. First, the use of a self-reported questionnaire to assess PA levels is susceptible to recall and social desirability bias.⁴⁵ This may have led to an overestimation of PA levels, and therefore the absolute PA level should be interpreted with caution. However, the PASE questionnaire is a valid measure to distinguish active from inactive people,²⁷ and therefore the direction of the associations may be considered valid. Second we assessed lifestyle-related and clinical factors only at diagnosis, and some of these outcomes might have changed at the time (e.g. smoking, alcohol use, comorbidity) of the questionnaire assessment. Third, the lack of associations of PA with clinical factors may be related to the sample size. However, the wide confidence intervals indicate heterogeneity in the association and standardized regression coefficients were small (≤ 0.23), which makes it is unlikely that associations will be significant and clinically relevant with larger samples. Finally, the cross-sectional design hampered us to draw conclusions about causality, and future studies are needed to investigate whether increasing PA would improve HRQoL.

In conclusion, in this cross-sectional study we found that HNC survivors, and particularly older survivors, are at risk for low levels of PA. Among HNC survivors, a large proportion of PA consists of household activities. HNC survivors with higher levels of PA had higher global QoL and role function. Future studies should investigate the causality of these associations.

REFERENCES

- Karim-Kos HE, de VE, Soerjomataram I, Lemmens V, Siesling S, Coebergh JW. Recent trends of cancer in Europe: a combined approach of incidence, survival and mortality for 17 cancer sites since the 1990s. Eur J Cancer. 2008;44(10):1345-89.
- 2. Pulte D, Brenner H. Changes in survival in head and neck cancers in the late 20th and early 21st century: a period analysis. Oncologist. 2010;15(9):994-1001.
- 3. Rogers SN, Ahad SA, Murphy AP. A structured review and theme analysis of papers published on 'quality of life' in head and neck cancer: 2000-2005. Oral Oncol. 2007;43(9):843-68.
- Hammerlid E, Silander E, Hornestam L, Sullivan M. Health-related quality of life three years after diagnosis of head and neck cancer--a longitudinal study. Head Neck. 2001;23(2):113-25.
- Funk GF, Karnell LH, Christensen AJ. Long-term health-related quality of life in survivors of head and neck cancer. Arch Otolaryngol Head Neck Surg. 2012;138(2):123-33.
- Bjordal K, Ahlner-Elmqvist M, Hammerlid E, Boysen M, Evensen JF, Biorklund A, et al. A prospective study of quality of life in head and neck cancer patients. Part II: Longitudinal data. Laryngoscope. 2001;111(8):1440-52.
- Speck RM, Courneya KS, Masse LC, Duval S, Schmitz KH. An update of controlled physical activity trials in cancer survivors: a systematic review and meta-analysis. J Cancer Surviv. 2010;4(2):87-100.
- Mishra SI, Scherer RW, Geigle PM, Berlanstein DR, Topaloglu O, Gotay CC, et al. Exercise interventions on health-related quality of life for cancer survivors. Cochrane Database Syst Rev. 2012(8):CD007566.
- 9. Ferrer RA, Huedo-Medina TB, Johnson BT, Ryan S, Pescatello LS. Exercise interventions for cancer survivors: a meta-analysis of quality of life outcomes. Ann Behav Med. 2011;41(1):32-47.
- Holmes MD, Chen WY, Feskanich D, Kroenke CH, Colditz GA. Physical activity and survival after breast cancer diagnosis. JAMA. 2005;293(20):2479-86.
- 11. Kenfield SA, Stampfer MJ, Giovannucci E, Chan JM. Physical activity and survival after prostate cancer diagnosis in the health professionals follow-up study. J Clin Oncol. 2011;29(6):726-32.
- Meyerhardt JA, Giovannucci EL, Ogino S, Kirkner GJ, Chan AT, Willett W, et al. Physical activity and male colorectal cancer survival. Arch Intern Med. 2009;169(22):2102-8.
- 13. Schmitz KH. Exercise for secondary prevention of breast cancer: moving from evidence to changing clinical practice. Cancer Prev Res (Phila). 2011;4(4):476-80.
- Richman EL, Kenfield SA, Stampfer MJ, Paciorek A, Carroll PR, Chan JM. Physical activity after diagnosis and risk of prostate cancer progression: data from the cancer of the prostate strategic urologic research endeavor. Cancer Res. 2011;71(11):3889-95.
- Capozzi LC, Nishimura KC, McNeely ML, Lau H, Culos-Reed SN. The impact of physical activity on health-related fitness and quality of life for patients with head and neck cancer: a systematic review. Br J Sports Med. 2016;50(6):325-38.
- 16. Sammut L, Fraser LR, Ward MJ, Singh T, Patel NN. Participation in sport and physical activity in head and neck cancer survivors: associations with quality of life. Clin Otolaryngol. 2016;41(3):241-8.
- Duffy SA, Ronis DL, McLean S, Fowler KE, Gruber SB, Wolf GT, et al. Pretreatment health behaviors predict survival among patients with head and neck squamous cell carcinoma. J Clin Oncol. 2009;27(12):1969-75.
- van Nieuwenhuizen AJ, Buffart LM, Brug J, Leemans CR, Verdonck-de Leeuw IM. The association between health related quality of life and survival in patients with head and neck cancer: a systematic review. Oral Oncol. 2015;51(1):1-11.
- 19. Duffy SA, Khan MJ, Ronis DL, Fowler KE, Gruber SB, Wolf GT, et al. Health behaviors of head and neck

cancer patients the first year after diagnosis. Head Neck. 2008;30(1):93-102.

- Rogers LQ, Courneya KS, Robbins KT, Malone J, Seiz A, Koch L, et al. Physical activity and quality of life in head and neck cancer survivors. Support Care Cancer. 2006;14(10):1012-9.
- Silver HJ, Dietrich MS, Murphy BA. Changes in body mass, energy balance, physical function, and inflammatory state in patients with locally advanced head and neck cancer treated with concurrent chemoradiation after low-dose induction chemotherapy. Head Neck. 2007;29(10):893-900.
- 22. Duman-Lubberding S, van Uden-Kraan CF, Jansen F, Witte BI, Eerenstein SEJ, van Weert S, et al. Durable usage of patient-reported outcome measures in clinical practice to monitor health-related quality of life in head and neck cancer patients. Support Care Cancer. 2017.
- 23. Duman-Lubberding S, van Uden-Kraan CF, Jansen F, Witte BI, van der Velden LA, Lacko M, et al. Feasibility of an eHealth application "OncoKompas" to improve personalized survivorship cancer care. Support Care Cancer. 2016;24(5):2163-71.
- 24. Washburn RA, McAuley E, Katula J, Mihalko SL, Boileau RA. The physical activity scale for the elderly (PASE): evidence for validity. J Clin Epidemiol. 1999;52(7):643-51.
- Washburn RA, Smith KW, Jette AM, Janney CA. The Physical Activity Scale for the Elderly (PASE): development and evaluation. J Clin Epidemiol. 1993;46(2):153-62.
- 26. Schuit AJ, Schouten EG, Westerterp KR, Saris WH. Validity of the Physical Activity Scale for the Elderly (PASE): according to energy expenditure assessed by the doubly labeled water method. J Clin Epidemiol. 1997;50(5):541-6.
- Liu RD, Buffart LM, Kersten MJ, Spiering M, Brug J, van MW, et al. Psychometric properties of two physical activity questionnaires, the AQuAA and the PASE, in cancer patients. BMC Med Res Methodol. 2011;11:30.
- 28. Bjordal K, Ahlner-Elmqvist M, Tollesson E, Jensen AB, Razavi D, Maher EJ, et al. Development of a European Organization for Research and Treatment of Cancer (EORTC) questionnaire module to be used in quality of life assessments in head and neck cancer patients. EORTC Quality of Life Study Group. Acta Oncol. 1994;33(8):879-85.
- Fayers P, Bottomley A. Quality of life research within the EORTC-the EORTC QLQ-C30. European Organisation for Research and Treatment of Cancer. Eur J Cancer. 2002;38 Suppl 4:S125-S33.
- http://www.scp.nl/Onderzoek/Lopend_onderzoek/A_Z_alle_lopende_onderzoeken/ statusscores updated 2014. Available from: www.scp.nl.
- 31. Piccirillo JF, Tierney RM, Costas I, Grove L, Spitznagel EL, Jr. Prognostic importance of comorbidity in a hospital-based cancer registry. JAMA. 2004;291(20):2441-7.
- 32. Sammut L, Ward M, Patel N. Physical activity and quality of life in head and neck cancer survivors: a literature review. Int J Sports Med. 2014;35(9):794-9.
- Murphy MH, Donnelly P, Breslin G, Shibli S, Nevill AM. Does doing housework keep you health? The contribution of domestic physical activity to meeting current recommendations for health. BMC Public Health. 2013;13:966.
- 34. Dong L, Block G, Mandel S. Activities Contributing to Total Energy Expenditure in the United States: Results from the NHAPS Study. Int J Behav Nutr Phys Act. 2004;1(1):4.
- 35. Rogers LQ, Malone J, Rao K, Courneya KS, Fogleman A, Tippey A, et al. Exercise preferences among patients with head and neck cancer: prevalence and associations with quality of life, symptom severity, depression, and rural residence. Head Neck. 2009;31(8):994-1005.
- Bauman AE, Reis RS, Sallis JF, Wells JC, Loos RJ, Martin BW. Correlates of physical activity: why are some people physically active and others not? Lancet. 2012;380(9838):258-71.

- Buffart LM, Thong MS, Schep G, Chinapaw MJ, Brug J, LV vdP-F. Self-reported physical activity: its correlates and relationship with health-related quality of life in a large cohort of colorectal cancer survivors. PLoS One. 2012;7(5):e36164.
- Forbes CC, Blanchard CM, Mummery WK, Courneya KS. A comparison of physical activity correlates across breast, prostate and colorectal cancer survivors in Nova Scotia, Canada. Support Care Cancer. 2014;22(4):891-903.
- Klepin HD, Mohile SG, Mihalko S. Exercise for older cancer patients: feasible and helpful? Interdiscip Top Gerontol. 2013;38:146-57.
- 40. Gollhofer SM, Wiskemann J, Schmidt ME, Klassen O, Ulrich CM, Oelmann J, et al. Factors influencing participation in a randomized controlled resistance exercise intervention study in breast cancer patients during radiotherapy. BMC Cancer. 2015;15:186.
- Rogers LQ, Courneya KS, Robbins KT, Malone J, Seiz A, Koch L, et al. Physical activity correlates and barriers in head and neck cancer patients. Support Care Cancer. 2008;16(1):19-27.
- 42. Courneya KS, Friedenreich CM. Utility of the theory of planned behavior for understanding exercise during breast cancer treatment. Psychooncology. 1999;8(2):112-22.
- Lynch BM, Cerin E, Newman B, Owen N. Physical activity, activity change, and their correlates in a population-based sample of colorectal cancer survivors. Ann Behav Med. 2007;34(2):135-43.
- 44. Trinh L, Plotnikoff RC, Rhodes RE, North S, Courneya KS. Correlates of physical activity in a population-based sample of kidney cancer survivors: an application of the theory of planned behavior. Int J Behav Nutr Phys Act. 2012;9:96.
- 45. Sallis JF, Haskell WL, Wood PD, Fortmann SP, Rogers T, Blair SN, et al. Physical activity assessment methodology in the Five-City Project. Am J Epidemiol. 1985;121(1):91-106.



CHAPTER 3

A COMPREHENSIVE ASSESSMENT PROTOCOL INCLUDING PATIENT REPORTED OUTCOMES, PHYSICAL TESTS, AND BIOLOGICAL SAMPLING IN NEWLY DIAGNOSED PATIENTS WITH HEAD AND NECK CANCER: IS IT FEASIBLE?

A.J. van Nieuwenhuizen, L.M. Buffart, J.H. Smit, R.H. Brakenhoff, B.J.M. Braakhuis, R. de Bree, C.R. Leemans, I.M Verdonck-de Leeuw

Supportive care in cancer (2014) 09-25

ABSTRACT

Purpose: Large cohort studies are needed taking into account cancer-related, personal, biological, psychobehavioural, and lifestyle-related factors, to guide future research to improve treatment and supportive care. We aimed to evaluate the feasibility of a comprehensive baseline assessment of a cohort study evaluating the course of quality of life (QoL).

Methods: Newly diagnosed head and neck cancer (HNC) patients were asked to participate. Assessments consisted of questionnaires (635 items), a home visit (including a (psychiatric) interview, physical tests, blood and saliva collection), and tissue collection. Representativeness of the study sample was evaluated by comparing demographics, clinical factors, depression, anxiety, and QoL between responders and non-responders. Feasibility was evaluated covering the number of questions, time investment, intimacy and physical burden.

Results: During the inclusion period (four months), 15 out of 26 (60%) patients agreed to participate. Less women participated, 13% in responders group versus 63% in non-responders group (p=0.008). No other differences were found between responders and non-responders. Responders completed more than 95% of the questionnaires items, and rated the number of questions, time investment and intimacy as feasible, and the physical and psychological burden as low. It took on average 3 hours to complete the questionnaires and 1,5 hours for the home visit.

Conclusions: This study reveals that a comprehensive assessment including various questionnaires, physical measurements and biological assessments is feasible according to patients with newly diagnosed HNC. A large prospective cohort study has started aiming to include 739 HNC patients and their informal caregivers in the Netherlands.

INTRODUCTION

Worldwide, each year, head and neck cancer (HNC) accounts for more than 633,000 new cases and over 355,000 deaths.¹ In the Netherlands, the incidence of HNC increased from 2474 in 2001 to 2970 in 2011,² mainly due to aging, increased tobacco consumption by females in the 1980s and an increasing number of human papilloma virus related oropharyngeal carcinomas.³⁻⁵ Current five year survival rate of patients with advanced HNC in the Netherlands is approximately 60%.² For certain subsites, e.g. oropharyngeal carcinoma, survival is improving.⁶ Due to the increasing incidence and improved survival rates, more patients with HNC have to cope with various physical and psychosocial problems associated with the cancer diagnosis and treatment, such as decreased general and mental health, oral dysfunction, swallowing and speech problems and emotional distress, severely compromising health-related quality of life (HROoL).⁷⁻¹⁴ Compared to other types of cancer, including breast, colon and prostate cancer, HNC patients report high levels of distress.¹⁵ At the same time, an increasing number of studies suggest that HRQoL has prognostic value for survival.¹⁶⁻²² However, most previous studies on HRQoL and survival in HNC patients had some limitations related to relatively small sample sizes, the focus on specific sub-sites of HNC, adjustment for only a few confounders (e.g. lifestyle, demographic and clinical characteristics), or inclusion of only a few aspects of HRQoL.¹⁶⁻²⁵ Furthermore, little is known about the course of HRQoL of patients with HNC and its determinants across the cancer trajectory. Previous studies showed that several domains of HRQoL, including general health, mental health, physical function, appearance, employment and social functioning declined during and immediately after treatment, and improved after 6 months.7;8;12-14;23-26 Studies including long term follow-up showed that HRQoL stabilized one year after treatment and was not significantly different from baseline levels at 5-year follow up.^{7-9;25;26} However Mehanna et al.¹⁴ reported 10 years after diagnosis, significantly lower HRQoL scores than before treatment, which was recently confirmed by Oskam et al.27

A recent review among patients with HNC²⁸ showed that the majority of studies examining the course of HRQoL over time were limited by their retrospective study design, their focus on only a few aspects of HRQoL instead of all domains,²⁹ the use of Patient Reported Outcomes (PRO's) only, the small sample size, and lack of pre-treatment measurements of HRQoL.³⁰ In addition studies examining the association between HRQoL and survival lacked to adjust for all relevant confounders and different study designs were used.³¹ Therefore, there is need for a large multiinstitutional prospective cohort study evaluating the course of HRQoL in patients with HNC, and its relation with survival integrating all relevant cancer-related, personal,

103

genetic, biological, psychobehavioural, physical, lifestyle-related and social factors. Comprehensive insight in all these factors assessed in a standardized manner is necessary to unravel these complex associations. A study of this magnitude has never been carried out among patients with HNC, and it is unclear whether it is feasible to conduct such an extensive objective assessment of physical and cognitive function, lifestyle, a psychiatric interview, and collection of blood, saliva and tumor tissue, in addition to a large number of PRO's. Thus, the aim of the present study is to assess the feasibility of a comprehensive baseline assessment among patients with HNC.-

MATERIALS AND METHODS

Sample and Setting

As part of clinical routine in our institution, all newly diagnosed HNC patients are asked to fill out questionnaires on HRQoL and emotional distress during their first visit via OncoQuest³² OncoQuest is a touch screen computer system to monitor HRQoL in clinical practice and includes three questionnaires: the European Organization for Research and Treatment of Cancer, (EORTC) Quality of Life Questionnaire core module (QLQ-C30),^{34,35}) EORTC Head and Neck Module (HN35)³⁷ and Hospital Anxiety and Depression Scale (HADS).³⁶

The 30-item EORTC QLQ-C30 includes a global quality of life scale, 5 function scales regarding physical, role, emotional, cognitive and social functioning, three symptom scales (nausea and vomiting, fatigue and pain) and 6 single items related to dyspnea, insomnia, loss of appetite, constipation, diarrhea and financial difficulties.^{33:34} The 35-item EORTC QLQ-HN35 is a tumor-specific module addressing symptoms specifically associated with HNC, including pain, swallowing, senses, speech, social eating, social contact and sexuality, as well as 10 single items covering problems with teeth, dry mouth, sticky saliva, cough, opening the mouth wide, weight loss, weight gain, use of nutritional supplements, feeding tubes, and painkillers.³⁵ The HADS is a 14-item scale for measuring emotional distress and includes a total scale and an anxiety (HADS-A) and depression (HADS-D) subscale [36]. A total score of > 15 is used as an indicator of a high psychological distress. For the subscales, cut off points of \geq 8 are indicators of high levels of anxiety or depression.

From January to Mid-April 2012, every new patients with HNC was screened for eligibility for the current feasibility study. Patients presenting with oral, oropharygeal, hypopharygeal, laryngeal cancer and patients with neck metastasis of unknown

primary tumor with proven squamous cell histology, aged 18 years or older, treated with curative intent), who were able to write, read and speak Dutch were eligible. Patients suffering from severe psychiatric co-morbidities (e.g. schizophrenia, Korsakov's syndrome, severe dementia) were excluded. Eligible patients were asked to participate in this feasibility study by the treating surgeon, and subsequently, the research-physician provided more detailed information about the study and handed out written information. Non-participants were asked for their reason for not participating. All patients signed an informed consent statement prior to participation. The Institutional Review Board of the VU University Medical Center approved the study.

Procedures and assessments

After all eligible patients filled out the three questionnaires (EORTC QLQ C30, EORTC HN35, HADS) as clinical procedure, study participants filled out the comprehensive baseline assessment, consisting of 36 questionnaires and in total 762 items, which took place before the start of treatment. The questionnaires consisted variable sub domains, including general and disease specific QoL, cancer related, personal, psychobehavioural, physical, lifestyle-related, social factors and health care costs (Table 1). According to patient's preference, the PRO's were sent by postal mail, or a link was e-mailed to fill out the PRO's via internet. Subsequently, the research-physician visited the patients at their homes or in the hospital according to preference of the patients, to conduct a (psychiatric) interview, physical tests, and collection of blood. At the end of the visit, patients were instructed to collect 5 saliva samples during the same evening and next morning, and to wear an accelerometer (Actitrainer, Actigraph, Fort Walton Beach, Florida) for the next 7 days. Three to five days after the home visit, a telephone interview was conducted to assess diet using a 24-hour recall. Tumor tissue was gathered during the participants panendoscopic procedure, which is performed as a part of the diagnostic work-up. This procedure aims to determine the field of surgery and to investigate the presence of other tumours. During this panendoscopy a supplemental biopsy was taken for the current study besides the diagnostic tissue collection for the pathology department. An overview of all outcome measures included in the baseline assessment protocol is presented in Table 1. The assessment protocol was developed in collaboration with the coordinator of the Netherlands Study of Depression and Anxiety (NESDA) study.³⁷ NESDA is a large prospective cohort study which aims to describe the long-term course and consequences of depressive and anxiety disorders.

105

Fable 1. Overview of all outcome measures included in the assessment protocol: patient reported outcome.
(home) visit (physical tests and interviews), and biological and clinical factors.

OUTCOMES	Measurement instrument	Number of items
patient reported outcome		
Quality of life		
Generic	EuroQuol-5D (EQ-5D)	6
Disease specific	European Organisation for Research and Treatment of Cancer Quality of Life Questionnaire-core 30 (EORTC QLQ-C30)	30
Tumor specific	European Organisation for Research and Treatment of Can- cer Quality of Life Questionnaire module Head and Neck 35 (EORTC QLQ-HN35)	35
Speech	Speech Handicap Index (SHI)	31
Swallowing	Swallowing QoL Questionnaire (SwalQoL)	47
Shoulder	Shoulder Disability Questionnaire (SDQ)	16
Hearing	Caron questionnaire on hearing	19
Malnutrition	Short Nutritional Assessment Questionnaire (SNAQ)	4
Personal factors		
Personality	Neuroticism-Extroversion-Openness-Five factor inventory (NEO-FFI)	60
Locus of control	Pearlin & Schooler mastery scale (PSMS)	5
Coping	Mental Adjustment to Cancer scale (MAC)	39
Self efficacy	Generalized self efficacy scale (GES)	10
Well-being	Post traumatic Growth Inventory (PTGI)	21
Physical appearance	Body Image Scale (BIS)	10
Life events	Brugha Questionnaire	13
Coping	Utrechtse Coping List (UCL)	46
Psychobehavioural factors		
Severity anxiety / depression	Hospital Anxiety and Depression Scale (HADS)	14
Fatigue	Multidimensional Fatigue Inventory (MFI)	20
Fatigue	Fatigue Quality List (FQL)	1
Sleep Quality	Pittsburgh Sleep Quality Index (PSQI)	15
Fear of recurrence	Cancer Worry Scale (CWS)	8
Cognition	Cognitive Failure Questionnaire (CFQ)	25

Physical factors

Sexual function

106

Table 1. Continued

OUTCOMES	Measurement instrument	Number of items
Males	International index of Erectile Function (IIEF)	19
Females	Female Sexual Function Index (FSFI)	15
Females	Study specific questionnaire on fertility	16
Lifestyle-related factors		
Alcohol intake	Study specific questions	13
Alcohol dependence	Study specific questions	7
Drug use	Study specific questions	7
Smoking	Study specific questions	8
Nicotine dependence	Study specific questions	5
Physical activity	Physical Activity Scale for the Elderly (PASE)	31
Leisure	Study specific questions	15
Social factors		
Social support	Social Support List Interactions (SSL-I12)	12
Participation	Impact on Participation and Autonomy (IPA)	43
Health care use / costs		
Need and use care	Study specific questionnaire	28
Costs	Trimbos and iMTA questionnaire on Costs associated with Psychiatric illness (TiC-P): Adapted version for HNC	30
(Home) visit		
Personal factors		
Demographic	Standard questions	23
Socioeconomic status / literacy	Standard questions	27
Psychobehavioural factors		
Presence MDD	Composite international diagnostic interview (CIDI) – Ma- jor Depressive Disorder (MDD)	Variable
Anxiety disorder	Composite international diagnostic interview (CIDI) – Anx- iety disorders (GAD, SOC, PAN, AG)	Variable
Pain	Brief Pain Inventory (BPI)	13

Table 1. Continued

OUTCOMES	Measurement instrument	Number of items
Physical factors		
Speech quality	Speech recording (perceptual and objective analyses)	n.a.
Pulmonary function	Peak flow	n.a.
Strength: upper extremity	JAMAR handgrip dynamometer	n.a.
Strength: lower extremity	30s chair stand test	n.a.
Cardiorespiratory fitness	Chester Step test	n.a.
Body composition	Height, weight, body mass index, waist + hip circumference, thickness of 4 skin folds	n.a.
Blood pressure	Systolic and diastolic blood pressure	n.a.
Visual motor processing speed	Trail making Test part A	n.a.
Executive functioning	Trail making Test part B	n.a.
Activity monitoring	Accelerometer	n.a.
Food	24h recall	n.a.
Social factors		
Loneliness	De Jong-Gierveld Loneliness Scale	11
Health care use / costs		
Work productivity	Productivity and disease questionnaire (PRODISQ)	14
Biological and clinical		
factors		
Cancer-related factors		
Cancer location / stage	Standard questions, from medical record	n.a.
Treatment modality	Standard questions, from medical record	n.a.
Co-morbidity	Standard questions, from medical record	
Biological factors		
Tumor markers	Tumor tissue	n.a.
Biomarkers	Blood	n.a.
General laboratory	Blood	n.a.
DNA	Blood	n.a.
Proteomics	Blood	n.a.
Gene-expression (RNA)	Blood	n.a.

Abbreviations: MDD: major depressive disorder. N.a: not applicable
Outcome measures

The primary outcome of this study was the feasibility of the baseline assessment protocol, as evaluated by representativeness of the study population, achievability of baseline assessments, and accuracy of the protocol.

Representativeness was assessed by the following questions:

- What percentage of eligible patients is willing to participate?
- What are the main reasons for not participating?
- Are there differences between participants and non-participants regarding age, gender, diagnosis, comorbidity, stage, treatment, HRQoL and emotional distress?

Achievability was evaluated by the following items:

- A study specific questionnaire consisting of 4-point Likert scales (not feasible a bit feasible quit feasible very much feasible) covering the number of items, time investment, intimacy and burden of the PRO's and the home visit.
- The number of items successfully completed was registered as well as the time needed to complete the PRO's, as estimated by patients and the home visits as measured by a research physician.

Accuracy of the assessment protocol was evaluated using the following questions:

- Is the protocol clear to other assessors (i.e. two experienced fieldworkers and a coordinator from the NESDA study)?
- Is the assessment protocol complete, clear and accurate according to the research physician (A.N.)?

Statistical analysis

Statistical analyses were performed using Statistical Package for the Social Sciences (SPSS) 20.0 (SPSS Inc. Chicago, IL). Descriptive statistics (mean / standard deviation (SD) / proportions) were generated for demographic and clinical characteristics, emotional distress and HRQoL and questions on representativeness and achievability.

Differences in age, gender, diagnosis, comorbidity, stage, treatment, emotional distress, and HRQoL between participants and non-participants were tested with Mann-Whitney U test or χ^2 test. For the comparison of emotional distress and HRQoL we used results from the OncoQuest database.

Results

Representativeness

During a time period of 14 weeks 26 eligible patients were asked, of whom 15 (58%) were willing to participate (Figure 1). The main reason for not participating was the high burden of recently being diagnosed with cancer, leaving no room for additional inconvenience (n= 8; 30%).



Figure 1. Flowchart of all eligible patients and the reasons for non-participation.

However, all of these patients indicated that they would have been interested in participation at another point in time. Other reasons for non-participation were not willing to participate in any research project (n=2; 8%) or not willing to participate in this specific study protocol (n=1; 4%).

Except for gender, no differences in demographic and clinical characteristics, emotional distress and HRQoL were found between participants and non-participants. The proportion of women among non-participants (64%) was higher than among participants (13%), p = 0.008 (Table 2).

One participant (7%) and two non-participants (18%) did not fill out OncoQuest, and consequently their data on HRQoL and psychological distress were missing.

Characteristics	Participants (n= 15)	Non- participants (n= 11)	Difference (p-value)
Gender, n (%) male	13 (87)	4 (36)	0.008
Age, mean ± SD (range) years	63 ±12 (40-80)	62 ± 8 (52-78)	0.926
Tumor location, n (%)			0.986
Oral Cavity	4 (27)	3 (27)	-
Oropharynx	5 (33)	3 (27)	-
Hypopharynx	0 (0)	o (o)	-
Larynx	5 (33)	4 (36)	-
Unknown Primary	1 (7)	1 (9)	-
Disease Stage, n (%)			0.749
Ι	2 (13)	2 (18)	-
II	1 (7)	1 (9)	-
III	5 (33)	3 (27)	-
IV	7 (46)	5 (45)	-
Type of treatment, n (%)			0.486
CHRT	8 (53)	4 (36)	-
RT	4 (27)	2 (18)	-
SURG	1 (7)	3 (27)	-
TOE + SN	2 (13)	2 (18)	-
Comorbidity, n (%)			
None	4 (27)	3 (27)	0.683
Mild	3 (20)	3 (27)	-
Moderate	7 (47)	3 (27)	-
Severe	1 (7)	2 (18)	-
	Participant (n= 14)	Non-partici- pant (n= 9)	
HADS			
Total score, mean ± SD	10 ± 8	11 ± 5	0.587
Depression score ≥ 8 , n (%)	3 (21%)	1 (11%)	0.546
Anxiety score ≥ 8 , n (%)	3 (21%)	4 (44%)	0.262
Total score > 15, n (%)	3 (21%)	2 (22%)	0.966
EORTC QLQ-C30, mean ± SD			
Global quality of life	78 ± 15	64 ± 29	0.145
Physical function	87 ± 15	81 ± 23	0.480
Role function	88 ± 24	74 ± 30	0.229

Table 2. Differences in demographic and clinical characteristics, psychological distress and health-related quality of life between participants and non-participants.

Table 2. Continued

Characteristics	Participants (n= 15)	Non- participants (n= 11)	Difference (p-value)
Emotional function	60 ± 22	69 ± 20	0.296
Cognitive function	88 ± 21	94 ± 8	0.402
Social function	92 ± 18	85 ± 28	0.508

Abbreviations: SD: standard deviation; CHRT = chemo radiation therapy; EORTC QLQ C30 = European Organization for Research and Treatment of Cancer Quality of Life Questionnaire Core 30; HADS = Hospital Anxiety and Depression Scale; RT = radiotherapy; SURG = surgery; TOE + SN = transoral excision and sentinel node procedure;

Achievability

All participating patients filled out the questionnaires with 95-99% (n=8) to 100% (n=7) of the items completed. Two patients did not fill out the questionnaire on sexuality because they were not sexually active. No problems were detected with other questionnaires. The majority of the patients (80%) preferred the pen-and-paper version over the internet-based method. Completing the PRO's took on average 167 (range 100-270) minutes. All patients filled out the PRO's within one week. One patient needed 270 minutes to complete the PRO's due to concentration problems. Compared to the other patients, this was exceptionally long (range without this particular patient: 100-210 minutes).

Most patients (n=12) preferred the research physician to conduct the interview, physical tests, and blood collection during a home visit. Assessments of the other three patients were conducted during a hospital visit. The visits took on average 100 (range 60-145) minutes.

Physical Assessments	Percentage performed	Reasons for not performing physical assessment	
Grip strength	100	-	
30 second chair stand test	93	Amputated leg $(n = 1)$	
Step test	80	Amputated leg $(n = 1)$ Cardiac history $(n = 1)$ Severe immobility $(n = 1)$	
Accelerometer	53	Amputated leg $(n = 1)$ Severe immobility $(n = 1)$ PEG* tube placement $(n = 2)$ Lost accelerometer $(n = 2)$ Surgery, within a few days after visit $(n = 1)$	
Biological sample col- lection	Percentage per- formed	Reasons for not performing biological assessment	
Blood collection	87	Unwillingness (n = 2)	
Saliva collection	87	Not returned $(n = 2)$	
Tissue collection	53	Biopsy already taken (n = 1) No extra biopsies, direct surgery (n = 3) No extra biopsies, unknown primary tumor (n = 2) Reason not registered (n = 1)	

Table 3. Achievability of physical tests and biological sample collection.

Abbreviations: PEG (percutaneous endoscopic gastrostomy)

Some of the physical tests and biological sample collections could not be performed (Table 3). The Chester step test was not conducted in three patients due to physical impairments such as an amputated leg, cardiac history or severe mobility problems. Accelerometer data of six patients were missing due to: surgery within a few days after the baseline measurement (n=1), a percutaneous endoscopic gastrostomy (PEG) tube placement within a few days after baseline assessment (n=2), wheelchair dependence (n=1), crutches dependence in daily life (n=1) and losing the accelerometer (n=1)1). All other physical tests were completed by all patients. The cognitive test and the psychiatric interview were conducted in all patients without experiencing any problems. Dietary telephonic interview at the end of the assessments were taken in 11 (73%) patients. Four interviews were missed because treatment already started. Blood and saliva samples were collected in 13 patients (87%) patients; two patients refused to give blood samples, and two patients did not return their saliva samples. Tumor tissue from eight participants (53%) was collected during panendoscopy at the VU University Medical Center. Reasons for not collecting supplemental biopsies were: unknown primary tumor (n=2), no extra biopsies during panendoscopy because of immediate surgery (n= 3), biopsy already taken in outpatient clinic (n= 1) and unknown (n= 1). Patients evaluated the number of items, the time investment and personal or intimate character of the PRO's and the physical tests as feasible to very much feasible (Table 4). Regarding intimacy of questions, one patient found a questionnaire on sexuality too intimate and therefore rated the item intimacy as 'a bit feasible' for intimacy. One patient found the time investment of the home visit too long. Due to the presence of a child, this home visit took much longer compared to the other patients (145 min).

Questionnaires, n (%)	Not feasible	A bit feasible	Feasible	Very much feasible
Number of items	-	-	12 (86)	2 (14)
Time investment	-	-	6 (40)	8 (57)
Personal or intimate character of questions	-	1 (7)	7 (50)	6 (43)
Home visits, n (%)				
Number of questions	-	-	3 (21)	11 (79)
Time investment	1 (7)	-	4 (29)	9 (64)
Personal or intimate character of questions	-	-	2 (14)	12 (86)
Burden of physical assessments	-	-	2 (14)	12 (86)

Table 4. Feasibility of the questionnaires and home visit

Protocols

Generally, the research-physician reported the home visits to be very pleasant. Concentration problems were present in two patients according to the research-physician. Another patient was somewhat long-winded and expanding to various topics during the (psychiatric) interview, and had to be redirected to the questions **continuously**. No adverse events occurred during the visits. Furthermore, the protocols were independently judged as clear by the research physician, two other experienced fieldworkers and a research coordinator.

DISCUSSION

This pilot study showed that it is feasible to conduct a comprehensive baseline assessment compromising a large number of PRO's, interviews, physical tests, and biological sample collection among newly diagnosed HNC patients. We found the study sample to be a representative reflection of patients with HNC and the achievability of the assessment protocol was high.

Representativeness

The present study showed that 58% of newly diagnosed HNC patients were willing to participate in a comprehensive assessment. Our response rate was somewhat lower compared to 76%-97% reported in the other prospective cohort studies on HRQoL in HNC patients.7;9;12;38;39 Differences in response rate may be related to the large number of questionnaires included, since only one to three questionnaires were included in the previously mentioned studies. Taking into account our comprehensive assessments, we consider our response rate to be acceptable. Of all eligible patients 30% would like to participate in the study, but not at this specific moment. The most important argument for not participating was that the protocol seems too burdensome. For a future study, we can now better inform eligible patients on the time investments and feasibility of the protocol. The remaining 12% of eligible patients were not willing to participate in any study, and it is likely that this proportion will also be missed in a forthcoming cohort study.

In our study, females were found to be less likely to participate, which is in contrast to the pilot study of Hammerlid et al.40 who found all non-participants to be men. Another prospective study on long term HRQoL in patients with HNC found no significant differences between participants and non-participants in demographic and clinical characteristics.9 Therefore, our under-representativeness of women may be coincidental. On the other hand, it may be related to the higher, but non-significant anxiety levels we found among non-participants, since, in patients with HNC, women are more likely to report higher levels of anxiety than men.41 However, due to the small sample size, the non-significant differences in anxiety between participants and non-participants should be interpreted with caution. Studies among newly diagnosed cancer patients are at risk for selection bias underestimating problems related to emotional distress and HRQoL. Therefore, we will closely monitor differences in main characteristics between responders and non-responders in a future cohort study.

Achievability

Despite the considerable time investment (average of 270 minutes in total), our results showed that almost all patients found the study to be feasible to very much feasible. However, some patients experienced some problems with the questionnaires on sexuality, particularly those who were not sexually active. To avoid unnecessary confrontation with intimate questions in a future cohort study, we will therefore add a remark at the start of this questionnaire that patients can skip this questionnaire if not applicable. Regarding the (home) visits, we noticed that a quiet environment is important to prevent unnecessary delay in conducting the interviews and physical tests.

The home visits were almost fully completed. Blood and saliva was collected in 87% of the patients which we considered to be acceptable. A relatively large proportion (40%) of patients did not wear the accelerometer due to various reasons, of which some were largely unavoidable such as amputated leg and crutches dependence, whereas others may be prevented or (rapidly) solved. In this study, we did not give an accelerometer to patients undergoing a PEG tube placement which potentially hampers physical activities for a certain period of time. However it seems that the patients may be able to wear an accelerometer. Furthermore, a new device may quickly be sent to patients who lost their device. Tissue collection was successful in 53% of the participants. In most cases, supplemental tumor biopsies for this study were not taken due to logistical reasons including immediate surgical treatment of tumor, biopsies already taken in outpatient clinic, or because of an unknown primary tumor. In this pilot study, we did not collect tumor tissue samples from the surgical specimen from the pathology department, but we are planning to do so in a future study if tumor tissue is available.

Strengths and limitations

To our knowledge this pilot study is the first to evaluate a comprehensive assessment protocol of this extent in newly diagnosed HNC patients before treatment. In addition, the inclusion of objective physical tests and biological sample collection in addition to PRO's is a strength of this study. Another strength of this study was the ability to compare data on HRQoL and psychological distress between participants and non-participants, providing a thorough investigation of the representativeness of the included study sample. This study was limited by its focus on baseline assessment only, and we did not collect data on the feasibility of follow-up measurements. However, other longitudinal studies on HRQoL in patients with HNC reported acceptable dropout rates varying between 6 and 14%.^{9:38} Whether drop-out rates will be similar in the forthcoming cohort study remains unclear. Furthermore, while almost all single questionnaires were validated it is unclear whether assessment of multiple single

valid questionnaires impacts the validity. A previous study examining the influence of the structure of questionnaires on response outcomes showed that changes in order of questionnaires did not substantially affect the outcomes.⁴² In the present study, assessments were conducted in a hierarchical order, starting with the main outcome measure health-related quality of life, following by questionnaires assessing covariates. In conclusion, this study showed that it is feasible to conduct a comprehensive assessment protocol including PRO's, interviews, physical tests, and collection of biological samples in newly diagnosed HNC patients before the start of treatment. Therefore, we will set up a large multicentre cohort study in patients with HNC evaluating the course of HRQoL over time starting at diagnosis, and the relationship between HRQoL and survival, taking into account cancer-related, personal, genetic, biological, psychobehavioural, physical, lifestyle-related and social factors, the NETherlands QUality of life and BIomedical Cohort studies in cancer, Head and Neck (NET-QUBIC_HNC).

REFERENCES

- Ferlay J, Shin HR, Bray F, Forman D, Mathers C, Parkin DM. Estimates of worldwide burden of cancer in 2008: GLOBOCAN 2008. Int J Cancer 2010 Dec 15;127(12):2893-917.
- 2. http://www.cijfersoverkanker.nl/. 2013. Ref Type: Online Source
- 3. Braakhuis BJ, Visser O, Leemans CR. Oral and oropharyngeal cancer in The Netherlands between 1989 and 2006: Increasing incidence, but not in young adults. Oral Oncol 2009 Sep;45(9):e85-e89.
- 4. Rietbergen MM, Leemans CR, Bloemena E, Heideman DA, Braakhuis BJ, Hesselink AT, et al. Increasing prevalence rates of HPV attributable oropharyngeal squamous cell carcinomas in the Netherlands as assessed by a validated test algorithm. Int J Cancer 2012 Sep 5.
- 5. http://data.euro.who.int/hfadb. 2014. Ref Type: Online Source
- 6. Braakhuis BJM, Leemans CR, Visser O. Incidence and survival trends of head and neck squamous cell carcinoma in the Netherlands between 1989 and 2011. Oral Oncol 2014;accepted for publication.
- Abendstein H, Nordgren M, Boysen M, Jannert M, Silander E, Ahlner-Elmqvist M, et al. Quality of life and head and neck cancer: a 5 year prospective study. Laryngoscope 2005 Dec;115(12):2183-92.
- Bjordal K, Ahlner-Elmqvist M, Hammerlid E, Boysen M, Evensen JF, Biorklund A, et al. A prospective study of quality of life in head and neck cancer patients. Part II: Longitudinal data. Laryngoscope 2001 Aug;111(8):1440-52.
- 9. de Graeff A, de Leeuw JR, Ros WJ, Hordijk GJ, Blijham GH, Winnubst JA. Long-term quality of life of patients with head and neck cancer. Laryngoscope 2000 Jan;110(1):98-106.
- 10. Funk GF, Karnell LH, Christensen AJ. Long-term health-related quality of life in survivors of head and neck cancer. Arch Otolaryngol Head Neck Surg 2012 Feb;138(2):123-33.
- 11. Goldstein DP, Hynds KL, Christensen AJ, Funk GF. Health-related quality of life profiles based on survivorship status for head and neck cancer patients. Head Neck 2007 Mar;29(3):221-9.
- Hammerlid E, Bjordal K, Ahlner-Elmqvist M, Boysen M, Evensen JF, Biorklund A, et al. A prospective study of quality of life in head and neck cancer patients. Part I: at diagnosis. Laryngoscope 2001 Apr;111(4 Pt 1):669-80.
- Hammerlid E, Silander E, Hornestam L, Sullivan M. Health-related quality of life three years after diagnosis of head and neck cancer--a longitudinal study. Head Neck 2001 Feb;23(2):113-25.
- 14. Mehanna HM, Morton RP. Deterioration in quality-of-life of late (10-year) survivors of head and neck cancer. Clin Otolaryngol 2006 Jun;31(3):204-11.
- Carlson LE, Angen M, Cullum J, Goodey E, Koopmans J, Lamont L, et al. High levels of untreated distress and fatigue in cancer patients. Br J Cancer 2004 Jun 14;90(12):2297-304.
- Fang FM, Liu YT, Tang Y, Wang CJ, Ko SF. Quality of life as a survival predictor for patients with advanced head and neck carcinoma treated with radiotherapy. Cancer 2004 Jan 15;100(2):425-32.
- Karvonen-Gutierrez CA, Ronis DL, Fowler KE, Terrell JE, Gruber SB, Duffy SA. Quality of life scores predict survival among patients with head and neck cancer. J Clin Oncol 2008 Jun 1;26(16):2754-60.
- Mehanna HM, Morton RP. Does quality of life predict long-term survival in patients with head and neck cancer? Arch Otolaryngol Head Neck Surg 2006 Jan;132(1):27-31.
- Meyer F, Fortin A, Gelinas M, Nabid A, Brochet F, Tetu B, et al. Health-related quality of life as a survival predictor for patients with localized head and neck cancer treated with radiation therapy. J Clin Oncol 2009 Jun 20;27(18):2970-6.
- 20. Oskam IM, Verdonck-de L, I, Aaronson NK, Kuik DJ, de BR, Doornaert P, et al. Quality of life as predictor of survival: a prospective study on patients treated with combined surgery and radiotherapy for advanced oral and oropharyngeal cancer. Radiother Oncol 2010 Nov;97(2):258-62.

- Osthus AA, Aarstad AK, Olofsson J, Aarstad HJ. Head and neck specific Health Related Quality of Life scores predict subsequent survival in successfully treated head and neck cancer patients: a prospective cohort study. Oral Oncol 2011 Oct;47(10):974-9.
- 22. Urba S, Gatz J, Shen W, Hossain A, Winfree K, Koustenis A, et al. Quality of life scores as prognostic factors of overall survival in advanced head and neck cancer: analysis of a phase III randomized trial of pemetrexed plus cisplatin versus cisplatin monotherapy. Oral Oncol 2012 Aug;48(8):723-9.
- 23. Borghgraef K, Delaere P, Van den BW, Feenstra L. Quality of life in head and neck cancer patients. Acta Otorhinolaryngol Belg 1997;51(2):69-72.
- 24. Chen AM, Jennelle RL, Grady V, Tovar A, Bowen K, Simonin P, et al. Prospective study of psychosocial distress among patients undergoing radiotherapy for head and neck cancer. Int J Radiat Oncol Biol Phys 2009 Jan 1;73(1):187-93.
- 25. de Graeff A, de Leeuw Jr, Ros WJ, Hordijk GJ, Blijham GH, Winnubst JA. Long-term quality of life of patients with head and neck cancer. Laryngoscope 2000 Jan;110(1):98-106.
- Hammerlid E, Ahlner-Elmqvist M, Bjordal K, Biorklund A, Evensen J, Boysen M, et al. A prospective multicentre study in Sweden and Norway of mental distress and psychiatric morbidity in head and neck cancer patients. Br J Cancer 1999 May;80(5-6):766-74.
- 27. Oskam IM, Verdonck-de Leeuw IM, Aaronson NK, Witte BI, de BR, Doornaert P, et al. Prospective evaluation of health-related quality of life in long-term oral and oropharyngeal cancer survivors and the perceived need for supportive care. Oral Oncol 2013 May;49(5):443-8.
- Verdonck-de Leeuw IM, van Nieuwenhuizen AJ, Leemans CR. The value of quality-of-life questionnaires in head and neck cancer. Curr Opin Otolaryngol Head Neck Surg 2012 Apr;20(2):142-7.
- 29. Sayed SI, Elmiyeh B, Rhys-Evans P, Syrigos KN, Nutting CM, Harrington KJ, et al. Quality of life and outcomes research in head and neck cancer: a review of the state of the discipline and likely future directions. Cancer Treat Rev 2009 Aug;35(5):397-402.
- 30. Quinten C, Coens C, Mauer M, Comte S, Sprangers MA, Cleeland C, et al. Baseline quality of life as a prognostic indicator of survival: a meta-analysis of individual patient data from EORTC clinical trials. Lancet Oncol 2009 Sep;10(9):865-71.
- 31. Verdonck-de L, I, van NA, Leemans CR. The value of quality-of-life questionnaires in head and neck cancer. Curr Opin Otolaryngol Head Neck Surg 2012 Apr;20(2):142-7.
- 32. Verdonck-de Leeuw IM, de BR, Keizer AL, Houffelaar T, Cuijpers P, van der Linden MH, et al. Computerized prospective screening for high levels of emotional distress in head and neck cancer patients and referral rate to psychosocial care. Oral Oncol 2009 Oct;45(10):e129-e133.
- 33. Aaronson NK, Ahmedzai S, Bergman B, Bullinger M, Cull A, Duez NJ, et al. The European Organization for Research and Treatment of Cancer QLQ-C30: a quality-of-life instrument for use in international clinical trials in oncology. J Natl Cancer Inst 1993 Mar 3;85(5):365-76.
- Fayers P, Bottomley A. Quality of life research within the EORTC-the EORTC QLQ-C30. European Organisation for Research and Treatment of Cancer. Eur J Cancer 2002 Mar;38 Suppl 4:S125-S133.
- 35. Bjordal K, Ahlner-Elmqvist M, Tollesson E, Jensen AB, Razavi D, Maher EJ, et al. Development of a European Organization for Research and Treatment of Cancer (EORTC) questionnaire module to be used in quality of life assessments in head and neck cancer patients. EORTC Quality of Life Study Group. Acta Oncol 1994;33(8):879-85.
- Spinhoven P, Ormel J, Sloekers PP, Kempen GI, Speckens AE, Van Hemert AM. A validation study of the Hospital Anxiety and Depression Scale (HADS) in different groups of Dutch subjects. Psychol Med 1997 Mar;27(2):363-70.
- 37. Penninx BW, Beekman AT, Smit JH, Zitman FG, Nolen WA, Spinhoven P, et al. The Netherlands Study of Depression and Anxiety (NESDA): rationale, objectives and methods. Int J Methods Psychiatr Res

2008;17(3):121-40.

- 38. de Graeff A, de Leeuw JR, Ros WJ, Hordijk GJ, Blijham GH, Winnubst JA. Pretreatment factors predicting quality of life after treatment for head and neck cancer. Head Neck 2000 Jul;22(4):398-407.
- Osthus AA, Aarstad AK, Olofsson J, Aarstad HJ. Health-related quality of life scores in long-term head and neck cancer survivors predict subsequent survival: a prospective cohort study. Clin Otolaryngol 2011 Aug;36(4):361-8.
- 40. Hammerlid E, Bjordal K, Ahlner-Elmqvist M, Jannert M, Kaasa S, Sullivan M, et al. Prospective, longitudinal quality-of-life study of patients with head and neck cancer: a feasibility study including the EORTC QLQ-C30. Otolaryngol Head Neck Surg 1997 Jun;116(6 Pt 1):666-73.
- Linden W, Vodermaier A, Mackenzie R, Greig D. Anxiety and depression after cancer diagnosis: Prevalence rates by cancer type, gender, and age. J Affect Disord 2012 Dec 10;141(2-3):343-51.
- Dunn KM, Jordan K, Croft PR. Does questionnaire structure influence response in postal surveys? J Clin Epidemiol 2003 Jan;56(1):10-6.



- CHAPTER 7

GENERAL DISCUSSION AND FUTURE PERSPECTIVES

This thesis focused on health-related quality of life (HRQoL) in patients with head and neck cancer (HNC). The research in this thesis aimed to obtain more knowledge on HRQoL in relation to HNC survival, and on the role of physical activity and its relation to HRQoL among patients with HNC. In this final chapter, the most important findings of this thesis are summarized and put into perspective based on current literature. The chapter concludes with implications for clinical practice and recommendations for further research.

Main findings

The literature up to 2012 on HRQoL in patients with HNC was reviewed in Chapter 2. This chapter clearly showed the growing interest in using patient-reported outcome measures (PROMs) to measure HRQoL for research and clinical purposes. Besides, HRQoL seemed to be associated with survival. However, important gaps in knowledge were identified. First, as most of the knowledge on HRQoL in relation to survival was based on cross-sectional studies, little was known about the course of HRQoL over time, and about factors associated with (change in) HRQoL. Second, there is a lack of understanding of the association between HRQoL and survival and how this may be influenced by various cancer-related, personal, biological, and psychological, physical, lifestyle-related, and social factors. Third, data from PROMs are important in clinical trials as well as clinical care for patients. Knowledge on HRQoL may facilitate communication with patients, it can be used to screen and/or monitor physical and psychosocial problems,¹ and to identify patients with HNC with a need for supportive care in routine clinical practice.² Fourth, there is a need to develop supportive care interventions to improve HRQoL of patients with HNC and survivors and to evaluate the effectiveness and cost-effectiveness of these interventions.

Health-related quality of life in relation to survival

The association between HRQoL and survival in patients with HNC was further investigated in Chapters 3 and 4. Both studies showed that HRQoL was significantly **associated with survival after adjusting for relevant sociodemographic**, lifestyle-related and clinical variables. More specifically, there was strong evidence for a positive association between pre-treatment physical function and survival (Chapter 3 and 4) and between change in global QoL (from pre-treatment to 6 months post-treatment) and survival (Chapter 3). Chapter 4 also reported a significant positive association of other domains of HRQoL assessed pre-treatment (global QoL, role function, emotional function, cognitive function and social function) with survival. Moreover, Chapter 4 showed that HRQoL changes in the first 6 weeks of treatment were not associated with survival, while changes in physical and emotional function from diagnosis to 6

months following treatment were associated with survival, as well as changes in global QOL, and physical, emotional, and social function from pre-treatment to 12 months after treatment. Apparently, while HRQoL can be deteriorated shortly after treatment, most likely due to acute side effects of treatment, this short-term deterioration is not likely to be predictive for survival.³⁻⁵ Findings suggest that long-term deterioration of HRQoL, particularly deteriorations in physical and emotional function are likely to be predictive for survival. This predictive value of physical and emotional function has been acknowledged in previous studies.⁶⁻⁹

Deterioration in global QoL and social function were only associated with reduced survival one year after treatment. Perhaps, reduced physical and emotional function over time also affects global QoL and social function. On the other hand, patients with advanced illness may not be able to perform social activities. In conclusion, this thesis showed that poor HRQoL at time of diagnosis, as well as a worsening of global quality of life, physical, emotional and social function at 6 and 12 months follow-up compared to time of diagnosis is significantly associated with reduced survival in patients with HNC. Findings from this thesis highlight the importance of monitoring these HRQoL in the first year after diagnosis and treatment, because people with worsening HRQoL might have an increased risk of mortality.

Physical activity and HRQoL

Chapter 5 showed that HNC survivors (mean time after treatment, 21 months), and particularly older survivors, are at risk for low levels of physical activity. A large proportion of physical activity consisted of household activities (54%), followed by leisure time (34%) and occupational activities (12%). Total levels of physical activity in HNC survivors were substantially lower as compared to a non-cancer elderly population also using the PASE questionnaire, i.e. 145 vs. 100 points.¹⁰ The low levels of physical activity in patients with HNC was recently confirmed in a study that measured physical activity level of 229 min/day, as compared to 375 min /day) reported in healthy people who were slightly older.¹²

A second important finding of Chapter 5 was the positive association between physical activity and HRQoL. This finding confirm the results of previous smaller studies in HNC survivors.^{13,14} The causal direction of this association, however, is unknown due to the cross-sectional design of the study. It is therefore unclear if improving physical activity may improve the HRQL of HNC survivors. There is strong evidence among patients with cancer types other than HNC, such as breast cancer, that exercise interventions during and following cancer treatment can improve HRQoL.¹⁵ However, it is currently unclear whether this is also the case for HNC survivors. At least, the association between physical activity and HRQoL found in the present thesis provides rationale to conduct

such exercise trials in HNC survivors. In fact, pilot studies indicated that it is feasible to conduct exercise trials aiming to examine the efficacy of exercise interventions on HRQoL.^{16,17} Adherence to the exercise intervention is difficult and may be better when it is administered after treatment completion.¹⁸ In addition to a possible positive association between physical activity and HRQoL as such, observational studies showed that higher levels physical activity are also associated with a 38% lower cancerspecific mortality risk in patients with breast, colon and prostate cancer.¹⁹ One study showed comparable effects for a HNC population.²⁰ However, currently the number of studies examining the association between physical activity and survival in HNC is limited and more research is needed.²¹

Feasibility a large prospective cohort study examining HRQoL in patients with HNC To be able to investigate the course of HRQoL from time of diagnosis to long-term follow-up, and to unravel complex associations between the course of HRQoL and survival, in relation to cancer-related, personal, genetic, biological, psychological, physical, lifestyle-related and social factors, a prospective cohort study among a large group of patients with HNC with assessments of all these variables is warranted. Chapter 6 reveals that it is feasible to conduct a comprehensive assessment (with an average time investment of 270 minutes in total) protocol including patient reported outcomes (PROMs), interviews, physical tests, and collection of biological samples in patients that were newly diagnosed with HNC before the start of treatment. Importantly, the study sample appeared to be a representative reflection of patients with HNC and the achievability of the assessment protocol was high. As a result, a large multicenter cohort study in patients with HNC evaluating the course of HRQoL over time starting at diagnosis, and the relationship between HRQoL and survival, taking into account cancer-related, personal, genetic, biological, psychological, physical, lifestyle-related and social factors, the NETherlands QUality of life and BIomedical Cohort studies in cancer, Head and Neck (NET-QUBIC HNC) was launched. Currently baseline data and samples of 739 patients with HNC, as well as 3-months follow-up data (n=541) and 6 months (n=585) follow-up data and samples are available. Also, baseline of all 262 caregivers are available, as well as 3-months and 6 months follow-up data.²²

METHODOLOGICAL CONSIDERATIONS

The chapters in this thesis describe literature reviews and observational studies (crosssectional and longitudinal), to answer the research questions. Both types of studies have methodological shortcomings which are discussed below.

(Systematic) reviews

Reviews and systematic reviews play an essential role in evidence based medicine.²³ Reviews are a high quality source of cumulative evidence, especially when a methodological quality assessment and best evidence syntheses is included.²⁴ These methods were incorporated in the systematic review presented in Chapter 4. However, the majority of the included studies did not report hazard ratios and confidence intervals of non-significant associations. This hampered the conduct of a meta-analysis. In the best evidence synthesis we included only high quality studies to reduce possible bias. However, similar to other reviews and meta-analysis, publication bias endangers the external validity, and it cannot be ruled out fully.

Observational studies

Data from observational studies, both cross-sectional and longitudinal can be an important source of evidence about patients' true experience and HRQoL of cancer survivorship in addition to randomized controlled trials (RCT's).²⁵ Furthermore, compared to RCTs, a more heterogeneous group of patients can be included, and therefore findings may be more representative for the total patient population.²⁵ However, a disadvantage of observational studies is the inability to determine causality. Consequently, from the associations between HRQoL and survival reported in Chapter 3 and 4, it is unclear whether improving HRQoL, and particularly improving physical and emotional function from pre-treatment to 6 months post-treatment, would benefit survival or whether the deteriorations in these HRQoL domains are reflective of the disease severity. Comparably, from the positive association between physical activity would improve HRQoL or whether higher HRQoL would result in higher physical activity levels. Nevertheless, these associations emphasize the potential for future RCT's providing information on causality.

Assessment of physical activity

In this thesis, physical activity was assessed by self-report. PROMs are well suited to identify the dimensions and domains of physical activity behaviors.²⁶ However, they are prone to recall and social desirability bias,²⁷ and less accurate to estimate light

intensity physical activity.²⁶ This may have resulted in an over- or underestimation of physical activity levels. Nevertheless, the lower physical activity levels that were found in this thesis among patients with HNC as compared to previous findings in the general healthy population using the same questionnaire,¹⁰ has recently been confirmed in a study that assessed physical activity in patients with HNC using objective measurements (i.e. accelerometers).¹¹ Therefore, the finding from this thesis that patients with HNC have low levels of physical activity is most likely valid. Likewise, there may have been a slight over- or underestimation of the strength of the associations studied. However, because questionnaires are generally able to distinguish physically active from inactive patients, it is unlikely that this would have altered the main findings on the associations.²⁸

IMPLICATIONS FOR CLINICAL PRACTICE

Using HRQoL data in clinical practice helps the clinician towards a more holistic approach to the patient. Especially since patients with HNC and survivors often have complex rehabilitation needs due to the anatomical complexity of the head and neck region. Moreover it is important to improve outcomes of people living with and beyond cancer by moving to care pathways with a more patient centred approach opposite from only treating the disease. It is important to stress out choices between potentially morbid but life prolonging interventions versus high quality palliative care.²⁹ Because (change in) HRQoL is associated with HNC survival (Chapter 3 and 4), monitoring HRQoL in routine care may facilitate identifying patients who might benefit more from palliative care than from survivorship care.⁵ Improving communication on HRQoL between doctors and patients, and improving symptom management.¹ is essential, since current literature shows that 68% of patients with HNC have unmet supportive care needs.² Often reported concerns of patients with HNC include fear of cancer recurrence, future uncertainties, sadness, and concerns about family or friends.¹ Also, regarding HNC specific problems, even before treatment, patients rank issues related to verbal communication and eating above all other concerns.² These unmet needs are closely associated with HRQoL and physical and psychosocial well-being.30

PROMs are increasingly used in routine care. However, more uniformity is highly needed. This will provide more comparable results in cancer research and will improve the applicability in clinical practice. When considering implementing PROMs in routine care, important aspects are to limit the time needed to fill out the questionnaires,³¹ and reduce the complexity of the questionnaire.³² The way a questionnaire is completed is important, for instance before the consultation with the physician, at home or in the hospital.^{1,33} A tailored implementation strategy for successful integration of PROMs in

clinical practice is described in the EORTC manual.¹ Regarding relevant time points along the cancer trajectory, pre-treatment, 6 months and 1 year after treatment seems to be relevant, when PROMs are used to identify patients who might have an increased risk of mortality (Chapter 3 and 4). A recent study in our hospital showed that the use of PROMs in clinical practice is durable, even 5 years after the introduction.³³

New developments in cancer survivorship are self-monitoring, eHealth, and stepped care. For example, the fully automated self-management application Oncokompas supports cancer survivors to self-monitor HRQoL and symptoms, it provides personalized feedback, information, and a tailored overview of supportive care options. Oncokompas is shown to be effective to improve HRQoL, reduce the burden of symptoms and is cost-effective.³⁴ Another example is a guided self-help exercise program for head and neck specific symptoms, maintaining mobility of head and neck region as well as the swallowing and speech function.³⁵⁻³⁷ Furthermore, a stepped care program that supports HNC survivors to reduce psychological distress was found to be effective and cost-effective.^{38,39}

Based on the research in this thesis which showed that physical activity in HNC survivors is low (especially in older survivors) and associated with worse HRQoL, physical exercise might be offered to HNC survivors, via physical therapist-guided or home-based exercise programs.⁴⁰ When offering these exercise interventions, medical clearance may be indicated in some cases (e.g. in presence of cardiovascular, renal or metabolic symptoms) prior to high intensity exercise.⁴¹ Additionally, one should be aware of barriers specific to HNC survivors, such as poor health, malnutrition, and lack of interest⁴² Tailored advise on physical exercise is vital because patients have to maintain exercising over time. One should be aware of patients' barriers and preferences to increase exercise adherence and maintenance.¹⁸ This can be facilitated by modifiable determinants, such as motivational (e.g. self-efficacy, outcome expectations, action planning and control) and environmental (e.g., access to physical activity) factors.⁴³

RECOMMENDATIONS FOR FURTHER RESEARCH

The body of evidence to incorporate HRQoL in clinical practice is substantial, but translation into clinical practice is difficult.⁴⁴⁻⁴⁷ Likewise, implementation of tailored **supportive care programs remains a challenge**.⁴⁸ Organizations can increase the likelihood of successful routine outcome measurement by providing appropriate training, sufficient administrative support and adequate allocation of resources.⁴⁹ Also, as mentioned in Chapter 2 there is a wide variety of PROMs (among 93 studies more than 60 different questionnaires were applied in clinical care and research) and due to the lack of standardization it is difficult to use these data in clinical research and

practice.⁵⁰ Careful selection of PROMs is important to sustain comparability between findings in clinical practice and applicability towards clinical care, as well as research purposes (e.g. evaluation of treatment and supportive care). Working towards a uniform use of PROMs is necessary to better compare research findings and facilitate clinical. More research is needed to provide a more accessible, uniform platform which is convenient in routine clinical practice.

Manuals and training sessions that aim to support and engage clinicians in PROM based dialogue and patient-centered communication are starting to be developed.⁵¹ Still more research to upscale and implement the use of PROMs is necessary, such as if the use of PROMs is sustainable over time and whether patient-centered communications leads to a beneficial use of PROMs. Concerns from the organization (patient privacy, data security), management (cost-effectiveness) and clinicians (validity of the tools, when should a result be interpreted as clinically relevant) must be taken into account.⁵² All relevant stakeholders e.g. patients, nurses, physicians, administrational and technical staff) have to be included in the implementation process.⁵² To engage these stakeholders, participatory or qualitative research could be considered.

Health-related quality of life and survival

To better understand the factors that influence the course of HRQoL over time, and the association between HRQoL and survival in patients with HNC, more research is needed. Previous research showed considerable variation between patients who are at risk for poor HRQoL and others that are protected for poor HRQoL.⁵³ In addition, regarding survival, lower HRQoL pre-treatment and with a steeper decline in HRQoL during diagnosis and treatment is associated with shorter survival.⁵ In addition, Chapter 3 and 4 revealed that pre-treatment physical function and changes in physical and emotional function 6 months after treatment are most strongly associated with survival. It is therefore highly relevant to understand the risk factors and the protective resources that predict HRQoL. Unraveling potential determinants of HRQoL is important to identify treatable ones. The multicenter cohort study NET-QUBIC²² that followed the feasibility study presented in Chapter 6, aims to unravel the complex associations between cancer-related, personal, genetic, biological, psychological, physical, lifestyle-related and social factors, and survival in patients with HNC and, currently, studies on the NET-QUBIC cohort are being conducted.^{11,22,54,55}

Physical activity and HRQoL

The positive association between physical activity and HRQoL reported in Chapter 5 indicates the need to conduct a RCT to study potential causal effects of physical activity and exercise interventions on HRQoL in patients with HNC. Currently, RCT's on the

(cost-) effectiveness of exercise programs targeting HNC survivors are scarce and limited by a small sample size.^{16,17} Although exercise interventions have shown to be effective on HRQoL in patients with cancer, 15,56-60 these studies were primarily based on studies among patients with breast and prostate cancer. It is unknown whether these research findings can be generalized to patients with HNC due to distinct cancer trajectories, symptoms and side effects (e.g. shoulder dysfunction, dry mouth or throat, difficulty eating, and shortness of breath).⁶¹ Additionally, patients with HNC often have an unhealthy lifestyle. Smoking, alcohol use and malnutrition are highly represented in newly diagnosed patients with HNC.^{62,63} It can be hypothesized that this could negatively influence the motivation to be physically active. Identifying barriers of physical activity, developing strategies to overcome these barriers and to increase motivation to increase and maintain physical activity is needed. In addition, research needs to be done to learn about the optimal type and dosage of physical activity and the impact of sedentary behavior on HRQoL. Furthermore, evidence on the association between physical activity, and cancer incidence, and mortality is increasing, particularly in patients with breast, colon and prostate cancer.^{19,21,41} Yet, the causal effects of physical activity and exercise interventions on clinical outcome is unknown. Previous RCT's have shown that exercise interventions during chemotherapy can benefit chemotherapy completion rates,64,65 but results are not consistent,66,67 and studies were limited by a lack of statistical power.⁶⁶ Several pre-clinical studies found a direct beneficial effect on exercise on tumor growth.^{68,69} If these findings translate from mice to (wo)men is not known. Few RCTs are currently examining the effects of exercise interventions on survival in patients with colon cancer,⁷⁰ metastatic prostate cancer,71 ovarian cancer72 and allogeneic stem cell transplantation.73 Future studies should reveal whether exercise interventions can benefit chemoradiation efficacy and potentially survival. When future studies prove a beneficial effect of physical activity intervention among HNC survivors, these interventions can be incorporated in a HNC survivorship care plan along with other programs aiming to improve lifestyle changes (such as smoking and alcohol use).

CONCLUSION

HRQoL is an important outcome in HNC treatment. Two key gaps in knowledge were addressed in this thesis. It was still unclear which HRQoL domains are associated survival in HNC, at which time-point across the cancer journey, and whether absolute values and/or changes in HRQoL are associated with survival. Pre-treatment physical function and changes in physical and emotional function in the first year after treatment are significantly associated with survival. Secondly, more knowledge on HRQoL in HNC survivorship was needed and especially on the role of physical activity. HNC survivors are not physically active (especially older HNC survivors) in general, and this is associated with lower HRQoL. This thesis contributes to further advancing interdisciplinary research on HRQoL in HNC and may help to develop care pathways with a more patient centred approach.

REFERENCES

- 1. Wintner LM, Sztankay M, Aaronson N, et al: The use of EORTC measures in daily clinical practice-A synopsis of a newly developed manual. Eur J Cancer 68:73-81, 2016
- 2. Henry M, Habib LA, Morrison M, et al: Head and neck cancer patients want us to support them psychologically in the posttreatment period: Survey results. Palliat Support Care 12:481-93, 2014
- Egestad H, Emaus N: Changes in health related quality of life in women and men undergoing radiation treatment for head and neck cancer and the impact of smoking status in the radiation treatment period. Eur J Oncol Nurs 18:339-46, 2014
- 4. Klein J, Livergant J, Ringash J: Health related quality of life in head and neck cancer treated with radiation therapy with or without chemotherapy: a systematic review. Oral Oncol 50:254-62, 2014
- Rettig EM, D'Souza G, Thompson CB, et al: Health-related quality of life before and after head and neck squamous cell carcinoma: Analysis of the Surveillance, Epidemiology, and End Results-Medicare Health Outcomes Survey linkage. Cancer 122:1861-70, 2016
- 6. Jansen F, Verdonck-de Leeuw IM, Cuijpers P, et al: Depressive symptoms in relation to overall survival in people with head and neck cancer: A longitudinal cohort study. Psychooncology 27:2245-2256, 2018
- Montazeri A: Quality of life data as prognostic indicators of survival in cancer patients: an overview of the literature from 1982 to 2008. Health Qual. Life Outcomes 7:102, 2009
- 8. Quinten C, Coens C, Mauer M, et al: Baseline quality of life as a prognostic indicator of survival: a meta-analysis of individual patient data from EORTC clinical trials. Lancet Oncol 10:865-71, 2009
- 9. Quinten C, Martinelli F, Coens C, et al: A global analysis of multitrial data investigating quality of life and symptoms as prognostic factors for survival in different tumor sites. Cancer 120:302-311, 2013
- 10. Washburn RA, McAuley E, Katula J, et al: The physical activity scale for the elderly (PASE): evidence for validity. J. Clin. Epidemiol 52:643-651, 1999
- Douma JAJ, Verdonck-de Leeuw IM, Leemans CR, et al: Demographic, clinical and lifestyle-related correlates of accelerometer assessed physical activity and fitness in newly diagnosed patients with head and neck cancer. Acta Oncol 59:342-350, 2020
- Dohrn IM, Sjostrom M, Kwak L, et al: Accelerometer-measured sedentary time and physical activity-A 15 year follow-up of mortality in a Swedish population-based cohort. J Sci Med Sport 21:702-707, 2018
- 13. Rogers LQ, Courneya KS, Robbins KT, et al: Physical activity and quality of life in head and neck cancer survivors. Support. Care Cancer 14:1012-1019, 2006
- 14. Sammut L, Fraser LR, Ward MJ, et al: Participation in sport and physical activity in head and neck cancer survivors: associations with quality of life. Clin Otolaryngol 41:241-8, 2016
- 15. Sweegers MG, Altenburg TM, Chinapaw MJ, et al: Which exercise prescriptions improve quality of life and physical function in patients with cancer during and following treatment? A systematic review and meta-analysis of randomised controlled trials. Br J Sports Med 52:505-513, 2018
- Capozzi LC, Nishimura KC, McNeely ML, et al: The impact of physical activity on health-related fitness and quality of life for patients with head and neck cancer: a systematic review. Br J Sports Med 50:325-38, 2016
- Capozzi LC, Boldt KR, Lau H, et al: A clinic-supported group exercise program for head and neck cancer survivors: managing cancer and treatment side effects to improve quality of life. Support Care Cancer 23:1001-7, 2015
- Capozzi LC, McNeely ML, Lau HY, et al: Patient-reported outcomes, body composition, and nutrition status in patients with head and neck cancer: Results from an exploratory randomized controlled exercise trial. Cancer 122:1185-200, 2016

- Patel AV, Friedenreich CM, Moore SC, et al: American College of Sports Medicine Roundtable Report on Physical Activity, Sedentary Behavior, and Cancer Prevention and Control. Med Sci Sports Exerc 51:2391-2402, 2019
- 20. Duffy SA, Ronis DL, McLean S, et al: Pretreatment health behaviors predict survival among patients with head and neck squamous cell carcinoma. J Clin Oncol 27:1969-1975, 2009
- 21. McTiernan A, Friedenreich CM, Katzmarzyk PT, et al: Physical Activity in Cancer Prevention and Survival: A Systematic Review. Med Sci Sports Exerc 51:1252-1261, 2019
- 22. Verdonck-de Leeuw IM, Jansen F, Brakenhoff RH, et al: Advancing interdisciplinary research in head and neck cancer through a multicenter longitudinal prospective cohort study: the NETherlands QUality of life and BIomedical Cohort (NET-QUBIC) data warehouse and biobank. BMC Cancer 19:765, 2019
- Moher D, Shamseer L, Clarke M, et al: Preferred reporting items for systematic review and metaanalysis protocols (PRISMA-P) 2015 statement. Syst Rev 4:1, 2015
- lavin RE: Best evidence synthesis: an intelligent alternative to meta-analysis. J Clin Epidemiol 48:9-18, 1995
- 25. Trentino K, Farmer S, Gross I, et al: Observational studies should we simply ignore them in assessing transfusion outcomes? BMC Anesthesiol 16:96, 2016
- Strath SJ, Kaminsky LA, Ainsworth BE, et al: Guide to the assessment of physical activity: Clinical and research applications: a scientific statement from the American Heart Association. Circulation 128:2259-79, 2013
- 27. Sallis JF, Haskell WL, Wood PD, et al: Physical activity assessment methodology in the Five-City Project. Am. J. Epidemiol 121:91-106, 1985
- 28. Douma JAJ, de Beaufort MB, Kampshoff CS, et al: Physical activity in patients with cancer: self-report versus accelerometer assessments. Support Care Cancer, 2019
- 29. Haun MW, Estel S, Rucker G, et al: Early palliative care for adults with advanced cancer. Cochrane Database Syst Rev 6:Cdo11129, 2017
- 30. So WK, Choi KC, Chen JM, et al: Quality of life in head and neck cancer survivors at 1 year after treatment: the mediating role of unmet supportive care needs. Support Care Cancer 22:2917-26, 2014
- Lubberding S, van Uden-Kraan CF, Te Velde EA, et al: Improving access to supportive cancer care through an eHealth application: a qualitative needs assessment among cancer survivors. J Clin Nurs 24:1367-79, 2015
- Duman-Lubberding S, van Uden-Kraan CF, Peek N, et al: An eHealth Application in Head and Neck Cancer Survivorship Care: Health Care Professionals' Perspectives. J Med Internet Res 17:e235, 2015
- 33. Duman-Lubberding S, van Uden-Kraan CF, Jansen F, et al: Durable usage of patient-reported outcome measures in clinical practice to monitor health-related quality of life in head and neck cancer patients. Support Care Cancer 25:3775-3783, 2017
- 34. van der Hout A, van Uden-Kraan CF, Holtmaat K, et al: Role of eHealth application Oncokompas in supporting self-management of symptoms and health-related quality of life in cancer survivors: a randomised, controlled trial. Lancet Oncol 21:80-94, 2020
- 35. Cnossen IC, van Uden-Kraan CF, Witte BI, et al: Prophylactic exercises among head and neck cancer patients during and after swallowing sparing intensity modulated radiation: adherence and exercise performance levels of a 12-week guided home-based program. Eur Arch Otorhinolaryngol 274:1129-1138, 2017
- 36. Jansen F, Cnossen IC, Eerenstein SE, et al: Effectiveness and cost-utility of a guided self-help exercise program for patients treated with total laryngectomy: protocol of a multi-center randomized controlled trial. BMC Cancer 16:580, 2016

- Jansen F, Eerenstein SEJ, Cnossen IC, et al: Effectiveness of a guided self-help exercise program tailored to patients treated with total laryngectomy: Results of a multi-center randomized controlled trial. Oral Oncol 103:104586, 2020
- 38. Jansen F, Lissenberg-Witte BI, Krebber AMH, et al: Stepped care targeting psychological distress in head and neck cancer and lung cancer patients: which groups specifically benefit? Secondary analyses of a randomized controlled trial. Support Care Cancer 27:4543-4553, 2019
- 39. Krebber AM, Jansen F, Witte BI, et al: Stepped care targeting psychological distress in head and neck cancer and lung cancer patients: a randomized, controlled trial. Ann Oncol 27:1754-60, 2016
- 40. Schmitz KH, Campbell AM, Stuiver MM, et al: Exercise is medicine in oncology: Engaging clinicians to help patients move through cancer. CA Cancer J Clin 69:468-484, 2019
- Campbell KL, Winters-Stone KM, Wiskemann J, et al: Exercise Guidelines for Cancer Survivors: Consensus Statement from International Multidisciplinary Roundtable. Med Sci Sports Exerc 51:2375-2390, 2019
- 42. Gellert P, Witham MD, Crombie IK, et al: The role of perceived barriers and objectively measured physical activity in adults aged 65-100. Age Ageing 44:384-90, 2015
- 43. van Stralen MM, De Vries H, Mudde AN, et al: Determinants of initiation and maintenance of physical activity among older adults: a literature review. Health Psychology Review 3:147-207, 2009
- 44. Basch E, Deal AM, Kris MG, et al: Symptom Monitoring With Patient-Reported Outcomes During Routine Cancer Treatment: A Randomized Controlled Trial. J Clin Oncol 34:557-65, 2016
- Gilbert A, Sebag-Montefiore D, Davidson S, et al: Use of patient-reported outcomes to measure symptoms and health related quality of life in the clinic. Gynecol Oncol 136:429-39, 2015
- 46. Klinkhammer-Schalke M, Koller M, Steinger B, et al: Direct improvement of quality of life using a tailored quality of life diagnosis and therapy pathway: randomised trial in 200 women with breast cancer. Br J Cancer 106:826-38, 2012
- 47. Kotronoulas G, Kearney N, Maguire R, et al: What is the value of the routine use of patient-reported outcome measures toward improvement of patient outcomes, processes of care, and health service outcomes in cancer care? A systematic review of controlled trials. J Clin Oncol 32:1480-501, 2014
- Lagergren P, Schandl A, Aaronson NK, et al: Cancer survivorship: an integral part of Europe's research agenda. Mol Oncol 13:624-635, 2019
- 49. Duncan EA, Murray J: The barriers and facilitators to routine outcome measurement by allied health professionals in practice: a systematic review. BMC Health Serv Res 12:96, 2012
- Howell D, Molloy S, Wilkinson K, et al: Patient-reported outcomes in routine cancer clinical practice: a scoping review of use, impact on health outcomes, and implementation factors. Ann Oncol 26:1846-58, 2015
- Skovlund PC, Ravn S, Seibaek L, et al: The development of PROmunication: a training-tool for clinicians using patient-reported outcomes to promote patient-centred communication in clinical cancer settings. J Patient Rep Outcomes 4:10, 2020
- 52. Buttner M, Zebralla V, Dietz A, et al: Quality of Life Measurements: Any Value for Clinical Practice? Curr Treat Options Oncol 18:30, 2017
- 53. Verdonck-de Leeuw IM, Buffart LM, Heymans MW, et al: The course of health-related quality of life in head and neck cancer patients treated with chemoradiation: a prospective cohort study. Radiother. Oncol 110:422-428, 2014
- Mirosevic S, Thewes B, van Herpen C, et al: Prevalence and clinical and psychological correlates of high fear of cancer recurrence in patients newly diagnosed with head and neck cancer. Head Neck 41:3187-3200, 2019

- Piai V, Prins JB, Verdonck-de Leeuw IM, et al: Assessment of Neurocognitive Impairment and Speech Functioning Before Head and Neck Cancer Treatment. JAMA Otolaryngol Head Neck Surg 145:251-257, 2019
- Speck RM, Courneya KS, Masse LC, et al: An update of controlled physical activity trials in cancer survivors: a systematic review and meta-analysis. J. Cancer Surviv 4:87-100, 2010
- 57. Mishra SI, Scherer RW, Geigle PM, et al: Exercise interventions on health-related quality of life for cancer survivors. Cochrane Database Syst Rev:CD007566, 2012
- Ferrer RA, Huedo-Medina TB, Johnson BT, et al: Exercise interventions for cancer survivors: a metaanalysis of quality of life outcomes. Ann. Behav. Med 41:32-47, 2011
- Buffart LM, Sweegers MG, May AM, et al: Targeting Exercise Interventions to Patients With Cancer in Need: An Individual Patient Data Meta-Analysis. J Natl Cancer Inst 110:1190-1200, 2018
- 60. Sweegers MG, Altenburg TM, Brug J, et al: Effects and moderators of exercise on muscle strength, muscle function and aerobic fitness in patients with cancer: a meta-analysis of individual patient data. Br J Sports Med, 2018
- Nguyen NA, Ringash J: Head and Neck Cancer Survivorship Care: A Review of the Current Guidelines and Remaining Unmet Needs. Curr Treat Options Oncol 19:44, 2018
- 62. Beynon RA, Lang S, Schimansky S, et al: Tobacco smoking and alcohol drinking at diagnosis of head and neck cancer and all cause mortality: Results from head and neck 5000, a prospective observational cohort of people with head and neck cancer. Int J Cancer 143:1114-27, 2018
- 63. Orell-Kotikangas H, Osterlund P, Makitie O, et al: Cachexia at diagnosis is associated with poor survival in head and neck cancer patients. Acta Otolaryngol 137:778-785, 2017
- 64. Courneya KS, Segal RJ, Mackey JR, et al: Effects of aerobic and resistance exercise in breast cancer patients receiving adjuvant chemotherapy: a multicenter randomized controlled trial. J Clin Oncol 25:4396-404, 2007
- 65. van Waart H, Stuiver MM, van Harten WH, et al: Effect of Low-Intensity Physical Activity and Moderate- to High-Intensity Physical Exercise During Adjuvant Chemotherapy on Physical Fitness, Fatigue, and Chemotherapy Completion Rates: Results of the PACES Randomized Clinical Trial. J Clin Oncol 33:1918-27, 2015
- 66. Bland KA, Zadravec K, Landry T, et al: Impact of exercise on chemotherapy completion rate: A systematic review of the evidence and recommendations for future exercise oncology research. Crit Rev Oncol Hematol 136:79-85, 2019
- 67. Mijwel S, Bolam KA, Gerrevall J, et al: Effects of Exercise on Chemotherapy Completion and Hospitalization Rates: The OptiTrain Breast Cancer Trial. Oncologist 25:23-32, 2020
- Ashcraft KA, Warner AB, Jones LW, et al: Exercise as Adjunct Therapy in Cancer. Semin Radiat Oncol 29:16-24, 2019
- Pedersen L, Idorn M, Olofsson GH, et al: Voluntary Running Suppresses Tumor Growth through Epinephrine- and IL-6-Dependent NK Cell Mobilization and Redistribution. Cell Metab 23:554-62, 2016
- Courneya KS, Booth CM, Gill S, et al: The Colon Health and Life-Long Exercise Change trial: a randomized trial of the National Cancer Institute of Canada Clinical Trials Group. Curr Oncol 15:279-85, 2008
- Newton RU, Kenfield SA, Hart NH, et al: Intense Exercise for Survival among Men with Metastatic Castrate-Resistant Prostate Cancer (INTERVAL-GAP4): a multicentre, randomised, controlled phase III study protocol. BMJ Open 8:e022899, 2018

- 72. Hayes S, Friedlander M, Obermair A, et al: Exercise during chemotherapy for ovarian cancer (ECHO): study design features and outcomes of a cancer Australia and cancer council australia funded randomised, controlled trial. International journal of gynecological cancer 24:200-201, 2014
- 73. Wiskemann J, Kuehl R, Dreger P, et al: Physical Exercise Training versus Relaxation in Allogeneic stem cell transplantation (PETRA Study) Rationale and design of a randomized trial to evaluate a yearlong exercise intervention on overall survival and side-effects after allogeneic stem cell transplantation. BMC Cancer 15:619, 2015



SUMMARY

Chapter 1 provides background information on head and neck cancer (HNC), health related quality of life (HRQoL) and cancer survivorship, with a special focus on physical activity. HNC encompasses tumors in the upper respiratory or digestive tract. To date, HNC is the sixth leading type of cancer worldwide, with approximately 705.781 new cases in 2018 and a 5 year survival rate of approximately 65%. Treatment for patients with HNC often involves surgery, radiotherapy, chemotherapy or a combination of these modalities. As a consequence of HNC and its treatment, many patients face physical and psychosocial problems. Additionally, they may be confronted with very specific problems, such as oral dysfunction, swallowing and speech problems. This may have a distinct impact on the HRQoL of patients with HNC. Interestingly the course of HRQoL has shown to be worse in non-survivors of HNC compared to HNC survivors during the first 2 years after treatment. In addition, previous studies showed an significant association between HRQoL and survival, independently from other known demographic, lifestyle-related and clinical factors.

In cancer survivorship promoting a healthy lifestyle is critical. In patients with HNC awareness of lifestyle-related factors such as smoking, alcohol use and nutritional status seems to be eminent. However, little information is available on physical activity. Two key gaps in knowledge were addressed in this thesis. First, it is still unclear which HRQoL domains are associated with survival in HNC, at which time-point across the cancer journey, and whether absolute values and/or changes in HRQoL are associated with survival. Second, more knowledge on HRQoL in HNC survivorship is needed and especially on the role of physical activity and the association with HRQoL. Therefore, this thesis aims to investigate the association between HRQoL and survival in patients with HNC, and to investigate physical activity levels in long-term HNC survivors and the association with HRQoL.

Chapter 2 reviewed literature on HRQoL in patients with HNC. HRQoL seems an independent predictor of survival, but this association may be influenced by various cancer-related, personal, biological, psychobehavioural, physical, lifestyle-related, and social factors. Less is known about the course of HRQoL over time and about the same above mentioned possible factors associated with (change in) HRQoL in patients with HNC. Symptom management and psychosocial care may be beneficial for HNC patients to improve HRQoL, but more randomised controlled trials are needed. Studies on HRQoL in HNC are most often based on cross-sectional designs. The variability in outcome measures hampers the generalizability of the results of these studies. Information on HRQoL of caregivers is scarce. Better information on all aspects of the course of HRQoL from diagnosis and treatment to long-term survivorship or death is highly needed in both patients and their caregivers. More evidence on the efficacy of (new) treatment options, symptom management, and psychosocial care is needed, also

in the context of increasing long-term survival and the growing attention for cancer survivorship.

Chapter 3 specifically studied the association between HRQoL and survival in patients with HNC via a systematic review of prospective studies. A systematic search was conducted in four electronic bibliographic databases. We included studies published up to January 2014, providing data on HRQoL and survival, and the association between HRQoL and survival, among HNC patients. Two researchers independently performed a quality rating. A best evidence synthesis was applied to draw conclusions. In total, nineteen studies were included. Twelve studies included all subscales of a HRQoL questionnaire and seven studies focused on specific subscales. The mean quality score was $72 \pm 17\%$ and 58% of the studies were of high quality. According to the best evidence synthesis, there was strong evidence for a positive association between pre-treatment physical functioning and survival and between change in global QoL from pre-treatment to 6 months after treatment and survival. Due to inconsistent findings, there was insufficient evidence for an association with survival of other HRQoL domains, including role, emotional, cognitive and social functioning, mental health and well-being. This chapter highlights the need for high quality studies with a longitudinal design to examine the complex associations between HRQoL and survival.

Chapter 4 describes a cohort study in which the associations between HRQoL (assessed pre-treatment, post-treatment and change in HRQoL) and survival was examined in patients with HNC. We included different time points along the cancer trajectory (pretreatment, post-treatment and change in HRQoL). The study included 948 patients newly diagnosed with HNC (oral cavity, oropharynx, hypopharynx and larynx), treated with primary or adjuvant (chemo)radiotherapy with curative intent. HRQoL (EORTC QLQ-C30) was assessed pre-treatment and at 6 weeks, 6 months and 12 months after completion of treatment. Multivariable Cox regression analyses were performed to examine whether pre-treatment HRQoL, post-treatment HRQoL or change in HRQoL were associated with survival, after adjusting for demographic, clinical and lifestyle related variables. HRQoL was significantly associated with survival at all time-points, except for the subscale global QoL at 6 weeks post-treatment. A change in HRQoL in the first 6 weeks was not associated with survival. Changes in physical and emotional functioning from pre-treatment to 6 months post-treatment and changes in global QOL, and physical, emotional, and social functioning from pre-treatment to 12 months post-treatment were significantly associated with survival.

Chapter 5 describes the results of a cross-sectional study that aimed to assess patientreported levels of physical activity and its association with HRQoL adjusted for

relevant demographic, lifestyle-related and clinical factors, among 116 HNC survivors. Physical activity was assessed with the Physical-Activity-Scale-for-the-Elderly (PASE) and HRQoL with the EORTC-QLQ-C30 and EORTC-HN35. Associations were studied using univariable and multivariable regression analyses. Median PASE score was 100.3 (interquartile range 65.1;170.8) of which 54% were household, 34% leisure time and 12% occupational activities. Younger HNC survivors had higher levels of physical activity. Higher physical activity was significantly associated with higher global QoL. Findings for physical function, role function, social function, fatigue and pain were in line, but not statistically significant. These results indicates that improving physical activity might be an intervention target to improve HRQoL. However, due to the cross-sectional design, it is not possible to make causal inferences and it is unclear whether improving physical activity levels would improve HRQoL, or whether HNC survivors with lower HRQoL are less physically active.

To be able to unravel complex associations between cancer-related, personal, genetic, biological, psychobehavioural, physical, lifestyle-related and social factors, and survival in patients with HNC, a cohort study among a large group of patients with HNC with assessments of all these variables is warranted. Chapter 6 aimed to evaluate the feasibility of a comprehensive baseline assessment of a cohort study evaluating the course of HRQoL. Assessments consisted of questionnaires (635 items), a home visit (including a (psychiatric) interview, physical tests, blood and saliva collection), and tissue collection. Representativeness of the study sample was evaluated by comparing demographics, clinical factors, depression, anxiety, and quality of life between patients who participated in the study and those who did not. Feasibility was evaluated covering the number of questions, time investment, intimacy and physical burden. During the inclusion period of four months, 15 out of 26 (60%) patients agreed to participate. Less women participated, 13% in participant group versus 63% in non-participant group (p= 0.008). No other differences were found between participants and non-participants. Study participants completed more than 95% of the questionnaires items, and rated the number of questions, time investment and intimacy as feasible, and the physical and psychological burden as low. It took on average 3 hours to complete the questionnaires and 1,5 hours for the home visit. This study revealed that a comprehensive assessment including various questionnaires, physical measurements and biological assessments is feasible according to patients with newly diagnosed HNC. A large prospective cohort study has started aiming to include 739 HNC patients and their informal caregivers in the Netherlands.

In Chapter 7 the main findings of this thesis are discussed, as well as, the methodological considerations, implications for clinical practice and recommendations for further

research. It is concluded that HRQoL is an important outcome in HNC treatment. Two key gaps in knowledge were addressed in this thesis. It was still unclear which HRQoL domains are associated with survival in HNC, at which time-point across the cancer journey, and whether absolute values and/or changes in HRQoL are associated with survival. Pre-treatment physical function and changes in physical and emotional function in the first year after treatment are significantly associated with survival. Secondly, more knowledge on HRQoL in HNC survivorship was needed and especially the role of physical activity. HNC survivors are not physically active (especially older HNC survivors) which is associated with lower HRQoL. This thesis contributes to further advancing interdisciplinary research on HRQoL in HNC and may help to develop care pathways with a more patient centred approach.


SUMMARY IN DUTCH

Hoofdstuk 1 geeft achtergrondinformatie over hoofd-hals kanker en kwaliteit van leven (KvL) voor, tijdens en na kanker, met een specifieke focus op lichamelijke activiteit. Hoofd-hals kanker bevat tumoren van de bovenste lucht- of voedselweg. Wereldwijd behoort hoofd-hals kanker tot de 6e meest voorkomende vorm van kanker met meer dan 700.000 nieuwe patiënten in 2018. Het overlevingspercentage na 5 jaar is rond de 65%. De behandeling van hoofd-hals kanker omvat in de regel, chirurgie, radiotherapie, chemotherapie of een combinatie van deze behandelmodaliteiten. Als gevolg van de diagnose en behandeling van hoofd-hals kanker worden patiënten geconfronteerd met de fysieke en psychosociale gevolgen ervan. Meer patiënten zullen hiermee om moeten gaan, gezien te toenemende incidentie en overlevingskansen. De algehele en mentale gezondheid verminderen als gevolg van de fysieke en psychosociale problemen die patiënten met hoofd-hals kanker ervaren. Daarbij worden de patiënten met specifieke klachten geconfronteerd, zoals orale dysfunctie, slik- en stemproblemen. Dit heeft een aanzienlijk effect op de KvL voor deze patiëntengroep. Tevens is het beloop van KvL slechter voor patiënten die binnen de eerste 2 jaar na behandeling overlijden. Daarnaast heeft eerder onderzoek aangewezen dat er een significante associatie is tussen KvL en overleven, onafhankelijk van bekende demografische, leefstijl-gerelateerde en klinische factoren.

Het bevorderen van een gezonde leefstijl bij patiënten die leven met of na de diagnose kanker is van groot belang. Hierbij komen adviezen rondom roken, alcohol gebruik en voeding aan bod. Echter, het lijkt erop dat het advies om lichamelijk actief te blijven relatief minder bekend is bij zorgverleners. Daarnaast is het zo dat de kennis rondom lichamelijke activiteit bij patiënten met hoofd-hals kanker relatief achter blijft, met name in vergelijking met de aanwezige kennis rondom hoofd-hals kanker specifieke symptomen van KvL.

Er zijn nog enkele belangrijke vraagstukken onbeantwoord gebleven. Twee specifieke kennishiaten komen in dit proefschrift aan bod. Zo is het nog onvoldoende duidelijk welke domeinen van KvL specifiek samenhangen met overleving, welke momenten in het ziektebeloop het meest van belang zijn en of absolute waarden of juist verandering in KvL geassocieerd zijn met overleven. Tevens is er meer kennis nodig rond KvL bij patiënten die de behandeling van kanker hebben afgerond. De rol van lichamelijke activiteit in de nazorg van patiënten met hoofd-hals kanker is onbekend. Om deze redenen heeft dit proefschrift als doel de associatie tussen KvL en overleving nader te onderzoeken bij patiënten met hoofd-hals kanker en om lichamelijke activiteit en de relatie met KvL te onderzoeken bij patiënten die leven na hoofd-hals kanker.

Hoofdstuk 2 beschrijft een literatuuronderzoek over KvL bij patiënten met hoofdhals kanker. In dit onderzoek komt naar voren dat KvL een belangrijk onderdeel is van klinische studies. Daarnaast lijkt KvL een onafhankelijke voorspeller te zijn voor

overleven, maar dit verband wordt beïnvloed door verscheidene kanker gerelateerde, persoonlijke, biologische, fysieke, leefstijl-gerelateerde en sociale factoren. We weten nog weinig over het beloop van KvL, met name wanneer er rekening wordt gehouden met de eerder genoemde factoren. Het managen van (hoofd-hals specifieke) symptomen en de psychosociale zorg hebben mogelijk een positief effect op KvL bij patiënten met hoofd-hals kanker. Echter om dit te kunnen bewijzen zijn meer gerandomiseerde klinische trials nodig. De studies die tot nu toe zijn gedaan naar KvL bij hoofd-hals kanker bevatten vaak beperkingen. Zo hadden ze meestal een cross-sectioneel ontwerp. Daarnaast was er een grote variatie in de meetinstrumenten waarmee KvL gemeten was, waardoor resultaten lastig met elkaar kunnen worden vergeleken. Bovendien is kennis van KvL bij naasten beperkt. Beter inzicht in het beloop van KvL over de tijd vanaf het moment van diagnose bij zowel patiënten als hun naasten is nodig. Ook is meer kennis nodig omtrent symptoombestrijding en psychosociale zorg, met name in de context van toenemende overlevingskansen en in het kader van de toegenomen **aandacht omtrent het leven met en na kanker**.

Hoofdstuk 3 beschrijft een systematisch literatuuronderzoek naar de relatie tussen KvL en overleven. Een systematische zoekopdracht werd uitgevoerd in vier elektronische bibliografische databases. Prospectieve observationele studies gepubliceerd tot januari 2014, met gegevens over KvL, overleven en de relatie tussen KvL en overleven bij patiënten met hoofd-hals kanker werden geïncludeerd. De kwaliteit van de studies werd door twee onderzoekers onafhankelijk van elkaar beoordeeld. Daarnaast werd een 'best evidence synthesis' toegepast om conclusies te kunnen trekken. In totaal werden 19 studies geïncludeerd. Twaalf studies includeerde alle domeinen van KvL en 7 studies beschreven alleen specifieke domeinen. De gemiddelde beoordeling van de kwaliteit was 72 ±17% (schaal 0-100) en 58% van de studies waren van hoge kwaliteit. Er werd een sterk bewijs gevonden voor een positieve associatie tussen fysiek functioneren voorafgaand aan de behandeling en overleving. Tevens vonden we sterk bewijs voor een positieve associatie tussen verandering in de algemene KvL, van voor de behandeling tot 6 maanden na de behandeling, en overleving. Door inconsistente bevindingen was er onvoldoende bewijs voor een associatie tussen andere KvL domeinen (zoals rol functioneren, emotioneel functioneren, cognitief functioneren, sociaal functioneren en mentaal welbevinden) en overleving. In de toekomst zijn kwalitatief goede longitudinale studies nodig om de complexe associaties tussen KvL en overleving verder te ontrafelen.

In hoofdstuk 4 werd de associatie tussen KvL en overleving onderzocht in een prospectieve cohort van patiënten met hoofd-hals kanker. Van januari 1999 tot oktober 2009 zijn 948 nieuw gediagnosticeerde patiënten met hoofd-hals kanker (mondholte,

orofarynx, hypofarynx en larynx) geïncludeerd, die in opzet curatief behandeld werden middels primaire of adjuvante (chemo)radiotherapie. Tevens hadden deze patiënten de vragenlijst EORTC-QLQ-C30, voorafgaand aan de behandeling, 6 weken, 6 maanden en een jaar na afloop van de behandeling ingevuld om inzicht te krijgen in de KvL. Middels multivariabele Cox regressie analyse werd het verband tussen KvL voor de behandeling, KvL na de behandeling of verandering in KvL enerzijds en overleving anderzijds onderzocht waarin werd gecorrigeerd voor relevante demografische, leefstijl-gerelateerde en klinische variabelen. Hier kwam uit naar voren dat KvL op elk gemeten moment significant geassocieerd was met overleven, met uitzondering van de subschaal algemene KvL gemeten op 6 weken na afronden van de behandeling. Verandering in KvL gemeten van 6 weken na de behandeling ten opzichte van voor de behandeling was niet voorspellend voor overleving. Verandering in fysiek functioneren en emotioneel functioneren van voor de behandeling tot 6 maanden na de behandeling was wel significant geassocieerd met overleving. Evenals verandering in algemene KvL, fysiek functioneren, emotioneel functioneren en sociaal functioneren van voor de behandeling tot 1 jaar na de behandeling.

Hoofdstuk 5 beschrijft een cross-sectionele studie waarin het lichamelijke activiteiten niveau en de relatie met KvL werd onderzocht bij 116 patiënten die zijn behandeld voor hoofd-hals kanker. Lichamelijke activiteit werd gemeten met de 'Physical-Activity-Scale-for-the-Elderly (PASE) vragenlijst en KvL met de EORTC-QLQ-C30 en EORTC-HN35. Associaties werden bestudeerd middels univariabele en multivariabele regressie analyse. De mediane PASE score was 100.3 (interkwartielafstand 65.1;170.8) waarvan 54% bestond uit huishoudelijke activiteiten, 34% activiteiten in vrije tijd en 12% werk gerelateerde activiteiten. Jongere patiënten waren significant actiever dan oudere patiënten. Patiënten die lichamelijk actiever waren hadden een significant hogere algemene KvL. De verbanden van fysiek functioneren, rol functioneren, sociaal functioneren, vermoeidheid en pijn waren in dezelfde richting maar niet statistisch significant.

Om de complexe associaties tussen kanker gerelateerde, persoonlijke, genetische, biologische, psychosociale, fysieke, leefstijl-gerelateerde en sociale factoren met overleving te ontrafelen bij nieuw gediagnosticeerde patiënten met hoofd-hals kanker is een prospectieve omvangrijke cohort studie noodzakelijk. Hoofdstuk 6 geeft de resultaten weer van de haalbaarheid van een dergelijke omvangrijke meting waarin al deze factoren worden onderzocht. Hierbij werden nieuw gediagnosticeerde patiënten met hoofd-hals kanker gevraagd deel te nemen. De basismeting bestond uit vragenlijsten (635 items), een thuismeting (interview, fysieke testen, verzameling van bloed en speeksel) en verzameling van tumorweefsel. Representativiteit werd

bepaald door vergelijking van sociaal-demografische en klinische gegevens en scores op angst, depressie en KvL tussen deelnemers en niet-deelnemers met een Mann-Whitney U test of een Chi-kwadraat toets. Na afloop werd de haalbaarheid van de basismeting geëvalueerd middels een korte studie specifieke vragenlijst. Gedurende de inclusieperiode (4 maanden) participeerden 15 van de 26 (60%) patiënten. Vrouwen participeerden minder vaak, 13% in de deelnemende groep tegenover 63% in de niet deelnemende groep (p= 0.008). Andere patiëntkenmerken verschilden niet tussen de deelnemers en niet-deelnemers. Meer dan 95% van de items in de vragenlijsten werd ingevuld. Het aantal vragen, de tijdsinvestering en de intimiteit van de basismeting waren volgens de deelnemers haalbaar en de fysieke en psychische belasting werd beoordeeld als gering.

De belangrijkste bevindingen van dit proefschrift worden bediscussieerd in Hoofdstuk 7, waarin ook ingegaan wordt op de methodologische overwegingen, de implicaties voor de klinische praktijken de aanbevelingen voor toekomstig onderzoek. Er wordt geconcludeerd dat KvL een belangrijke uitkomstmaat is in de zorg voor patiënten met hoofd-hals kanker. Twee belangrijke kennishiaten zijn in dit proefschrift aan de orde gekomen. Ten eerste was het onvoldoende duidelijk welke KvL domeinen geassocieerd zijn met overleving, welke momenten in het ziektebeloop het meest van belang zijn en of absolute waarden of juist verandering in KvL geassocieerd zijn met overleving. Resultaten van de studies in dit proefschrift laten zien dat fysiek functioneren voorafgaand aan de behandeling, alsook de veranderingen in fysiek functioneren en emotioneel functioneren van voor de behandeling tot 6 maanden na de behandeling sterk geassocieerd zijn met overleving. Ten tweede was de kennis omtrent KvL van lange termijn overlevenden van hoofd-hals kanker en de rol die lichamelijke activiteit daarin heeft onbekend. Patiënten die de behandeling van hoofd-hals kanker hebben afgerond zijn lichamelijk inactief (met name de ouderen), wat geassocieerd is met een lagere KvL. De studies in proefschrift dragen bij aan bevorderen van interdisciplinair onderzoek omtrent KvL bij hoofd-hals kanker. De kennis omtrent patiënten met hoofdhals kanker die leven met en na de diagnose kanker zal bijdragen aan multidisciplinaire zorg waarbij de patiënt centraal staat.



ADDENDUM

DANKWOORD

Eindelijk is het zover, het proefschrift is afgerond. Een lange weg bereikt het eindpunt. Veel is er gebeurt en verandert sinds de aanvang van mijn tijd als onderzoeker in 2011. Nu beginnen met het einde. Het gedeelte wat vanaf de start al in mijn hoofd heeft gezeten en wat gaande weg steeds meer is uitgebreid. Het dankwoord.

Allereerst wil ik de patiënten bedanken voor hun deelname aan de studies beschreven in dit proefschrift. Zonder hen was het niet tot stand gekomen, de inzet en bereidheid om, geregeld uitgebreide, vragenlijsten in te vullen hebben mijn bewondering, zeker gezien de moeilijke tijden van het ziekteproces. Met name de huisbezoeken zal ik niet snel vergeten.

Daarnaast wil ik een aantal mensen in het bijzonder bedanken:

Prof. Dr. I.M. Verdonck-de Leeuw, beste Irma, bedankt voor het vertrouwen in mij als persoon en onderzoeker. Bedankt dat je mij hebt benaderd voor het opstarten van NET-QUBIC, een groot project waar ik veel van heb geleerd. Bedankt dat je me de vrijheid hebt gegeven om mijn promotie in mijn eigen tempo af te ronden, en de hulp die je hebt geboden wanneer deze nodig was.

Prof. Dr. C.R. Leemans, beste René, graag wil ik je bedanken voor de mogelijkheid om op de afdeling KNO / hoofd-hals mijn promotieonderzoek en opleiding te volbrengen. Bedankt voor je kritische blik op de stukken, het proefschrift en het vertrouwen in mij als persoon, onderzoeker en arts.

Dr. L.M. Buffart, beste Laurien, wat een werk! En wat ben ik je dankbaar voor jouw input, motivatie en visie. Ik heb grote bewondering voor jouw enthousiasme voor statistiek en je bevlogenheid om mij hier ook in mee te nemen. Jouw energie en optimisme werken aanstekelijk.

De leden van de leescommissie, Prof. Dr. B.J. Slotman, Prof. Dr. R.J. Baatenburg de Jong, Prof. Dr. C.H.J. Terhaard, Dr. M.M. Stuiver, Dr. M. Vergeer wil ik allen bedanken voor de bereidheid om dit manuscript te beoordelen en voor het plaatsnemen in de **promotiecommissie**.

Graag wil ik ook alle personen uit mijn onderzoekerstijd bedanken vanuit de vleugel 2Y. Een combinatie van kwaliteit van leven en tumorbiologie was snel gemaakt. Ruud en Boudewijn, bedankt voor het meenemen van mij in jullie team. Samen met alle

andere onderzoekers Michelle, Charlotte, Sarah, bedankt voor de gezelligheid, lunches, leer over de natuur en relativering. Derrek, bedankt voor alle motivatie, gezelligheid, koffie en pragmatiek in onderzoek en opleiding, door jou lijkt alles een eitje.

Ton Houffelaar, bedankt voor al je hulp rondom de datamanagement van mijn onderzoek. Zonder jouw hulp was het niet gelukt. Bedankt voor de goede sfeer, je interesse en je luisterend oor. Ook Jacqeline Geskus wil ik hiervoor bedanken. Met veel plezier zat en zit ik in jullie kamer.

Het lijkt een klein mensenleven geleden en volgens mij ben ik de laatste die gaat promoveren, maar uiteraard wil ik ook al mijn kamergenoten bedanken: Anne-Marie Krebber, Sanne Duman-Lubberding, Femke Jansen, Ingrid Cnossen, bedankt voor jullie raad en daad, jullie theater en plezier, jullie inspiratie en ventilatie. Met jullie is onderzoek doen regelmatig een feestje geweest. Sandra Biemans, de spil in het wiel van onze onderzoeksgroep, bedankt voor al je persoonlijke steun en de enorme inzet die je voor onze onderzoeksgroep hebt.

De staf van de afdeling KNO van het Amsterdam UMC, locatie VUmc, van het Tergooi ziekenhuis en het Spaarne Gasthuis, bedankt voor de opleidingsmomenten, de KPB's waar nooit een einde aan komt, jullie geduld en vertrouwen. Daarnaast de dames van het secretariaat waar ooit mijn KNO carrière begon, ik voel me nog altijd thuis bij jullie. Alle medewerkers van 1C, de polikliniek, de OK en het audiologisch centrum. Inmiddels ben ik overal kind aan huis, bedankt voor dat gevoel.

Mijn (oud) AIOS collega's, bedankt voor het maken van mijn opleidingstijd. Ik heb me altijd onderdeel van een hecht team gevoeld. Velen weekenden, wintersporten en feestjes zal ik niet meer vergeten. Om de laatste fase van de opleiding af te ronden met de moeders van de KNO is natuurlijk helemaal mooi. Sharon en Fedja, heerlijk om met jullie over hele andere dingen te kunnen praten dan het werk. Een al feest der herkenning en relativering.

Mijn paranimfen

Lieve Maarten, sopje of was het nou sipje? Wat een eer dat je naast me staat. Zoals altijd niet zonder slag of stoot. Wat ben je een prettig mens om bij aanwezig te zijn. Ik heb zoveel met je gelachen en ook geleerd van hoe jij in het leven staat. Je bent een geweldige dokter, collega, vriend en Franse buurman.

Lieve Saskia, wat ben ik trots dat jij naast me staat. Degene die vanaf het eerste moment mijn artikelen daadwerkelijk heeft gelezen. Mijn grote zus, wat ben ik ongelofelijk blij dat jij in mijn leven bent. Lief en leed hebben we gedeeld en ik kan altijd bij je terecht.

Lieve papa, veel heb ik aan je te danken, een warme jeugd waarbij het leven om de kinderen draaide. Een stimulerende opvoeding om uit het leven te halen wat erin zit. Alle vertrouwen bij elke stap die ik in mijn leven zette. De juiste raad wanneer ik deze nodig had. Grote bewondering heb ik voor je hoe jij in het leven staat en je optimisme, met name in moeilijke tijden toen je er alleen voor kwam te staan. Lieve mama, er gaat geen dag voorbij zonder dat ik aan je denk of je mis. Vaak vraag ik me af hoe jij dingen zou aanpakken en vraag ik in mijn hoofd je mening. Wat zou je genoten hebben van je kleinkinderen. Ik doe er alles aan om jouw gedachtegoed in leven te houden. Ik draag dan ook dit proefschrift op aan jou. De voorvechter van kwaliteit van leven, zeker als het leven anders loopt dan je had verwacht.

Lieve Lars, wat ben ik blij dat je mijn nummer hebt opgezocht in de Almanak 15 jaar geleden. Het leven is een feestje met jou en zal nooit gaan vervelen. Je bent mijn rots in de branding én mijn grootste plaaggeest. Je weet wat ik nodig heb en daagt me uit. Jij bent zonder enige twijfel de liefde van mijn leven en de meest fantastische vader die Elise en Philip zich kunnen wensen.

Lieve Elise en Philip, elk moment van de dag brengen jullie een glimlach op mijn gezicht en geniet ik van jullie. Lief, vrolijk en eigenwijs, nu al grote persoonlijkheden. Heerlijk is het om jullie te zien opgroeien en de wereld te zien ontdekken. Blijf dat nog maar heel lang doen.

CURRICULUM VITAE

Annette van Nieuwenhuizen was born on August 6, 1986 in Oegstgeest, the Netherlands. She attended the Kennemer Lyceum in Overveen, where she received her athenaeum diploma in 2004. In the same year she started her study in medicine at the Vrije Universiteit in Amsterdam. During her study, she studied quality of life in patients with laryngeal papillomatosis, supervised by Rinkel and prof. Verdonck de Leeuw. After obtaining her medical degree in 2011 she started her first research project at the department of Otolaryngology and Head and Neck Surgery of the VU University medical center Amsterdam. She studied health related quality of life in patients with head and neck cancer, which resulted in this thesis. During this period, several chapters of this thesis have been presented in national and international conferences. On june 6, 2013, Annette and her team of colleges climbed the Alpe d'Huez by bike to raise money for the Alpe d'HuZes foundation/Dutch cancer society. In 2014 she commenced her training as resident otorhinolaryngology at the VU University medical center Amsterdam under supervision of prof. dr. C.R. Leemans and Dr. Merkus consecutively. During her residency she worked at the Tergooi ziekenhuis in Blaricum and the Spaarne Gasthuis in Hoofddorp, under supervision of Dr. Borgstein and Dr. E.J.B. van Nieuwkerk, respectively. Currently Annette is in her last year of her specialization in the field of otorhinolaryngology. She is married to Lars Wehman and together they have two children, Elise (2016) and Philip (2019).



LIST OF PUBLICATIONS

A.J. van Nieuwenhuizen, L.M. Buffart, H. Langendijk, M. Vergeer, J. Voortman, C.R Leemans, IM Verdonck-de Leeuw. Health Related Quality of Life and overall survival: a prospective study in patients with head and neck cancer treated with radiotherapy. Submitted to Health and Quality of Life Outcomes (IF 2.32, Q1 Rehabilitation)

A.J. van Nieuwenhuizen, L.M. Buffart, C.F. van Uden-Kraan, C.R. Leemans, I.M Verdonck-de Leeuw. Patient-reported physical activity and the association with health related quality of life in head and neck cancer survivors. Supportive care in cancer 2017-11-21 (IF 2.59, Q1 Rehabilitation)

A.J. van Nieuwenhuizen, L.M. Buffart, J. Brug, C.R. Leemans, I.M Verdonck-de Leeuw. The association between health related quality of life and survival in patients with head and neck cancer: a systematic review. Oral Oncol. 2015;51(1):1-11. (IF 2.70, Q1 Dentistry)

A.J. van Nieuwenhuizen, L.M. Buffart, J.H. Smit, R.H. Brakenhoff, B.J.M. Braakhuis, R. de Bree, C.R. Leemans, I.M Verdonck-de Leeuw. A comprehensive assessment protocol including patient reported outcomes, physical tests, and biological sampling in newly diagnosed patients with head and neck cancer: is it feasible? Supportive Care in Cancer. 2014, (IF 2.59, Q1 Rehabilitation)

Verdonck-de Leeuw IM, van **Nieuwenhuizen A**, Leemans CR. The value of qualityof-life questionnaires in head and neck cancer. Curr Opin Otolaryngol Head Neck Surg. 2012 Apr;20(2):142-7. (IF 1.87, Q1 Otolaryngology)

van Nieuwenhuizen AJ, Rinkel RN, de Bree R, Leemans CR, Verdonck-de Leeuw IM. (2010) Patient reported voice outcome in patients with recurrent respiratory papillomatosis. The Laryngoscope, 2010 Jan;120(1):188-92. (IF1.83, Q2 Otolaryngology)

Joffrey van Prehn, **A. J. van Nieuwenhuizen**, S. van Weert, M. van Seijen and A. E. Budding (2015). Salivary gland infection with Candida tropicalis: antibiotic suppression therapy resulting in selection of uncommon pathogens. JMM Case Reports